ISSN 1015-7999 (print) ISSN 2313-7835 (online) South African Computer Journal

SACJ - Volume 37 Number 1 - July 2025

July 2025 - Volume 37 Number 1



Volume 37 Number 1, July 2025 ISSN 1015-7999 (print) – ISSN 2313-7835 (online)

A publication of the South African Institute of Computer Scientists and Information Technologists Open Access: https://www.sacj.org.za

Editor-in-Chief

sacj.editor@gmail.com

Prof Katherine Malan University of South Africa malankm@unisa.ac.za

Assistant Editor

Prof Ian Sanders University of the Witwatersrand dr.id.sanders@gmail.com

Associate Editors

Prof Marijke Coetzee North West University marijke.coetzee@nwu.ac.za

Prof Aurona Gerber Stellenbosch University auronagerber@sun.ac.za

Prof Tendani Mawela University of Pretoria tendani.mawela@up.ac.za

Prof Hanlie Smuts University of Pretoria hanlie.smuts@up.ac.za Prof Sigrid Ewert University of the Witwatersrand sigrid.ewert@wits.ac.za

Prof Hugo Lotriet University of South Africa lotrihh@unisa.ac.za

Prof Deshen Moodley University of Cape Town deshen@cs.uct.ac.za

Prof Terence van Zyl University of Johannesburg tvanzyl@uj.ac.za

Editorial Board

Prof Judith M Bishop Stellenbosch University, South Africa

Prof Judith van Biljon University of South Africa

Prof Rossouw van Solms Nelson Mandela University, South Africa

Prof Irwin Brown University of Cape Town, South Africa Prof Anthony Maeder University of Western Sydney, Australia

Prof Derrick Kourie Stellenbosch University, South Africa

Prof Paula Kotzé University of Pretoria, South Africa

Prof Sue Conger University of Dallas, Irving, TX, USA

Production Editor sacj.productioneditor@gmail.com

> Prof Etienne van der Poel University of South Africa

NOTES FOR CONTRIBUTORS

The *South African Computer Journal* is an accredited specialist academic journal, publishing research articles, technical reports and communications in English in the Computer Science, Computer Systems and Information Systems domains. Its primary target is research of interest in Africa or published by African researchers, but all quality contributions are considered. All research articles submitted for publication are rigorously refereed by independent peer reviewers. The journal publishes original work that is of international stature. The editorial board comprises local and international scholars of high repute. The journal is published online using the *open access* model, making papers accessible in developing countries where funding to subscribe is scarce.

Submissions

Authors should submit papers for publication at the web site at http://www.sacj.org.za/inde x.php/sacj/about/submissions. Please also check there for the latest version of the below guidelines.

Form of Manuscript

Manuscripts for *review* should be prepared according to the following guidelines, which summarize more detailed instructions on the web site.

SACJ has a double-blind reviewing policy. No author's name or affiliation should appear anywhere. Citing of previous work by the author or authors should be anonymised if appropriate. Acknowledgments and thanks should not be included in the draft for review. If you use Microsoft Word please make sure that your name and affiliation are not saved in the document properties.

- The article should start as follows:
 - the title (as brief as possible)
 - an abstract of less than 200 words
 - an appropriate keyword list
 - a list of ACM Categories (2012 scheme).
- Tables and figures should be numbered, in sequence, titled, and referenced in the text by number.
- *SACJ* uses American Psychological Association 7th edition style in final, published papers, as described here (https://apastyle.apa.org/). Manuscripts for *submission* should use numeric citation, as APA version 7 is difficult to get right, and the numeric citations are easier to process in the production side of the paper. References should be listed at

the end of the text in alphabetic order of the (first) author's surname and cited in the text.

If you use an appropriate $BIBT_EX$ style, this will work automatically; do not spend a lot of time on reference citation minutiae since the production editor will take care of that sort of detail.

- If a DOI is available for any reference cited, include the DOI in the reference list. The DOI should be a complete URL (ideally clickable), for example: https://doi.org/10.184 89/sacj.v34i1.1115 (DOI display guidelines here: http://www.crossref.org/02publish ers/doi_display_guidelines.html)
- If a DOI is not available a URL to the original document or the publication containing the document should be provided.

SACJ is produced using the LTEX document preparation system and we recommend using the Overleaf system. A SACJ template is available on Overleaf: https://www.overleaf.com/latex/templates/south-african-computer-journal/smnhsnmsnfdy. Though we can also accept Microsoft Word submissions, delays in publication are more likely with the latter format. While we encourage submission in a format similar to the published paper, authors should not waste undue time on layout as this will be redone by the production editor.

Authors retain the right to republish their work, subject to any republished version including a reference to the *SACJ* version.

Publication Charges

A charge of R6000 will be levied on papers accepted for publication to cover costs of open access publication. Where the author's institution or research budget is unable to meet this charge, it may be waived upon request of the author and at the discretion of the editor-in-chief.

Proofs

Proofs of accepted papers will be sent to the corresponding author to ensure that typesetting is correct, and not for addition of new material or major amendments to the text. Corrected proofs should be returned to the production editor within three days.

Extended Conference Papers

Authors of conference papers are welcome to submit extended papers to *SACJ* for consideration on these terms:

- a covering letter accompanying submission should explain what is added to the paper to make it worth publishing as a journal paper
- the paper includes at least 30% new material

- a pointer is provided to the original paper or, if it is not freely available, upload it as supplementary material when submitting the extended paper to *SACJ*
- evidence is provided that republication in this form does not violate copyright of the conference publication

Book Reviews, Letters and Communications

Book reviews are welcome, as are letters to the editor; both should carry the author's full name and affiliation, and should be limited to 500 words. Communications and Viewpoints up to two pages in length (or longer, by negotiation with the editor-in-chief) may also reflect minor research contributions, works-in-progress or ideas to stimulate debate.

This category of submission is accepted at the discretion of the editor-in-chief, not refereed and does not qualify as a research publication for South African government subsidy purposes. The major criteria for acceptance are that the item is coherently written and does not require significant editing, that it is timely and it is likely to be of interest to readers.

Volume 37 Number 1, July 2025 ISSN 1015-7999 (print) – ISSN 2313-7835 (online)

A publication of the South African Institute of Computer Scientists and Information Technologists Open Access: https://www.sacj.org.za

CONTENTS

Special Issue

Editorial: Impact of Innovative IT Solutions

Katherine M. Malan 💿 – sacj.editor@gmail.com

Department of Decision Sciences, University of South Africa

All of the research articles published in this issue synergistically focus on the potential impact of innovative IT solutions on society and business. The topics include the use of big data analytics in rural hospitals, digital sharing platforms to support SMEs, detective analytics to mitigate financial crime, virtual reality to support psychotherapy, the impact of digital technology on teaching in schools, and the transformative potential of generative AI. The six research papers included in this issue are:

- Identifying factors affecting the adoption of big data analytics in South African ruralbased hospitals to improve service delivery by Rambau, Munyoka and Mashau.
- Contextualising Design: Aligning digital sharing economy platforms with local SMEs' sharing practices in resource-constrained countries by Abebe and Twinomurinzi.
- The use of detective analytics for mitigating financial crimes: A South African perspective by Mlambo and Iyamu.
- An investigation into the feasibility of using virtual environments as an induction method in SHIP[®] therapy by Malan, Bosman and Bothma.
- A multilevel analysis of digital technology to support teachers to improve their professional practice by Dlamini and Fleisch.
- Generative AI: Concerns, usage, challenges, opportunities and sentiments by Twinomurinzi and Gumbo. This is the second paper of a special issue of extended papers from the SAICSIT 2023 conference.

Authors of these papers are affiliated with seven different higher education institutions in South Africa, showing that SACJ remains an important publication outlet for South African researchers.

Submission statistics

For information, we present the following updated statistics around submissions, acceptance rates and the processing time of articles.

Malan, K.M. (2025). Editorial: Impact of Innovative IT Solutions [Editorial]. *South African Computer Journal* 37(1), vii–viii. https://doi.org/10.18489/sacj.v37i1.22860

Copyright © the author(s); published under a Creative Commons NonCommercial 4.0 License (SACJ is a publication of SAICSIT. ISSN 1015-7999 (print) ISSN 2313-7835 (online)

Year	2022	2023	2024
Submissions	111	95	96
Desk rejection	54	62	56
Desk rejection rate	49%	65%	58%
Final acceptance	17	12	
Acceptance rate	15%	13%	

The figures above translate into an overall rejection rate of approximately 86%. This may seem high, but it can be attributed to the large number of submissions from all over the world due to the indexing of SACJ on Scopus. To quantify the reach of SACJ, between 1 January and 30 June 2025 we received 56 submissions from 22 different countries (11 from South Africa, 15 from other African countries, and 30 from outside Africa).

Similar to SACJ, South African Journal of Science reported a rejection rate of 82% in 2024. In terms of submission processing time, the median time to a first decision (desk reject / send to review) since 2023 was 2 days and the median time to final decision for submissions sent to review was 5 months.

Thank you to all authors for submitting your manuscripts to SACJ and for entrusting us with your research. Thank you also to the editorial team for all your work behind the scenes and to the many reviewers who support our journal.

Identifying factors affecting the adoption of big data analytics in South African rural-based hospitals to improve service delivery

Thifhindulwi Maxwell Rambau^(b), Willard Munyoka^(b), Nkhangweni Lawrence Mashau^(b)

University of Venda, Thohoyandou, 0950, South Africa

ABSTRACT

The healthcare sector is progressively recognising the potential of big data analytics to revolutionise patient care, operational efficiency and decision-making processes. However, rural-based hospitals face challenges in embracing big data analytical tools to improve their service delivery to their patients. The adoption of big data analytics will assist rural-based hospitals in predicting and making informed decisions to manage the limited resources that they have and allocate them appropriately. However, there is a lack of literature that explores and identifies the factors that affect the adoption of big data analytics in rural-based hospitals to improve service delivery. This research employs a quantitative research methodology using surveys to identify the factors that affect the adoption of big data analytics in rural-based hospitals. This study found that rural-based hospitals prefer a hybrid method to collect data and it impedes the adoption of big data analytics in rural-based hospitals. Therefore, there is a need to implement modern infrastructure to integrate various data collection methods to promote the adoption of big data analytics in rural-based hospitals.

Keywords big data analytics, rural-based hospitals, healthcare, decision-making, service delivery

 $\label{eq:categories} \textbf{ o Information systems} \sim \textbf{big data, data analytics}$

Email	Article history	
Thifhindulwi Maxwell Rambau – thifhindulwi.rambau@univen.ac.za	Received:	31 January 2024
Willard Munyoka – willard.munyoka@univen.ac.za	Accepted:	12 March 2025
Nkhangweni Lawrence Mashau – lawrence.mashau@univen.ac.za (corresponding)	Online:	2 July 2025

1 INTRODUCTION

Globally, rural-based hospitals play a critical role in the local communities by providing emergency medical services to the citizens (Miller et al., 2020). These hospitals use a traditional method to collect data and provide services to the citizens (Khalifa & Househ, 2021; Malelelo-Ndou et al., 2019; Nevhutalu, 2016). However, urban hospitals use diverse platforms to collect data from the patients, which results in the government storing tons and tons of data (Kaur

Rambau, T.M., Munyoka, W, and Nkhangweni, L.M. (2025). Identifying factors affecting the adoption of big data analytics in South African rural-based hospitals to improve service delivery. *South African Computer Journal* 37(1), 1–13. https://doi.org/10.18489/sacj.v37i1.17830

Copyright © the author(s); published under a Creative Commons NonCommercial 4.0 License (1)(5) SACJ is a publication of SAICSIT. ISSN 1015-7999 (print) ISSN 2313-7835 (online)

et al., 2021; Kelvin & Morrisson, 2023; Rehman et al., 2021). In the disposal of this data, rural-based hospitals are still grappling with managing their resources and allocating them accordingly in order to provide efficient service delivery (Batko & Ślęzak, 2022; McCarthy et al., 2021).

Nevhutalu (2016) reported that patients in many rural-based hospitals express frustration with the services received. These complaints often stem from the lengthy process of retrieving their files and waiting to be seen by a medical practitioner (McCarthy et al., 2021; Miller et al., 2020; Nevhutalu, 2016). Furthermore, patients with chronic diseases are subjected to long queues during their compromised health state (McCarthy et al., 2021; Nevhutalu, 2016; Zolbanin et al., 2022). However, other scholars have suggested big data analytics (BDA) to curb these challenges (Gomes et al., 2022; Imran et al., 2021; Ramesh & Santhi, 2020). Recently, BDA has become a popular mechanism for the survival and growth of many institutions worldwide (Brossard et al., 2022; Khanra et al., 2020; Miller et al., 2020).

Likewise, healthcare institutions are also progressively recognising the potential of BDA to transform patient care, operational efficiency and decision-making processes (Batko & Ślęzak, 2022; Ghaleb et al., 2021; Ramesh & Santhi, 2020). In recent years, these institutions have witnessed a paradigm shift with the advent of BDA, presenting unprecedented opportunities to manage and use resources efficiently daily (Gomes et al., 2022; Kelvin & Morrisson, 2023; Yu et al., 2021). However, these potential benefits of BDA are well documented from an urban hospitals' perspective (Shafqat et al., 2020; Shahbaz et al., 2019; Zolbanin et al., 2022).

On the contrary, there is a lack of evidence of the opportunities presented by BDA in ruralbased hospitals. This challenge is attributed to the lack of widespread adoption of BDA in rural-based hospitals. In the context of this study, adoption refers to embracing and using BDA tools and technologies to inform rural-based hospitals' decision-making to improve service delivery. Adoption, in this sense, means the incorporation of BDA tools into hospitals' day-today operations to respond quickly in addressing patients' needs.

Other studies explored the challenges of data quality and implementation of computerbased knowledge management systems in public healthcare systems in urban areas (Botha et al., 2014; Maramba et al., 2020). Some of the studies that were conducted in rural public hospitals investigated the use of a virtual ICT framework to support the doctors (Nevhutalu, 2016). However, various scholars have postulated that it is significant for rural-based hospitals to adopt BDA to improve service delivery to their patients (Brossard et al., 2022; Miller et al., 2020; Nevhutalu, 2016). Therefore, this study seeks to address the gap in the existing literature by delving into the factors that affect the adoption of BDA in rural-based hospitals, with a primary focus on improving service delivery. Furthermore, the ultimate goal is to promote the successful adoption of BDA in rural-based hospitals, thereby enhancing service delivery, improving patient care, and addressing healthcare disparities in underserved communities (Batko & Ślęzak, 2022; Ghaleb et al., 2021).

This study is divided into six sections: Section 2 presents a literature review on big data analytics in healthcare institutions. Section 3 presents the research approach to choosing rural-based hospitals and their personnel who responded to the questionnaire. Section 4 presents

the results of the study and Section 5 discusses the results pertinent to the identification of the factors that affect the adoption of the BDA in rural-based hospitals. Section 6 concludes and provides recommendations to the rural-based officials who are contemplating adopting BDA (Khanra et al., 2020).

2 BIG DATA ANALYTICS IN THE HEALTHCARE INSTITUTIONS

Various scholars see big data as a precursor to big data analytics (Benzidia et al., 2021; Kelvin & Morrisson, 2023). Big data refers to enormously large and complex datasets that surpass the processing capabilities of traditional data analytical technologies and require sophisticated analytical methods and technologies for storing, processing and analysing (Hiremath et al., 2023; Kelvin & Morrisson, 2023; Seefong et al., 2023). Big data is characterised by four V's: velocity, volume, veracity and variety (Hiremath et al., 2023; Leow et al., 2023).

However, in the healthcare institutions big data is described as a high volume of health data collected from various sources, including physician notes, electronic health records (EHR), medical imaging, lab reports, X-Ray reports, case history, diet regime, genomic sequencing, payer records, pharmaceutical research, wearables, medical devices and many other sources that are not mentioned (Kaur et al., 2021; Ramesh & Santhi, 2020).

To analyse big data, there is a need of BDA techniques to uncover hidden information to assist organisations with decision marking (Leow et al., 2023; Yu et al., 2021). BDA refers to the process of examining and uncovering meaningful patterns, trends, and insights within vast and complex datasets, commonly known as big data (Ghaleb et al., 2021; Leow et al., 2023). In healthcare institutions, BDA involves the systematic analysis of vast and diverse datasets to extract valuable insights, identify patterns, and support informed decision-making for improving patient care, operational efficiency, and overall healthcare quality (Khanra et al., 2020; Zolbanin et al., 2022).

BDA in healthcare institutions includes the use of different techniques, such as statistical analysis, machine learning, data mining, and predictive modeling (Gomes et al., 2022; Shafqat et al., 2020). Most healthcare institutions leverage these analytics capabilities to gain more patient insights, optimise hospital processes, improve operational efficiency, and drive innovation (Batko & Ślęzak, 2022; Hiremath et al., 2023; Kaur et al., 2021). Overall, BDA plays a crucial role in helping organisations make informed decisions, identify opportunities and address challenges in today's data-intensive and rapidly evolving business landscape (Botha et al., 2014; Gomes et al., 2022).

Furthermore, integrating BDA in healthcare institutions with big data applications has the potential to enhance shared knowledge, monitor the outcomes of prevention focused approaches and enhance the efficiency of patient management (Gomes et al., 2022; Khalifa & Househ, 2021; Rehman et al., 2021). This context, often ignored by managers, engineers and policymakers, may present an opportunity that warrants exploration within the healthcare sector (Khalifa & Househ, 2021; Shafqat et al., 2020).

3 RESEARCH METHODOLOGY

This research employed a quantitative research methodology to investigate the factors that affect the adoption of big data analytics in rural-based hospitals to generalise the findings from a sample to the given population and draw conclusions that will apply beyond the sampled hospitals (Cresswell & Clark, 2011; Kothari, 2004; Saunders et al., 2019). The study focused on four rural-based hospitals in the Vhembe district in the Limpopo province, South Africa, utilising a convenience sampling method for selecting the hospitals. In this case, convenience sampling was used to select rural hospitals based on their availability and accessibility (Klein & Myers, 1999; Myers, 2013). Furthermore, this method was used for its simplicity and practicality, especially because of the tight given timeframe, budget and other constraints that made it difficult to implement more rigorous sampling techniques (Borgstede & Scholz, 2021; Crotty, 1998).

The target population included workers from the selected rural-based hospitals' Information and Communication Technology (ICT) departments, administrators, doctors and nurses. The sample comprised both computer-users and non-computer-users in the ICT departments who worked as data collectors. Computer users comprised of administrators and IT personnel. In addition, non-computer users are comprised of doctors and nurses. A total of 200 participants were sampled using simple random sampling techniques. Simple random sampling is considered one of the most straightforward and unbiased techniques for selecting a representative sample from a larger population (Myers, 2013). Furthermore, this method afforded all the participants from the population an equal opportunity to be selected (Kothari, 2004; Saunders et al., 2019). Fifty questionnaires were distributed to each selected hospital.

To collect data, this study utilised a structured questionnaire designed to gather information on factors influencing the adoption of BDA in rural-based hospitals. The questionnaire exclusively featured closed-ended questions, allowing for standardised responses and quantitative data analysis (Myers, 2013; Saunders et al., 2019). The questionnaire covered key aspects related to the adoption of BDA and questions were designed to assess participants' perspectives on the barriers that affect the incorporation of big data analytics into their daily work.

Prior to data collection, ethical approval was obtained from the University of Venda (UNI-VEN), Faculty of Management, Commerce and Law ethics review committee. Participants were provided with informed consent forms explaining the purpose of the study, ensuring confidentiality and emphasising their voluntary participation. The anonymity and privacy of participants were strictly maintained throughout the research process.

4 RESULTS

From the distributed questionnaires, 174 were returned. Out of 174 questionnaires, 17 were not completed in full and were discarded from the analysis. Out of the collected questionnaires, 150 questionnaires were analysed using appropriate statistical methods, such as descriptive

statistics and inferential statistics. The aim of data analysis was to identify patterns, correlations and trends in the responses, providing insights into the factors that affect the adoption of big data analytics in rural-based hospitals. Results extracted from participants are presented in the following subsections:

4.1 Participants' demographic information

Participants' demographic information is depicted in Figure 1. The gender distribution in Figure 1 shows a slightly higher representation of females (59,3%) compared to males (40,7%). This gender balance is important for ensuring diverse perspectives in the study, particularly considering the potential impact of gender on attitudes and perceptions.



Figure 1: Demographic information

The age distribution shows that few participants (12,00%) were between the age of 22–29 and 24% of the respondents were between the age of 45 and above. While a diverse range of participants, with a significant portion falling within the 30–36 age groups (30,70%) and 37–44 age group (32,70%). This diversity is beneficial for capturing a broad range of experiences and viewpoints, as different age groups may have varying levels of familiarity and comfort with technology, including BDA tools.

Similarly, the educational backgrounds of participants are diverse, with a substantial proportion of participants holding university degrees. A total of 21,30% of the participants hold a diploma, while 34,70% hold a bachelor's degree. The highest proportion (36,00%) of the participants hold postgraduate qualifications. This indicates that the rural-based hospitals have a well-balanced sample of educated staff, which may positively impact their ability to comprehend and engage with complex topics such as BDA.

Furthermore, the results in Figure 1 show that the majority of participants (70,70%) reported having either good or excellent computer literacy skills proficiency. This high level of

computer literacy is a positive aspect, suggesting that a significant portion of the sample may be well-equipped to engage with and adopt new technologies, such as BDA tools.

4.2 Status of big data

A significant majority of respondents (67,30%) indicated that they collect data using both data collection methods (see Figure 2). While 32,70% of the respondents indicated that they use traditional data collection method. However, there was no participant who used a digital platform as a single means of collecting data from the source systems. The results show that rural-based hospitals are using a hybrid data collection method.



Figure 2: Status of big data

The researchers also explored data storage in rural-based hospitals. Figure 2 shows that the majority of respondents (75.00%) indicated that the data is stored in a centralised storage. While 25,00% of the respondents indicated that the data is not centralised. Furthermore, the researcher examined if the participants were aware of BDA tools. All the participants indicated that they were aware of the BDA tools.

Figure 2 presents the results of the respondents' perceptions of the extent to which ruralbased hospitals use BDA. The majority of respondents (76.00%) indicated that rural-based hospitals use BDA to an average extent. This suggests that, according to the respondents' perceptions, these hospitals are actively incorporating BDA into their operations, albeit not at an extremely high level. A notable portion of respondents (23.00%) expressed the perception that the usage of BDA in rural-based hospitals is very little. While 1,00% of the respondents expressed that the usage of BDA is very much.

4.3 Perception of big data analytics

The results in Figure 3 show that, a vast majority of 92,00% of the respondents strongly agree that, in general, they find the integration of big data analytics useful in the rural-based hospitals and 6,70% of the respondents support this statement. This high level of agreement indicates a strong belief in the value and utility of incorporating BDA into hospital practices.

While 1,30% of the respondents are not sure of this statement. The results also show that there are no participants who dispute this statement.



Figure 3: Perceptions of big data analytics

This study also explored the capability of BDA tools to enhance rural-based hospitals' dayto-day operations. The results reveal that 19,3% of the participants agree and 80,00% of the participants strongly agree that the capabilities of BDA tools enhance day-to-day operations and can be compatible with the hospital's operations and service delivery. This underscores a positive perception of the potential impact of these tools on hospital functions.

Furthermore, the participants responded to a question about the need for BDA in ruralbased hospitals. An overwhelming majority of 6,70% of the respondents agree and 92,00% strongly agree that rural-based hospitals require BDA tools to address the integration of data, heterogeneity, complexity, and incompleteness of data. This suggests a consensus among participants regarding the critical role of BDA tools in managing diverse and complex healthcare data. Few participants (0,70%) were not sure about this and 0,60% strongly disagreed that rural-based hospitals require BDA tools to address the integration of data, heterogeneity, complexity, and incompleteness of data.

The majority of the respondents (85,30%) strongly agree that using big data analytics will improve rural-based hospitals' performance. While 12,00% of the respondents concur with the majority. This positive perception aligns with the potential benefits associated with the integration of analytics in healthcare operations. Only 2,00% of the participants indicated that they are unsure about using data analytics to improve healthcare performance. Fewer

respondents (0,70%) disagree that using big data analytics will improve rural-based hospitals' performance.

Furthermore, the researchers probed infrastructure compatibility with BDA tools in ruralbased hospitals. The results show a consensus that there is a lack of infrastructure that accommodates BDA tools in rural-based hospitals. A total of 98,00% of the participants agree (17,30%) and strongly agree (80,00%) that BDA tools are not compatible with the existing infrastructure in rural-based hospitals. However, 2,70% of the respondents were not sure about this statement.

In Figure 3 the results show a consensus that it is easy to use BDA tools. A total of 83,30% of the participants agree, out of 83,30%, 40,00% agree and 43,30% strongly agree that, in general, BDA tools are easy to use. On the other hand, 12,00% of the participants are uncertain about this statement. In addition, 4,70% of the participants highlighted that BDA tools are difficult to use.

As depicted in Figure 3, the results show that 29,30% of the respondents agree and 22,70% strongly agree that learning to use new features of BDA tools is easy. However, 38,70% of the respondents were uncertain about this statement. A total of 9,30% disagree that learning new features of BDA tools is easy. The responses on learning new features of BDA tools show a more varied pattern, indicating that opinions on ease of learning differ among participants.

The results in Figure 3 show that 44,60% of the respondents agree and 4,00% of the respondents strongly agree. Therefore, a substantial portion 48,60% of the respondents believe that organisational decision processes depend on information obtained from non-integrated systems. On the contrary, 46,00% of the respondents were uncertain about the organisational decision process, which depends on the information extracted from the data collected from non-integrated systems. This indicates a significant concern or acknowledgment among participants about the influence of non-integrated systems on decision-making. A total of 5,40% of the respondents disagree with this statement.

Lastly, a notable 32,70% percentage feel that learning how to use BDA tools takes too long, which may impact the perceived value of investing time and effort in utilising these tools. While on the other hand, 24,00% of the respondents are not sure how long it takes to learn BDA tools. A total of 43,30% disagree with the statement.

In Figure 3 the results show a consensus that it is easy to use BDA tools. A total of 80,30% of the participants agree (40,00%) and strongly agree (43,30) that, in general, BDA tools are easy to use. On the other hand, 12,00% of the participants are uncertain about this statement. In addition, 4,70% of the participants highlighted that BDA tools are difficult to use.

The majority of the respondents (85,30%) strongly agree that using big data analytics will improve rural-based hospitals' performance. While 12,00% of the respondents concur with the majority. This positive perception aligns with the potential benefits associated with the integration of analytics in healthcare operations. Only 2,00% of the participants indicated that they are unsure about using data analytics to improve healthcare performance. Fewer respondents (0,70%) disagree that using big data analytics will improve rural-based hospitals' performance.

Lastly, a vast majority of 92,00% of the respondents strongly agree that, in general, they find the integration of big data analytics useful in the rural-based hospitals and 6,70% of the respondents support this statement. This high level of agreement indicates a strong belief in the value and utility of incorporating BDA into hospital practices. While 1,30% of the respondents are not sure of this statement. The results also show that there are no participants who dispute this statement.

5 DISCUSSION OF THE RESULTS

The results showed a gender balance, and it is essential to ensure diverse perspectives in the study, particularly considering the potential impact of gender on attitudes and perceptions. Furthermore, age was explored to solicit a diverse opinion and capture a broad spectrum of experiences and viewpoints because different age groups may have different perceptions when it comes to the adoption of innovation in an organisation (Rogers, 2003). The results show that all genders and ages are aware that BDA will enhance decision-making to improve service delivery in rural-based hospitals. This is in support with McCarthy et al. (2021) and Mashau and Mokwena (2017) who indicated that awareness has a positive impact on the adoption of new innovation.

The results also showed a good balance of education level and computer literacy skills proficiency. This may positively impact the ability to adopt BDA in rural-based hospitals because their staff are well-equipped to engage and adopt new technologies such as big data analytics tools. Various scholars postulate that education and computer literacy are paramount when adopting new technologies (Mashau et al., 2021; Rogers, 2003; Yadegaridehkordi et al., 2019).

The data suggests a preference for a hybrid approach in rural-based hospitals for data collection, incorporating both traditional paper-based and digitalised data collection methods. An enormous amount of data is still collected using the traditional paper-based data collection method. This could be a hindrance to adopting BDA in rural-based hospitals. Various scholars highlight the need to automate processes to collect data and to analyse it for decision-making purposes (Gomes et al., 2022; McCarthy et al., 2021; Shafqat et al., 2020). Furthermore, 25,00% of rural-based hospitals' data is not centralised, which will result in wrong information. When analysing data, there is a need to centralise all collected data to ensure the accuracy of the information (Benzidia et al., 2021; Leow et al., 2023). Decentralisation of data could be the reason for the low usage of BDA in rural-based universities (Benzidia et al., 2021; Ramesh & Santhi, 2020).

The results also revealed that infrastructure is incompatible with BDA tools in rural-based hospitals. Thus, the literature postulates for the successful adoption of innovation, the infrastructure should be up to date (Mashau et al., 2021; Rehman et al., 2021). However, organisations (hospitals) may improve their infrastructure to accommodate innovation (Imran et al., 2021; Rehman et al., 2021). Lack of infrastructure is a main barrier to the adoption of innovation in most organisations (Mashau et al., 2021; Shafqat et al., 2020).

6 CONCLUSION AND RECOMMENDATIONS

This study explored the factors that affect the adoption of big data analytics in rural-based hospitals with a focus of improving service delivery. The insights gathered through the survey responses presented the current landscape by providing valuable information about the staff of rural-based hospitals. The demographic information revealed a diverse and well-educated sample with a balanced gender distribution. The majority of participants demonstrated a high level of computer literacy, a crucial factor for successful implementation of big data analytics tools in rural-based hospitals.

Furthermore, this study found that rural-based hospitals prefer a hybrid method to collect data from the sources. This was highlighted as one of the factors that impede the adoption of big data analytics in rural-based hospitals. In addition, this study also found that data is not centralised, which may result in inaccurate information. Centralised data may present potential benefits in terms of streamlined access, improved consistency and enhanced security, but it also necessitates careful consideration of governance and security measures. Lastly, this study found that infrastructure is still an issue in rural-based hospitals. In rural-based hospitals, the infrastructure must be upgraded to be compatible with big data analytics tools. Therefore, these findings of this study contribute to the body of knowledge by offering a deeper understanding of the factors influencing the adoption of big data analytics in rural hospitals, thereby informing strategies for enhancing healthcare service delivery and quality in rural areas.

This study recommends that rural-based hospitals should automate all their processes because it is difficult to clean and extract meaningful information from data that are in different forms. This will also assist rural-based hospitals in centralising their data, making it easier for knowledge workers to analyse it. In addition, this study recommends that rural-based hospitals should invest more in upgrading their infrastructure to adopt big data analytics successfully.

Since the healthcare landscape continues to evolve, the findings from this study serve as a foundation for healthcare managers, policymakers, decision-makers, administrators, and technology developers. Furthermore, healthcare stakeholders can leverage the insights gained from this study and navigate the complexities of adopting big data analytics in rural-based hospitals, ultimately contributing to more effective and patient-centric healthcare services. The recommendations derived from this study aim to guide future initiatives, ensuring that the integration of big data analytics aligns with the unique needs and challenges of rural-based hospitals.

Future studies may explore critical factors that can affect the implementation of modern digital infrastructure that will support big data analytics and Industry 4.0 in rural-based hospitals to enhance decision-making. Finally, it was significant to investigate the factors that affect the adoption of big data analytics in South African rural-based hospitals in order to assist them in addressing all key factors before implementing big data analytics to potentially address critical gaps in healthcare service delivery.

References

- Batko, K., & Ślęzak, A. (2022). The use of big data analytics in healthcare. *Journal of Big Data*, *9*(1), 1–24. https://doi.org/10.1186/S40537-021-00553-4
- Benzidia, S., Makaoui, N., & Bentahar, O. (2021). The impact of big data analytics and artificial intelligence on green supply chain process integration and hospital environmental performance. *Technological Forecasting and Social Change*, 165, 120557. https://doi.org/10.1016/J.TECHFORE.2020.120557
- Borgstede, M., & Scholz, M. (2021). Quantitative and qualitative approaches to generalization and replication – A representationalist view. *Frontiers in psychology*, *12*. https://doi.or g/10.3389/FPSYG.2021.605191
- Botha, M., Botha, A., & Herselman, M. (2014). Compiling a prioritized list of health data quality challenges in public healthcare systems. *2014 IST-Africa Conference Proceedings*, 1–8. https://doi.org/10.1109/ISTAFRICA.2014.6880649
- Brossard, P. Y., Minvielle, E., & Sicotte, C. (2022). The path from big data analytics capabilities to value in hospitals: A scoping review. *BMC Health Services Research*, *22*(1), 1–16. htt ps://doi.org/10.1186/S12913-021-07332-0
- Cresswell, J. W., & Clark, V. L. P. (2011). *Designing and conducting mixed methods research* (2nd ed.). California: Sage Publication.
- Crotty, M. J. (1998). The foundation of social research. London, New Delhi: Sage Publications.
- Ghaleb, E. A., Dominic, P. D., Fati, S. M., Muneer, A., & Ali, R. F. (2021). The assessment of big data adoption readiness with a technology–organization–environment framework: A perspective towards healthcare employees. *Sustainability*, *13*(15), 8379. https://doi.org/10.3390/SU13158379
- Gomes, M. A. S., da Silva, V. L., Kovaleski, J. L., & Pagani, R. N. (2022). Big data analytics applied to health services: A literature review. *Exacta*, *20*(3), 647–665. https://doi.or g/10.5585/exactaep.2021.17297
- Hiremath, S., Shetty, E., Prakash, A. J., Sahoo, S. P., Patro, K. K., Rajesh, K. N., & Pławiak, P. (2023). A new approach to data analysis using machine learning for cybersecurity. *Big Data and Cognitive Computing*, 7(4), 176. https://doi.org/10.3390/BDCC7040176
- Imran, S., Mahmood, T., Morshed, A., & Sellis, T. (2021). Big data analytics in healthcare. a systematic literature review and roadmap for practical implementation. *IEEE/CAA Journal of Automatica Sinica*, 8(1), 1–22. https://doi.org/10.1109/JAS.2020.1003384
- Kaur, K., Verma, S., & Bansal, A. (2021). IOT big data analytics in healthcare: Benefits and challenges. Proceedings of IEEE International Conference on Signal Processing, Computing and Control, 2021-October, 176–181. https://doi.org/10.1109/ISPCC53510.2021.96 09501
- Kelvin, J. W., & Morrisson, M. (2023). Big data analytics capabilities and performance of private hospitals in Nairobi City County, Kenya. *Journal of Business and Management Sciences*, 11(3), 169–175. https://doi.org/10.12691/JBMS-11-3-1

- Khalifa, M., & Househ, M. (2021). Utilizing health analytics in improving the performance of hospitals and healthcare services: Promises and challenges. In *Multiple perspectives on artificial intelligence in healthcare* (pp. 23–39). Springer Science; Business Media Deutschland GmbH. https://doi.org/10.1007/978-3-030-67303-1_3
- Khanra, S., Dhir, A., Islam, N., & Mäntymäki, M. (2020). Big data analytics in healthcare: A systematic literature review. *Enterprise Information Systems*, *14*(7), 878–912. https://d oi.org/10.1080/17517575.2020.1812005
- Klein, H. K., & Myers, M. D. (1999). A set of principles for conducting and evaluating interpretive field studies in information systems. *MIS Quarterly*, 23(1), 67–94. https://doi.o rg/10.2307/249410
- Kothari, C. R. (2004). *Research methodology: Methods and techniques*. New Delhi: Bangalore: New Age International (P) Ltd.
- Leow, K. R., Leow, M. C., & Ong, L. Y. (2023). A new big data processing framework for the online roadshow. *Big Data and Cognitive Computing*, 7(3), 123. https://doi.org/10.339 0/BDCC7030123
- Malelelo-Ndou, H., Ramathuba, D. U., & Netshisaulu, K. G. (2019). Challenges experienced by health care professionals working in resource-poor intensive care settings in the Limpopo province of South Africa. *Curationis*, *42*(1), 8. https://doi.org/10.4102 /CURATIONIS.V42I1.1921
- Maramba, G., Coleman, A., & Ntawanga, F. F. (2020). Causes of challenges in implementing computer-based knowledge management systems in healthcare institutions: A case study of private hospitals in Johannesburg, South Africa. *African Journal of Information Systems*, 12(1), 4. https://digitalcommons.kennesaw.edu/ajis/vol12/iss1/4/
- Mashau, N. L., Kroeze, J. H., & Howard, G. R. (2021). An integrated conceptual framework to assess small and rural municipalities' readiness for smart city implementation: A systematic literature review. *Lecture Notes in Computer Science*, 13117 LNCS, 262–273. https://doi.org/10.1007/978-3-030-91540-7_28
- Mashau, N. L., & Mokwena, S. N. (2017). Adoption of instant messaging for mathematics lessons in rural schools. *International Electronic Journal of Mathematics Education*, *12*(3), 447–462. https://doi.org/10.29333/IEJME/624
- McCarthy, S., Moore, D., Smedley, W. A., Crowley, B. M., Stephens, S. W., Griffin, R. L., Tanner, L. C., & Jansen, J. O. (2021). Impact of rural hospital closures on health-care access. *Journal of Surgical Research*, 258, 170–178. https://doi.org/10.1016/j.jss.2020.08.05 5
- Miller, K. E., James, H. J., Holmes, G. M., & Houtven, C. H. V. (2020). The effect of rural hospital closures on emergency medical service response and transport times. *Health Services Research*, 55(2), 288–300. https://doi.org/10.1111/1475-6773.13254
- Myers, M. D. (2013). *Qualitative research in business and management* (2nd ed.). London: Sage Publications Ltd.

- Nevhutalu, H. K. (2016). Patient's rights in South Africa's public health system: Moral-critical perspectives [Doctoral dissertation, University of Stellenbosch]. http://hdl.handle.net /10019.1/98591
- Ramesh, T., & Santhi, V. (2020). Exploring big data analytics in health care. *International Journal of Intelligent Networks*, *1*, 135–140. https://doi.org/10.1016/J.IJIN.2020.11.0 03
- Rehman, A., Naz, S., & Razzak, I. (2021). Leveraging big data analytics in healthcare enhancement: Trends, challenges and opportunities. *Multimedia Systems*, 28(4), 1339–1371. ht tps://doi.org/10.1007/S00530-020-00736-8
- Rogers, E. M. (2003). Diffusion of innovations (5th ed.). New York: The Free Press.
- Saunders, M., Lewis, P., & Thornhill, A. (2019). *Research methods for business students* (8th ed.). Pearson Education Limited.
- Seefong, M., Wisutwattanasak, P., Se, C., Theerathitichaipa, K., Jomnonkwao, S., Champahom, T., Ratanavaraha, V., & Kasemsri, R. (2023). Big data analytics with the multivariate adaptive regression splines to analyze key factors influencing accident severity in industrial zones of Thailand: A study on truck and non-truck collisions. *Big Data and Cognitive Computing 2023, Vol. 7, Page 156*, 7(3), 156. https://doi.org/10.3390/BDCC7030156
- Shafqat, S., Kishwer, S., Rasool, R. U., Qadir, J., Amjad, T., & Ahmad, H. F. (2020). Big data analytics enhanced healthcare systems: A review. *Journal of Supercomputing*, 76(3), 1754–1799. https://doi.org/10.1007/S11227-017-2222-4
- Shahbaz, M., Gao, C., Zhai, L. L., Shahzad, F., & Hu, Y. (2019). Investigating the adoption of big data analytics in healthcare: The moderating role of resistance to change. *Journal* of Big Data, 6(1), 1–20. https://doi.org/10.1186/S40537-019-0170-Y
- Yadegaridehkordi, E., Shuib, L., Nilashi, M., & Asadi, S. (2019). Decision to adopt online collaborative learning tools in higher education: A case of top Malaysian universities. *Education and Information Technologies*, 24(1), 79–102. https://doi.org/10.1007/S10639-018-9761-Z
- Yu, W., Zhao, G., Liu, Q., & Song, Y. (2021). Role of big data analytics capability in developing integrated hospital supply chains and operational flexibility: An organizational information processing theory perspective. *Technological Forecasting and Social Change*, 163, 120417. https://doi.org/10.1016/J.TECHFORE.2020.120417
- Zolbanin, H. M., Davazdahemami, B., Delen, D., & Zadeh, A. H. (2022). Data analytics for the sustainable use of resources in hospitals: Predicting the length of stay for patients with chronic diseases. *Information & Management*, *59*(5), 103282. https://doi.org/10.1016/J.IM.2020.103282

Article history

Accepted: 20 September 2024

Received:

Online:

24 April 2024

2 July 2025

Contextualising Design: Aligning digital sharing economy platforms with local SMEs' sharing practices in resource-constrained countries

Sertse Abebe^{*, b}, Hossana Twinomurinzi^c

^a College of Computer Science and Information Technology, Sudan University of Science and Technology, Khartoum, Sudan

^b Bahir Dar University, Bahir Dar Institute of Technology (BiT), Ethiopia

^c Centre for Applied Data Science, School of Consumer Intelligence & Information Systems, University of Johannesburg, South Africa

ABSTRACT

This paper addresses the challenge of designing digital sharing platforms that align with the local sharing practices of SMEs in resource-constrained countries, with a specific focus on Ethiopian SMEs. The research utilises the elaborated action design research (EADR) methodology and begins with the diagnosis stage, employing thematic analysis with activity theory (AT) to gain an understanding of the local sharing practices. Results revealed elements such as activities, stakeholders, motives, resources, goals, actions, rules, and tools that inform the design requirements of the digital sharing platform. Building on the diagnosis stage, the design cycle established key principles and the architecture of 13 local sharing models that form the features of the platform. In the implementation cycle, the paper presented a specific instance of a "property rental package." Results indicated the positive acceptance of the digital platform. Factors such as technological compatibility, perceived direct benefit, ease of use, and perceived indirect benefit contributed to this acceptance. The study contributes to the existing literature by providing valuable insights on how to contextualise the design of digital platforms in the local sharing practice contexts. Additionally, it showcases the effective utilisation of the EADR methodology and activity theory.

Keywords Digital Sharing Platform, Contextualising designs, SMEs, Local Sharing Practices, Information System Design, Digital platform design

 $\textbf{Categories} \bullet \textbf{Information Systems} \sim \textbf{Computing platforms} \bullet \textit{General and reference} \sim \textit{Cross-computing tools and techniques, Design}$

Email

Sertse Abebe – sertse.abebe@bdu.edu.et, sertse26@gmail.com Hossana Twinomurinzi – hossanat@uj.ac.za

1 INTRODUCTION

Although digital platforms offer unprecedented opportunities for economic sharing and collaboration, the utilisation of these platforms remains low among SMEs operating in resourceconstrained countries (Bardhi & Eckhardt, 2012). Several factors contribute to this low adoption rate, including inadequate technological infrastructure (Myovella et al., 2020), lack of digital literacy and skills (Amornkitvikai et al., 2022), security and trust concerns (Ochinanwata & Ochinanwata, 2023), absence of contextually designed technology (Smidt & Jokonya,

Abebe, S., and Twinomurinzi, H. (2025). Contextualising Design: Aligning digital sharing economy platforms with local SMEs' sharing practices in resource-constrained countries. *South African Computer Journal* 37(1), 14–44. https://doi.org/10.18489/sacj.v37i1.18435

Copyright © the author(s); published under a Creative Commons NonCommercial 4.0 License (5) SACJ is a publication of SAICSIT. ISSN 1015-7999 (print) ISSN 2313-7835 (online)

2021), and a lack of design knowledge (Atinaf et al., 2023). Particularly, studies indicated a significant research gap concerning the contextualisation of digital sharing platform design concerning the local sharing practices of SMEs within a community's local business ecosystem (Pankomera & van Greunen, 2019). While many existing platforms have primarily been designed based on the business models of major tech companies in the Western world (K. S. Rahman & Thelen, 2019), it is important to note that SMEs in resource-constrained economies frequently participate in a range of local sharing practices deeply rooted in the community's culture (DiBella et al., 2022). Hence, it is crucial to examine the design of digital platforms by exploring the local sharing practices of SMEs.

This study specifically investigated the design of a digital sharing economy platform within the context of local sharing practices among SMEs in Ethiopia. The research employed an elaborated action design research (EADR) methodology, chosen for its suitability in studying the designs of socio-technical systems. The study makes a contribution to the discourse on information systems (IS) by exemplifying how the design of digital platforms can be contextualised within the context of local sharing practices. It identifies essential design aspects that need to be taken into consideration when designing digital platforms for specific contexts and showcases the practical application of activity theory (AT) in exploring and analysing activities from local sharing practices and comprehending the design requirements of IS artefacts.

Moving forward, the structure of the remaining paper is as follows: The subsequent section presents a comprehensive review of the existing literature. This is followed by an in-depth explanation of the methodology employed. The later section delves into the discussion of the study's findings. Finally, the paper concludes by highlighting the key contributions of the research and its implications for theory, practice, and policymaking.

2 LITERATURE REVIEW

This section presents a comprehensive literature review covering key areas such as the sharing economy (SE), the design of digital platforms for SMEs, the contextualisation of digital platform design, and activity theory (AT).

2.1 Sharing Economy

There is no universally agreed definition for the SE (Govindan et al., 2020). Studies often describe the SE as an activity synonymous with collaboration consumption (Belk, 2014). It indicates online sharing activities such as bartering, swapping, lending, trading, renting, and donating economic resources (Ranjbari et al., 2018). Many studies also associate the SE with sharing activities in innovation, redistribution, co-finance, and co-creation (Upadhyay et al., 2021). The term has also been used to explain crowd-based networks and the new forms of "gig" work that are encroaching on traditional employment (Chen et al., 2019).

From this study's perspective, the term is used to indicate activities related to collaboration consumption, the circular economy, crowdsourcing, crowdfunding, and open innovation activities. The circular economy refers to an economic model that favours product maintenance, reuse, and recycling. It is an effective way to tackle the scarcity of resources in the value chain and reduces the cost of materials, thereby promoting a sustainable environment (Peng et al.,

2014). Crowdsourcing is a problem-solving model that enables enterprises to outsource jobs and assignments to networked people (Tiberius & Hauptmeijer, 2021). Crowdfunding refers to a model for gathering funds from crowds to connect a particular project and investors (Köhler et al., 2022). Finally, open innovation is a framework that promotes the adoption and generation of new technologies (Achtenhagen et al., 2013).

2.2 Designing digital platforms for SMEs

Despite the pivotal role of digital platforms in enabling SMEs to restructure and enhance their value-creation processes, and the intricate link between SMEs' value-creation processes and social activities, especially in resource-constrained countries (Rumanyika et al., 2021), few studies have recognised the critical importance of addressing the specific challenges associated with designing digital platforms for SMEs in their local practices context.

In this regard, a paper that reported the design of "Machinga" showcased the design process of a mobile application that supports local street traders in their problem of limited market access (Ameller et al., 2015). The study employed a participatory design approach, involving stakeholders in the requirement gathering, design, and evaluation stages. Other studies have also explored the features of digital platform designs within industrial associations, outlining requirements for SMEs' internationalisation and the establishment of a collaboration partner marketplace (Hirota et al., 2022). Similarly, a separate study was conducted focusing on designing platform-based circular economies by exploring the business patterns of a circular economy model (de Reuver et al., 2018). While these studies provide valuable insights into contextualising IS artefacts, it is important to note that designing digital platforms goes beyond individual applications that were primarily taken into account by those studies. Designing digital platforms requires a thorough comprehension of the design aspects involved, as they are complex systems that interact with heterogeneous actors and typically integrate multiple applications.

In this context, a digital actor engagement platform was designed to demonstrate the contextual design of a digital platform that supports local businesses in the high street threads industry (Bartelheimer et al., 2023). Using the action design research (ADR) method, it investigated the engagement of actors in the street markets. This study addressed the local practices and actor networks in the street trade (Bonina et al., 2021). However, it lacks contextual relevance for designing digital sharing platforms in the context of resource-constrained economies. Therefore, further research is necessary to gain a comprehensive understanding of how to contextualise the design of digital platforms based on the local sharing practices of SMEs in resource-constrained economies.

2.3 Design aspects for contextualising the design of digital platforms

Contextualising the design of a digital platform with local practices is crucial for its success. However, to achieve effective contextualisation, it is first essential to have a comprehensive understanding of the core constructs of digital platforms (Wulfert et al., 2022). In this respect, various design aspects come into play, including the purpose (Choudhary et al., 2021; Eisape, 2022), boundaries (Gawer, 2021b), components (Spagnoletti et al., 2015), structures/organisations (Hou & Shi, 2021), and contexts (Janowski, 2015) of these platforms

The purpose of digital platforms plays a crucial role in shaping their design, functionality, and overall direction. By their purpose, digital platforms can be innovation or transactionoriented (Bonina et al., 2021). Achieving effective contextualisation in the design of digital platforms relies on obtaining a clear understanding of their purpose and its alignment with the motives and goals of the stakeholders. The boundary of platforms defines the scope of functionalities and services to be provided, as well as those that will be excluded (Gawer, 2021b). It is characterised by its scope, sides, and boundary resources (Gawer, 2021b). Defining the scope of the platform is essential for establishing the context of its operations and interactions. It is often defined by activities and resources the platform will manage (Gawer, 2021b). In this regard, it is necessary to examine the activities and resources that the platform will handle. Besides, platforms serve as a nexus that brings together two or more groups of actors and enables the exchange of value between them (Gastaldi et al., 2023). Understanding the characteristics and needs of the stakeholders involved is crucial to defining the boundaries of the platform (Gawer, 2021b). Moreover, contextualising the design of boundary resources is vital for defining the boundary of a digital platform. In this regard, gaining an understanding of the interactions within different activities in local practices can prove to be an effective approach for comprehending the design requirements of digital platforms and their boundary resources. Boundary resources are application interfaces (APIs) that facilitate interactions of several applications within the boundary of digital ecosystems (Gawer, 2021b).

Defining the components and organisation/structure of a digital platform is a also fundamental aspect of its design (Tura et al., 2017). It helps in shaping its architecture, functionalities, and overall user experience. Digital platforms often contain a set of stable cores, evolving peripherals, and interfacing components (de Reuver et al., 2018). While the core components of digital platforms serve as the pillars and provide generic services, the evolving peripheral components cater to the diverse and expanding functional needs of heterogeneous stakeholders (Spagnoletti et al., 2015). In this respect, it is common for digital platforms to have multiple interdependent or third-party components (Bartelheimer et al., 2023; Weiss et al., 2023). These components, closely interconnected with the actions of multiple actors involved in communication, collaboration, and collective action (Hein et al., 2019). Hence, designers need to understand the stakeholders' actions and interactions in creating effective platform experiences that align with objectives of different actors. By its structure, digital platforms are often seen as a digital ecosystem consisting of interconnected and interdependent components (Allen et al., 2021).

The context in which digital platforms operate is another aspect of the design of utmost importance when it comes to contextualising the design of their business processes, governance, and interfaces (Allen et al., 2021; Bartelheimer et al., 2023). This context encompasses the social, cultural, economic, and regulatory factors that shape the platform's development, operation and governance. Hence, designers must consider community structures, norms, regulations, cultural practices and tools to design platforms to the specific needs, preferences, and constraints of stakeholders.

In summary, contextualising the design of a digital platform for local practices requires considering various design aspects, including the platform's purpose, boundaries, components, structure/organisation, and contextual factors. To elicit the design requirements associated with these aspects, the design process should involve understanding the motives and goals of

local stakeholders, exploring the activities and resources involved, and analysing the actions and interactions of stakeholders within the local practices. Additionally, considering the social, cultural, linguistic, and regulatory context is crucial for tailoring the design of business processes and interfaces to suit the specific context. To effectively explore the elements, this study employed a structured approach to investigating activities in local sharing practices by utilising activity theory (AT) as a conceptual framework. AT offers a holistic perspective that enables the examination of purposeful human activities and their interactions within the sociocultural context. The following section presents a concise overview and discussion of activity theory (AT).

2.4 Activity Theory

AT is a theoretical framework for studying purposeful human activities (Kaptelinin & Nardi, 2017). AT represents human activities using an activity system diagram that includes a subject, objects, tools, community, norms/rules, and division of labour (DOL) (Engeström, 1993). Figure 1 shows the activity system diagram adopted from Engeström (1993). "Subject" refers to an "actor" who owns the activity. This paper uses the term actor instead of subject since the term actor is more familiar in information systems research. "Object" refers to the state of a thing (abstract, concrete, or people). This paper uses the term "resource" instead of object since resources are the central object of sharing activities. "Tool" is an instrument that mediates the interaction between an actor and resources. "Community" refers to the "stakeholders" in an activity. This paper uses the term stakeholders instead of community due to the familiarity of the term. "Rules/norms" refer to the community laws that govern an activity. Finally, "DOL/roles" refers to the distribution of roles among the stakeholders.



Figure 1: Activity system diagram^a

^{*a*} adopted from Engeström (1993)

AT has several principles that dictate the study of activities. While these principles generally emphasise the specification, design, and evaluation of technology to be accomplished within the context of an activity (Kaptelinin & Nardi, 2017) the "hierarchical principle" in particular provides insights to study platform design by breaking down activities to smaller units (Engeström, 1993).

It depicts an activity using a three-level hierarchical abstraction comprising activity, actions, and operations. Activity is an abstraction parallel to motives (Kaptelinin & Nardi, 2017). Actions are abstractions of an activity parallel to the goal of an activity. Operations are actions performed within conditions/contexts. They refer to routines in the lower hierarchy of activities (Kaptelinin & Nardi, 2017).

3 METHODOLOGY

This research used the EADR process model (Mullarkey & Hevner, 2018). EADR is a process model for action design research consisting of diagnosis, design, implementation, and evolution (Mullarkey & Hevner, 2018). Figure 2 shows the EADR processes of the study.



Figure 2: The EADR process and model^a

^{*a*} adopted from Mullarkey and Hevner (2018)

3.1 Diagnosis Cycle

The diagnosis cycle investigates the problem that needs to be addressed through practical design. In this research perspective, the diagnosis cycle examined SMEs' local sharing practices through thematic analysis. Ethiopian SMEs were used as a case study. Data was collected from 32 SMEs working in four cities in Ethiopia using semi-structured interviews. Interviews were conducted with SME owners or managers in key Ethiopian cities – Addis Ababa, Adama, Bahir Dar, and Gonder. Data collection sites covered a distance exceeding 800 kilometres. Table 1 outlines the diverse range of 18 product types across four sectors offered by these SMEs. The data was analysed in three stages using thematic analysis based on abductive reasoning. AT provided the lens to guide the thematic analysis. The first iteration investigated activities, stakeholders, resources, and motives in the local SMEs' sharing practices related to collaborative consumption, circular economy, crowdfunding, crowdsourcing, and open innovation. Based on the findings in the first iteration, the second iteration examined the goals and actions in the identified activities. In the third iteration, the analysis examined the rules, the DOL, and the tools used in the sharing practices.

Sector	City	Main product	Enterprise age	Academic status
Manufacturing	Adama	Leather related	5-10 years	Technical School
Manufacturing	Adama	Furniture	1-5 years	Technical School
Manufacturing	Addis Ababa	Weaving	10-15 years	10th grade
Manufacturing	Addis Ababa	Leather	10-15 years	University
Manufacturing	Bahir Dar	Household	5-10 years	Technical School
Manufacturing	Bahir Dar	Garment	5-10 years	Technical School
Manufacturing	Gonder	Garment	5-10 years	Technical School
Manufacturing	Gonder	Furniture	1-5 years	Technical School
Construction	Adama	Mining	1-5 years	10th grade
Construction	Adama	Material production	5-10 years	10th grade
Construction	Addis Ababa	Machine rent	1-5 years	University
Construction	Addis Ababa	Material production	1-5 years	Technical School
Construction	Bahir Dar	Machine rent	5-10 years	University
Construction	Bahir Dar	Material production	5-10 years	Technical School
Construction	Gonder	Material production	1-5 years	10th grade
Construction	Gonder	Machine rent	1-5 years	University
Service	Adama	Hotel	> 15 years	University
Service	Adama	Transport	1-5 years	10th grade
Service	Addis Ababa	Hotel	> 15 years	10th grade
Service	Addis Ababa	Laundry	1-5 years	University
Service	Bahir Dar	Hotel	5-10 years	University
Service	Bahir Dar	Transport	> 15 years	10th grade
Service	Gonder	Car rental	1-5 years	University
Service	Gonder	Freight	5-10 years	10th Grade
Trade	Adama	Electronics	5-10 years	Technical School
Trade	Adama	Stationery	5-10 years	Technical School
Trade	Addis Ababa	Electronics	1-5 years	Technical School
Trade	Addis Ababa	Pharmacy	1-5 years	University
Trade	Bahir Dar	Spice	5-10 years	Technical School
Trade	Bahir Dar	Spare part retails	1-5 years	Technical School
Trade	Gonder	Cleaning material	1-5 years	University
Trade	Gonder	Stationery	1-5 years	Technical School

Table 1: Demography of respondents and SMEs

3.1.1 Activities, Stakeholders, Resources, and Motives

Results of the first iteration show the SMEs' engagement in local sharing practices related to collaborative consumption (CC), circular economy (CE), crowdfunding (CF), and crowd-sourcing (CS) activities. The findings did not show the SMEs' engagement in open innovation (OI) activities.

Private and public resources were the main drivers that attracted SMEs to the sharing activities. instance, the SMEs' major sharing activities were related to accessing private resources such as equipment, services, raw materials, and consumer products. Government support was also one aspect of the SMEs' sharing. Sharing practices were mainly done to ensure profitable businesses, minimise costs, obtain alternative finances, obtain access to raw materials, start businesses, share investment costs, and access government support. Suppliers, consumers, complementary asset providers (CAPs), and regulators were the main stakeholders in the sharing activities. Figure 3 shows the SMEs' activities, actors, stakeholders, resources, and motives in their local sharing practices.



Figure 3: Activities, actors, stakeholders, resources and motives in SME's local sharing practices

3.1.2 Goals and Actions in the SMEs' Sharing Practices

Actions are sequences of tasks (goal-oriented activities) taken to achieve a higher-level motiveoriented activity (Kaptelinin & Nardi, 2017). In the second iteration of the thematic analysis, the study explored the goals of SMEs in their sharing practices and the corresponding actions taken by these SMEs to achieve those goals. Results show fourteen major goals that encompassed diverse areas such as purchasing services, products, and product services; selling services, products, and product services; accessing government support; engaging in equipment swapping with private owners; finance collaborations (peer-to-peer); joint purchasing initiatives; participating in joint investments; participating in savings and credit unions; buying used and leftover products; selling second-hand items; hiring permanent workers; hiring temporary workers per time; hiring workers on a result-based basis; and engaging in subcontract work.

Furthermore, the thematic analysis has revealed a variety of actions related to the identified goals in the context of SMEs' local sharing practices. For goals related to purchasing, major actions were preparing specifications, searching for resources, negotiating, providing price quotations, placing orders, validating deliveries, and settling payments. On the other hand, for goals related to selling, actions were preparing and promoting a catalogue, managing customer orders, delivering products/services, collecting payments, and handling feedback. Engaging in financial collaborations entails initiating and promoting collaborations, registering members, collecting and disbursing funds, auditing collaborations, and completing the process. Saving and credit activities encompass promoting services, registering members, collecting savings, managing loans and payments, auditing collaborations, and maintaining membership. Joint investment activities involve initiating collaborations, registering collaborators, selling shares, facilitating share exchanges, settling payments, auditing collaborations, and managing dividends. Joint lease and procurement activities consist of initiating collaborations, registering collaborators, collecting funds, purchasing products, settling payments, distributing products, and auditing collaborations. Crowdsourcing activities revolve around hiring workers and subcontracting work. Actions in hiring workers include identifying human resource needs, selecting workers, negotiating terms, assigning jobs, validating completed work, settling payments, and managing employee information. Subcontracting work involves searching for providers, negotiating terms, accepting work orders, handing over assignments, and settling payments. Lastly, circular economy practices encompass buying and selling second-hand products. Buying actions involve identifying needs, finding products, settling payments, and maintaining relationships. Goals related to selling second-hand products include specifying and promoting products, managing customer orders, delivering products/services, collecting payments, and handling feedback. These actions provide a comprehensive overview of the steps involved in achieving the identified goals within SMEs' local sharing practices concerning the SE.

3.1.3 Rules, DOL, and Tools

According to AT, rules, DOL, and tools are the three elements that determine contexts in an activity (Engeström, 1993). Findings in the third iteration of the thematic analysis showed the existence of several rules/norms associated with:

- i. trustful transactions and collaboration;
- ii. payment accuracy, appropriateness, and timeliness;
- iii. deliveries' quality, timeliness and appropriateness;
- iv. participants' duties;
- v. refunding for undelivered services/products; and
- vi. many other business domain-specific rules concerning SMEs' sharing practices.

Findings have also shown the distribution of tasks among stakeholders in their local sharing practices. The result shows the role of SMEs as consumers, providers, collaborators, brokers, and employers. It also shows the government's role as a provider, consumer, and regulator. In addition, they show the non-governmental institutions' roles as providers and consumers. The financial institutions have a role in facilitating payments among the sharing actors.

Tools mediate human activities. AT classifies tools into technical (physical) and psychological (Kaptelinin & Nardi, 1997). Psychological tools are instruments such as languages and signs. Physical tools are external tools such as technological artefacts (Kaptelinin & Nardi, 1997). Results show the frequent utilisation of local languages and technical devices such as mobile phones in the SMEs' sharing practices.

3.2 Design Cycle

The design cycle was the second stage of the EADR. It examines the design principles, design features, and implementation methods (Mullarkey & Hevner, 2018). The major design activities concerning the design of the digital sharing economy platform were eliciting contextual design principles and proposing design features, and implementation methods.

3.2.1 Contextual Design Principles

The contextual design principles are formulated by taking into account two crucial factors: first, the objective of customising the design of a digital platform to suit the specific context of SMEs local sharing practices, and second, designing based on a comprehensive understanding of design aspects and core constructs that are inherent to digital platforms. The contextual design principles are:

Designing for a purpose Designing for specific purposes has been emphasised in multiple studies (Bonina et al., 2021). In this aspect, it is crucial to contextualise the purpose of the digital platform in the SMEs' motives and goals for their local sharing activities.

Design with boundary Designing with defined boundaries and interfaces is one aspect of designing digital platforms (Daradkeh, 2023; Gawer, 2021a). It is important to establish clear boundaries by investigating activities, resources, and stakeholders of the design of digital platforms.

Designing the components The studies emphasise the need to design both the core functionalities and peripheral components of a digital platform through the analysis of the actions of stakeholders (Bonina et al., 2021).

Designing for the context Context forms a critical facet in the design of various technologies. Its essence lies in ensuring that new designs seamlessly integrate into their context by consolidating data about social structure, cultural norms and environmental contexts (Murer et al., 2015; Stamps, 2014).

Design as an ecosystem IT emphasises the interconnectedness and interdependencies of various components of a platform (Engert et al., 2023).

Design for modularity and reusing This principle advocates for the design of modular components that facilitate the reuse of functionalities (Dai, 2023; Naik et al., 2020). Reusability enhances efficiency, reduces development time, and enables the adaptation of the platform to evolving needs.

3.2.2 Features of the Digital Sharing Economy Platform

The proposed design described the features of the digital sharing economy platform by its purpose, boundaries, components, structure/organisation, and context.

Purpose The outcomes of the diagnosis cycle have revealed the motives driving local sharing activities among SMEs. Furthermore, the results have identified 14 specific goals that SMEs strive to accomplish through their local sharing practices. Building upon these findings, the purpose of the digital sharing economy platform has been defined in its design. Consequently, the platform's design was customised to meet the unique needs of SMEs, including initiating new businesses, streamlining transactions, enhancing access to raw materials, providing alternative finance options like peer-to-peer lending, promoting cost reduction through shared resources and bulk purchasing, facilitating collaborative investment opportunities, and connecting SMEs with valuable government support programs and resources.

Boundaries The digital sharing economy platform's boundary was set by defining its scope, sides, and boundary resources. The platform's scope was determined based on the knowledge acquired from the diagnosis cycle, which examined the goal-oriented activities and resources of SMEs involved in local sharing activities. Results in the diagnosis cycle showed 14 goal-oriented activities of the SMEs' local sharing activities. Moreover, they revealed varied types of products and services as primary resources of the SMEs' local sharing practices. Based on the information, initially, the design mapped the goal-oriented activities with six packages. Figure 4 shows platform packages mapped from the SMEs' goal-oriented activities.



Figure 4: Sharing economy platform packages elicited from the SME's activities

Next, the design used knowledge about resources to define additional compartments of the packages. Accordingly, 13 sub-packages are proposed within the six main packages. The sides are modelled by analysing the DOL mong stakeholders. Results in the diagnosis cycle showed that SMEs, government, individuals, non-governmental organisations (NGOs), complementary asset providers (CAPs), and Brokers were the main stakeholders of the SMEs' local sharing practices. Figure 5 shows the boundary of the digital platform in terms of the activities, resources, and sides.



Figure 5: The boundary of the digital platform

The boundary resources are modelled by assessing the interaction and interdependence of various activities and actions. However, these resources are further refined during the component modelling stage by analysing the relationships and dependencies among different actions.

Components and Modules The design of components is influenced by goals and actions associated with local sharing practices. Findings on goals and actions have played a crucial role in determining the requirements for the platform's components. By understanding the actions undertaken by SMEs to achieve their key objectives, components were tailored to enable stakeholders to perform those actions. Furthermore, a modular design approach is recommended by several researchers (Dai, 2023). However, there is a lack of specific guidance on how to break down packages into modular components. This study addressed this gap by defining modules through the analysis of interrelated actions among actors involved in collaboration

processes. Collaboration is a dynamic process that encompasses various phases, including initiation, formation, collaboration, and finalisation (Twinomurinzi & Ayalew, 2022).

To design modules within each solution package, the study examines the actions performed by actors in each stage of the collaboration process. By analysing the interdependencies and relationships between these actions, the design of modular components within the packages is determined. For instance, in the PROPERTY-RENTAL activity, creating a rental property, listening to rental requests, agreeing on terms, receiving payments, managing property delivery, managing the return of the property, and viewing feedback are actions of the providers. In the other dimension, searching for rental property, creating rental requests, agreeing on terms, settling payments, validating and accepting properties, returning properties, and rating or complaining about the service delivered are actions of the consumers. Thus, the design process first organised the providers' and consumers' actions in the collaboration lifecycles diagram. Next, potential modules are abstracted by assessing related actions in each phase.

Figure 6 shows the analysis of modules based on actors' related actions in the collaboration stages. Table 2 shows a list of modules under the packages of the digital sharing economy platform.



Figure 6: Example of module definition from property rental activity

Contexts Contextualising the digital platform largely depends on the extent in which designs incorporats local rules (norms), processes, structures, and interaction tools (such as languages and cultural tools). In this design context, each module is crafted with careful attention to integrating local business processes, community power structures, norms, and rules identified during the diagnosis stage that are prominent in local sharing practices. For instance, in property rental activities, local rules require the tenant to pay a certain amount of down-payments, known locally as 'qabd', after both parties have reached an agreement on the rental terms. This payment is typically considered a guarantee to uphold the agreement. Therefore, the "Agreement Management Module" should be designed to incorporate this and other similar rule. In general, designing the digital platform modules requires consideration of and incorporation of such local rules, which is an essential aspect of contextual design. Moreover, the

Package	Sub-Package	Modules
Product Service Market	Product Market	Product Registration, Search, Negotiation and Chatting, Order Management, Agreement Management, Payment Management, Sales Management, Delivery Management, Feedback and Compliant Management
	Service Market	Service Registration, Search, Negotiation Management, Booking, Agreement Management, Payment Management, Service Delivery Management, Feedback and Compliant Management
	Property Rental	Asset Registration, Rental Asset Search, Rental Request, Agreement Management, Rental Management, Payment Management, Feedback and Compliant Management
Government Support	Government's Finance Provision	Announcement Management, Application and Selection Management, Financial Provision Management, Payment Management, Follow-up Management
	Government's Workspaces Provision	Announcement Management, Application and Selection Management, Financial provision Management, Payment Management, Follow-up Management
Supply Collaboration	Machine and Equip- ment Swap	Equipment Registration, Search Management, Agreement Management, Swapping Management, Equipment Returning Management, Follow-up Management
	Joint Purchasing	Create Activity, Activity Announcement, Create Group, Payment Management, Purchasing Management, Distribution Management, Follow-up Management
Financial Collaboration	Peer-to-Peer Finance (EQUIB)	Initiate and Announce EQUIB, Application and Selection, Role Assignment, Collect Contribution, Identify Fund Recipients, Reimburse Funds, Payment, Audit EQUIB, Claim Settlement, Settle Dividend
	Saving and Credit Services	Initiate and Promote the Credit Union, Membership application and selection, Periodic saving management, Credit processes Management, Audit the credit unions, Resignation Management, Report Management
Investment Collaboration	Joint Investment (AXIONs)	Investment Initiation and Promotion Management, Shareholders Registra- tion, Shareholders Assembly and Signature, Share Exchanges, Payments, Audit Management, Dividends Management, Report Management
	Employment	CV Management, Application Management, Talent Selection Management, Recruitment and Contract Management, Job Assignment Management, Payment Management
	Sub-Contract Work	Contract Work Registration Management, Application and Selection, Job Assignment Management, Contract Management, Payment Management

Table 2: List of Packages, Sub-Packages, and Modules

design took into account the importance of user interface (UI) design by considering the physical tools, such as users' experience with digital devices, as well as the psychological tools, such as language usage in local sharing practices.

Structures and Organisation The proposed design suggested a platform architecture that emphasises interoperability between the core components and the identified packages. Figure 7 shows the proposed architecture of the digital sharing platform. The core components were modelled by including user management, activity tracking and log management, secure authentication and authorisation mechanisms, communication and messaging features, and administrative tools.

The design organised each package as an independent layered collection of components containing data storage, business, and presentation logic. Separated databases are proposed for the platform core as well as for each solution package. Separating databases allows each package to perform autonomous operations with distinct data (Deryabin et al., 2023; Laigner



Figure 7: Architecture of the digital sharing platform

et al., 2021). Moreover, the design emphasised the inclusion of boundary resources (APIs) within both the core of the platform and each package. These boundary resources serve as interfaces that allow the core and packages to expose their functionalities to other components within the platform ecosystem. APIs may be internal or external to the platform. Internal APIs facilitate the interfacing between modular components or application packages in the platform core and other internally developed solution packages. External APIs often interface the platform with external applications. The presentation layers contain the user interfaces.

3.3 Implementation Cycle

In the implementation cycle of the EADR process, the emphasis is on instantiating artefacts. In this research, an incremental approach was adopted for the implementation process, allowing for ongoing improvement and refinement.

During the first iteration, the platform core modules were implemented. To accomplish this, the research utilised an open-source platform framework called DNN_Platform_9.12.1 Community Edition. This framework was selected for its inclusion of essential platform core modules that handle crucial tasks such as log management, message management, user account management, role management, and security. Furthermore, it provided the flexibility to develop modules as independent projects using Entity Framework 6.02 and MSSQL Express 8th edition or above. In addition to these basic modules, the implementation stage incorporated geographic locations, enterprises, activities, and resource modules as the core components of the digital platform.

Moving to the second iteration, the focus shifted to implementing the APIs for the platform
core. For the sake of simplicity, APIs are implemented as RESTful services. In the third iteration, this study focused on implementing modules that facilitate actions within SMEs' local sharing practices. Due to limitations in time and funding, it was not feasible to implement all of the solution packages. Consequently, the study prioritised the implementation of the 'property rental' package, as it is closely associated with SE and collaborative consumption activities. Figure 8 shows a snapshot of the home page for rental management packages.



Figure 8: Property rental home page

The rental market solution contains 'asset registration', 'rental-asset-search', 'rental-requests', 'rental-agreement', 'rent-registration', payment, 'reminder', and 'feedbacks' modules. The 'asset registration' module allows providers (individuals or enterprises) to post, view, edit, and delete rental assets. 'Rental asset search' modules allow consumers to view details, search, and, 'create cart lists'. The module allows consumers to see the posted item based on the enterprise. The 'rental request' module allows consumers to create, view, modify, and delete rental agreement' module permits providers to approve or reject consumers' requests. The 'rental agreement' module permits providers to create, modify, delete, view, and print agreements. It also allows consumers to create, transfer, and view payments. The 'rent registration' module allows the providers to create, delete, modify, or view rental transaction information. The 'reminders' module allows providers to post, modify, or delete different reminders to consumers. It also allows the consumer to read the reminders. The 'feedback' module allows the consumer to read the reminders. The 'feedback' module allows the consumer to read the reminders. The 'feedback' module allows the consumer to create, delete, modify, and delete feedback. It also allows providers to view the feedback of the consumer.

3.4 Evolution Cycle

The final stage of the EADR focuses on analysing the evolution of the artefact over time, considering changes in the problem environment and the iterative development of the artefact solution (Mullarkey & Hevner, 2018). Typically, this stage of the process spans a long-term project. However, in this study, the initial acceptance of the digital sharing platform was included as part of the evolution cycle.

The initial acceptance of the designed digital sharing economy platform is evaluated based on users' intention to use the services running on the digital sharing platforms.

Researchers have so far utilised theories such as the Technology Acceptance Model (TAM) (Kamal et al., 2020), the Theory of Planned Behaviour (TPB) (Shneor & Munim, 2019), the Unified Theory of Acceptance and Use of Technology (UTAUT) (S. A. Rahman et al., 2019), Diffusion of Innovation (DOI) (Attié & Meyer-Waarden, 2022), and the Technology, Organisation, and Environment (TOE) frameworks (Kouhizadeh et al., 2021) to evaluate technology acceptance related research. This research selected the TOE framework since it is suitable for evaluating technology acceptance in organisations including SMEs and allows us to see environmental as well as technological factors (Ghobakhloo & Ching, 2019).

3.4.1 Conceptual Model and Hypothesis

The TOE is a technology acceptance assessment framework that is based on organisationallevel theory (Bokolo & Petersen, 2022). It incorporates technological, organisational, and environmental contexts as factors to evaluate the acceptance of technology in organisations. This study applied the TOE framework to test nine different factors under three separate contexts (technological, organisational, and environmental) in terms of their direct effect on SMEs' intention to use the designed digital sharing economy platform for their sharing activities. Figure 9 shows the property rental system's initial acceptance evaluation model.



Figure 9: The 'property rental system's' initial acceptance evaluation model

The technological context indicates factors concerning the features of technologies that affect technology acceptance (Bokolo & Petersen, 2022). The complexity, compatibility, perceived direct benefit, and perceived indirect benefit are factors that affect technology acceptance (Abed, 2020). The perceived direct benefit indicates the potential of the platform to enhance operational efficiency and reduce operational costs. The perceived indirect benefits

indicate the anticipated improvements by the user in expanding relations and customer bases. Ease of use refers to the degree to which the user estimates being able to use the technology without much effort (Abed, 2020). Compatibility indicates the degree to which the technology is aligned with the values, rules, processes, cultures, and languages (Abed, 2020).

In the context of technological factors, four hypotheses were formulated regarding the designed digital sharing economy platform. The assumptions were as follows:

- H1 The perceived direct benefits observed from the newly designed digital sharing platform have positively influenced individuals to use the technology for their local sharing activities.
- **H2** The perceived indirect benefits observed from the digital platform have positively influenced individuals to use the technology for their local sharing activities.
- **H3** The ease of use of the digital platform has positively influenced individuals to use the technology for their local sharing activities.
- **H4** The compatibility of the digital platform has positively influenced individuals to use the technology for their local sharing activities.

The organisational context indicates multiple institutional factors that affect the users' technology use intention (Abed, 2020). This study considered organisational factors that affect the use intentions of the designed digital sharing economy platform. In terms of this, the study produced two hypotheses as follows:

- **H5** The resources available in the SMEs have positively influenced the users' use intention of the designed sharing platform.
- **H6** The SMEs' previous digital technology experience has positively influenced the users' use intention of the designed sharing platform.

Lastly, the environmental context indicates factors such as consumer readiness, competitor pressure, and government support that can affect users' intention to use the designed digital sharing economy platform. In this respect, the study produced three hypotheses:

- **H7** The rent consumers' readiness and experiences on digital platforms have positively influenced the users' use intention of the designed sharing platform.
- **H8** The competition from other SMEs' digital platforms has positively influenced the users' use intention of the designed sharing platform.
- **H9** Government pressure has positively influenced the users' use intention of the designed sharing platform.

3.4.2 Sampling and Data Collection

Data was collected from a sample of 123 individuals, including SME owners or employees and rental consumers who had experience in providing or consuming rental properties. The participants were conveniently selected from the town of Bahir Dar, considering its proximity to the researchers and the ease of introducing them to the digital sharing economy platform. To gather the necessary information, standardised questionnaires comprising demographic questions and 28 5-Likert scale questions were prepared. Before completing the questionnaires, participants were given 20 days to explore and utilise the rental market application that operates on the digital sharing economy platform.

3.4.3 Participants

The results indicate that the majority of participants were male, aged between 22 and 30 years, with technical education backgrounds and business experience ranging from one to five years. The respondents were equally distributed across various sectors including trade, construction, manufacturing, and services. For further details, please refer to Figure 10, which presents the demographic characteristics of the participants.



Figure 10: Demographic characteristics of the participants

3.4.4 Data Analysis

The study used a structural equation modelling (SEM) technique (Gupta & Shankar, 2022). The study specifically applied the partial least squares SEM (PLS-SEM) technique since it is suitable for research with a small sample size (Dash & Paul, 2021). It is suggested that the minimum number of participants in the study should be ten times the number of indicators associated with the most complex constructs or the number of antecedent constructs linked to an endogenous construct (Kock & Hadaya, 2016). In this study, the number of indicators

for the complex constructs and the number of antecedent constructs linking to an endogenous construct was six and nine respectively. Hence, the sample size of 123 was acceptable in terms of using the PLS-SEM technique (Hair et al., 2020). Analysis in PLS-SEM consists of measurement and structural modelling tasks (Hair et al., 2020).

3.4.5 Measurement Model

The measurement model examins the relationship between variables and their measures (Hair et al., 2020). It assesses the reliability and validity of instruments (Legate et al., 2021). Reliability indicates the consistency of the interpretation of the instrument's questions. The study assessed the internal consistency of the measurement using Cronbach- α and Composite Reliability (CR) scores (Legate et al., 2021). The minimum Cronbach- α score found in the study was 0.799, which is above the accepted value (0.5) (Vaske et al., 2016). CR measures the internal consistency of indicator loading on the latent variable (Vaske et al., 2016). The result showed a minimum CR (ρ -a) value of 0.800, which is also above the accepted value (0.7) (Vaske et al., 2016).

Convergent validity measures the closeness of the new scale with other variables and other measures of the same variable. Convergent validity is often measured by its factor loading and average variance extracted (AVE) score. The results show a minimum of 0.783 AVE score for a construct, that is above the threshold of the accepted score (0.5). The results also show that all outer factor loading scores are greater than 0.823 which is above the minimum accepted value of 0.7. The discriminant validity indicates the degree to which items differentiate a construct from other constructs and measure distinct concepts (Hwang et al., 2023). This research used the Hetrotrait–Monotrait ratio (HTM) matrix score to assess the discriminant values. The results show the maximum HTMT score value of 0.844, which is valid and less than the maximum threshold value of 0.85 (Hwang et al., 2023). Table 3 shows the reliability measures of Cronbach- α , CR, and AVE.

Contexts Type	Variables	Code	Items	Cronbach's α	CR	AVE
Technological	Perceived Direct Benefit	PDBft	5	0.938	0.949	0.787
Technological	Perceived Indirect Benefit	PIBft	2	0.874	0.940	0.887
Technological	Ease of Use	EseUs	5	0.933	0.948	0.783
Technological	Compatibility	Cmpty	4	0.912	0.938	0.791
Organisational	Enterprise Resources	ErRsrs	2	0.923	0.942	0.844
Organisational	Enterprise Experiences	ErExps	3	0.880	0.942	0.891
Environmental	Consumer Readiness	CmRds	2	0.800	0.909	0.833
Environmental	Competitor Pressure	CptPrss	2	0.904	0.952	0.908
Environmental	Government Pressure	GvPrss	2	0.921	0.962	0.926
Dependent Variable	Use Intentions	UseInt	2	0.914	0.958	0.919

Table 3: Reliabilit	y measures	according t	o Cronbach's	α,	CR,	AVE
---------------------	------------	-------------	--------------	----	-----	-----

3.4.6 Structural model

The structural model indicates the relationship between the independent and dependent variables. In this regard, the study first conducted hypothesis testing. The hypothesis testing estimated the significance of the relationships between the independent variables and dependent variables in each assumption using a critical t-value from the first-round bootstrap. Table 4 shows the first round of bootstrap results. The results show unacceptable relationships (>0.05) for two assumptions (A8 (0.173) and A9 (0.138) in H8 and H9). The structure was then remodelled by removing the two assumptions. The second-round bootstrap result shows significant relationships between the independent and the dependent variables in the remaining assumptions. Validity and reliability measures were again done for the new model and were found to be acceptable. An analysis of the remodelled structure was then done.

Paths	T statistics O/STDEV	P Values	Status
$CmRds \rightarrow UseInt$	2.879	0.004	Accepted
Cmpty→UseInt	3.443	0.001	Accepted
$ErExps \rightarrow UseInt$	2.950	0.003	Accepted
$ErRsrs \rightarrow UseInt$	2.262	0.024	Accepted
${\sf EseUse}{\rightarrow}{\sf UseInt}$	3.353	0.001	Accepted
$PDBft{ o}UseInt$	2.218	0.027	Accepted
${\sf PIBft}{\rightarrow}{\sf UseInt}$	2.442	0.015	Accepted
$CptPrss \rightarrow UseInt$	1.485	0.138	Rejected
${\sf GvPrss}{\rightarrow}{\sf UseInt}$	1.361	0.173	Rejected

Table 4: First round bootstrap results

The coefficient of the determinant R^2 indicates the amount of variance in a dependent variable explained by the independent variable. The closer the R^2 value to one, the better the model's ability to predict the dependent variable. The findings show R^2 values of 0.879 for the dependent variable related to variables in the remaining assumptions. Similarly, the results of the path coefficient (β) values of each construct show the positive relationship between the independent and dependent variables. Table 5 shows the results of the structural model analysis. Figure 11 shows the structural model of the study with its R^2 and path coefficient (β) values.

3.4.7 Initial User Acceptance Assessment Results

Results show a positive association between the selected factors and the users' digital platform use intentions. Comparatively, technological factors influenced users to develop intentions to use the designed digital sharing economy platform for their rental activities. In particular, the compatibility of the digital platform (β =0.225, f^2 =205), the perceived direct benefit observed from the digital platform (β =0.207, f^2 =132), and the ease of use of the platform (β =0.186, f^2 =109) significantly influenced users to develop an intention to use the designed

Paths	Sample Mean	Standard Deviation	T statistics	Р	Path coefficient	f^2	Correlation
	Μ	STDEV	O STDEV $ $		β		
CmrRds ightarrow UseInt	0.119	0.044	2.766	0.006	0.121	0.084	0.581
$Cmpty \to UseInt$	0.225	0.069	3.279	0.001	0.225	0.205	0.770
$ErExps \to UseInt$	0.155	0.061	2.535	0.011	0.154	0.107	0.717
$ErRsrs \to UseInt$	0.183	0.061	2.996	0.003	0.184	0.153	0.706
$EseUs \to UseInt$	0.191	0.058	3.184	0.001	0.186	0.109	0.782
$PDBft \to UseInt$	0.208	0.079	2.614	0.009	0.207	0.132	0.790
$PIBft \to UseInt$	0.148	0.042	3.634	0.000	0.152	0.140	0.565

Table 5: T Statistics, P Values, Path Coefficients (β)



Figure 11: Structural model of the study with its R^2 and path coefficient variables, β .

digital sharing economy platform. Moreover, the indirect perceived benefit observed from the platform positively influenced ($\beta = 0.152$, $f^2 = 140$) the users' use intentions. Results also show the positive influence of organisational contexts. Enterprise resources ($\beta = 0.184$, $f^2 = 153$) and the users' previous experience ($\beta = 0.154$, $f^2 = 107$) show a positive association with the users' digital platform use intentions. Environmental contexts, however, showed little influence on the users' digital platform use intentions.

4 DISCUSSION

The goal of this study was to explore the design of a digital sharing economy platform for SMEs in resource-constrained countries. It specifically examined how these platforms can be tailored to meet the needs of SMEs based on their local sharing practices. While previous

research has highlighted the importance of the sharing economy and associated digital platforms in promoting collaboration and resource optimisation (Belk, 2014), this study addressed a significant gap: the need to adapt the sharing economy concept to the unique cultural and economic conditions of SMEs in developing countries. By analysing local sharing practices, this research connected these practices with the design of digital sharing platforms, offering new insights from Ethiopian SMEs to the literature on the sharing economy.

The study employed the Elaborated Action Design Research (EADR) methodology, which is effective for technology design (Mullarkey & Hevner, 2018). In the diagnosis phase of EADR, the research examined the activities involved in local sharing practices among SMEs, using Activity Theory (AT) as a guiding framework. Although AT is recognised for understanding social and technical systems (Kaptelinin & Nardi, 2017), this study uniquely utilised it to explore local sharing practices within Ethiopian SMEs. This approach can enrich existing literature by demonstrating how established theories like AT can enhance our understanding of activities in specific contexts, ultimately aiding in the design of digital platforms that fit local needs. This approach could enhance conventional participatory design methods, which often rely on inductive idea gathering through brainstorming without a solid theoretical foundation. By incorporating proven theories such as Activity Theory (AT) to guide the process, it becomes a more effective and structured approach.

Applying AT had several benefits. Firstly, it facilitated the identification of sharing practices unique to the local community, such as EQUIB (a traditional peer-to-peer collaboration practice), AXION (a local practice of joint investments), second-hand product exchanges, rental activities, labor market interactions, and savings and credit activities. Secondly, the theory provided valuable insights into the structural elements of these local practices, encompassing activities, actors, stakeholders, motives, resources, goals, actions, rules, and tools. Thirdly, AT emphasises that the specification, design, and evaluation of technology should occur within the context of an activity (Kaptelinin & Nardi, 2017), allowing for the direct elicitation of design requirements from the results of the activity analysis.

In the design cycle, the researchers elicited contextual design principles and defined the features of the proposed digital sharing economy platform. These principles were based on the local sharing practices of SMEs and the unique aspects of designing digital platforms. This approach allows the design process to consider both the users' context and the specific attributes of the technology. Assessing these contextual design principles helped to identify SMEs' requirements, which can then be translated into automated solutions within the digital platform. Investigating key design aspects – such as purpose, boundaries, components, and context – was also essential for understanding the constructs of the technology concerning the unique nature of digital platforms. While contextual design is typically viewed as user-centered, focusing on user interests, needs, and scenarios, this study expands the concept by incorporating the specific nature of technology and best practices in its design. These considerations align with existing literature that conceptualises digital platform technology in terms of its purpose, boundaries, components, and modularity (de Reuver et al., 2018).

Beyond the design principles, the characterisation of the digital platform was developed to support local sharing activities. Results from the diagnosis cycle informed the design requirements, including the platform's purpose, boundaries, components, structures, and contexts. Understanding the motives and goals of local sharing practices was crucial for aligning the platform's purposes with these factors (Bonina et al., 2021). Knowledge about the types of sharing activities, resources, and stakeholders helped define the platform's scope and boundaries, contextualising its design (Gawer, 2021b). The findings regarding goals and actions were instrumental in eliciting requirements for the components, which are essential for facilitating collaboration and communication among stakeholders (Spagnoletti et al., 2015). In this context, the identification of 13 local sharing models provides a practical framework that has been largely absent in prior research, which often presents theoretical constructs without actionable insights. Additionally, the analysis of rules, tools, and norms was significant in contextualising the design of components that provide functionalities aligned with each stake-holder's actions.

Beyond the design principles, the characterisation of the digital platform's design was done in a way that supports the local sharing activities. Results obtained from the diagnosis cycle were utilised to inform the design requirements of the digital sharing economy platform, including its purpose, boundaries, components, structures, and contexts. Understanding the motives and goals in the local sharing practices was crucial for making informed decisions on the purposes of the digital sharing economy platform by aligning with these motives and goals. The knowledge about the type of sharing activities, resources, and stakeholders was crucial in defining the scope and sides of the platform, which helped to contextualise the boundary of the digital platform. The findings from the diagnosis cycle regarding goals and actions were instrumental in eliciting requirements for the components, as these components are crucial in offering functionalities closely linked to stakeholders' actions, facilitating collaboration, communication, and collective actions. In this aspect, the identification of 13 local sharing models offers a practical framework that has been largely absent in prior research, which often presents theoretical constructs. Furthermore, the analysis of rules, DOL, and tools played a significant role in contextualising the design of components that provide functionalities aligned with each stakeholder's actions.

During the implementation cycle, an incremental approach was adopted. The core modules of the platform were implemented in the first iteration, followed by the implementation of APIs in the second iteration. In the third and final iteration, modular components and APIs responsible for specific tasks related to SMEs' sharing practices were implemented.

The evolution cycle of the study focused on empirically assessing the initial acceptance of the designed digital sharing economy platform. This assessment was based on individuals' intentions to use the platform, specifically targeting the rental activity management solution integrated within its services. At this stage, the study adopted a Technology, Organisational, and Environmental (TOE) framework, which is typically used to assess technology acceptance within organisations. While the TOE framework is appropriate for evaluating technology acceptance, this study offers new insights by embedding it within the EADR process model to empirically assess the adoption of new designs. The results indicated that both the technological and organisational contexts positively influenced SMEs' willingness to use the digital platform.

In summary, the study's findings provided valuable insights into contextualising the design of digital sharing economy platforms through the EADR process model. This process model proved instrumental in conducting comprehensive investigations and incorporating diverse ideas, theories, and practices that were relevant to the research problem at hand.

5 LIMITATIONS

The study does have certain limitations that should be acknowledged. Firstly, the focus on understanding design requirements solely based on existing SMEs' sharing practices may have restricted the exploration of potential new sharing models derived from international best practices. This limitation suggests that there could be missed opportunities to introduce innovative sharing approaches that could benefit SMEs in resource-constrained countries.

Additionally, it is important to recognise the limitations of solely assessing the acceptance of the designed digital platform based on users' use intention. By solely relying on this measure, the study overlooks the indirect network effects that are often associated with digital platforms. These effects, such as the number of users on the platform, play a crucial role in shaping its overall quality and success. Therefore, it is essential to consider these indirect network effects when evaluating the acceptance and effectiveness of the designed digital platform. By incorporating a holistic assessment approach, researchers can gain a more comprehensive understanding of the platform's impact and potential for success.

Based on the limitations identified, the study recommends expanding future research to explore innovative sharing models from international best practices that could benefit SMEs in resource-constrained countries. It also highlights the importance of incorporating indirect network effects when assessing digital platform acceptance, advocating for a holistic evaluation that includes both qualitative and quantitative measures. Finally, engaging diverse stakeholders in the design process is recommended, as this can enhance the platform's relevance and usability, ultimately improving its effectiveness for SMEs.

6 CONCLUSION

This study explored the design of a digital sharing economy platform tailored for SMEs in resource-constrained countries, specifically focusing on Ethiopian local sharing practices. Employing the Elaborated Action Design Research (EADR) methodology, the research encompassed diagnosis, design, implementation, and evolution cycles. The findings revealed critical activities and goals in local sharing, leading to the development of a platform architecture that incorporates 13 innovative sharing models. The implementation phase demonstrated practical realisations of the platform, while the evolution phase indicated positive initial acceptance driven by technological and organisational factors. Overall, this study contributes a replicable methodology for designing context-specific digital sharing platforms and highlights policy implications that can guide interventions to support the digital transformation of SMEs, ultimately enhancing their competitiveness in the digital economy.

7 CONTRIBUTIONS

In practical terms, this study presents a replicable methodology for designing digital sharing platforms that can be tailored to specific local contexts. A key focus is placed on identifying core design aspects such as purpose, boundaries, components, structure/organisation, and contexts, which can be adapted to fit local requirements. By utilising Activity Theory (AT),

the study effectively investigates and understands local sharing practices, providing valuable insights to inform the design requirements of digital sharing economy platforms.

In terms of policy implications, the study identifies key areas where policymakers can intervene to facilitate the digital transformation of SMEs. By understanding the essential sharing areas highlighted in the study, policymakers can focus their efforts on supporting and promoting the development of digital sharing economy platforms within these specific domains. This insight can help policymakers shape policies and initiatives that foster the adoption and utilisation of digital platforms, ultimately empowering SMEs in their digital journey.

In summary, the study's contributions to the practice of IS and policy of digital transformation. The methodology it offers for investigating local sharing activities and designing digital platforms can guide practitioners in the field. Policymakers can leverage the study's insights to identify areas for intervention and support in the digital transformation of SMEs.

References

- Abed, S. S. (2020). Social commerce adoption using TOE framework: An empirical investigation of Saudi Arabian SMEs. *International Journal of Information Management*, *53*, 102118. https://doi.org/10.1016/j.ijinfomgt.2020.102118
- Achtenhagen, L., Melin, L., & Naldi, L. (2013). Dynamics of business models strategizing, critical capabilities and activities for sustained value creation. *Long Range Planning*, 46(6), 427–442. https://doi.org/10.1016/j.lrp.2013.04.002
- Allen, K. E., Castellano, C., & Pessagno, S. (2021). Using dialogue to contextualize culture, ecosystem services, and cultural ecosystem services. *Ecology and Society*, *26*(2). https://doi.org/10.5751/es-12187-260207
- Ameller, D., Burgués, X., Collell, O., Costal, D., Franch, X., & Papazoglou, M. P. (2015). Development of service-oriented architectures using model-driven development: A mapping study. *Information and Software Technology*, 62, 42–66. https://doi.org/10.1016/j.infs of.2015.02.006
- Amornkitvikai, Y., Tham, S. Y., Harvie, C., & Buachoom, W. W. (2022). Barriers and factors affecting the e-commerce sustainability of Thai micro-, small- and medium-sized enterprises (MSMEs). *Sustainability*, *14*(14), 8476. https://doi.org/10.3390/su14148476
- Atinaf, M., Anteneh, S., & Kifle, M. (2023). Socio-technical design principles for a multistakeholder agriculture extension information system in Ethiopia. *Information Technology for Development*, *30*(3), 493–521. https://doi.org/10.1080/02681102.2023.227 9309
- Attié, E., & Meyer-Waarden, L. (2022). The acceptance and usage of smart connected objects according to adoption stages: An enhanced technology acceptance model integrating the diffusion of innovation, uses and gratification and privacy calculus theories. *Technological Forecasting and Social Change*, 176, 121485. https://doi.org/10.1016/j.techf ore.2022.121485
- Bardhi, F., & Eckhardt, G. M. (2012). Access-based consumption: The case of car sharing. *Journal of Consumer Research*, 39(4), 881–898. https://doi.org/10.1086/666376

- Bartelheimer, C., zur Heiden, P., Berendes, C. I., & Beverungen, D. (2023). Designing digital actor engagement platforms for local high streets: An action design research study. *European Journal of Information Systems*, *33*(5), 768–801. https://doi.org/10.1080/0960085x.2023.2242847
- Belk, R. (2014). You are what you can access: Sharing and collaborative consumption online. *Journal of Business Research*, *67*(8), 1595–1600. https://doi.org/10.1016/j.jbusres.20 13.10.001
- Bokolo, A. J., & Petersen, S. A. (2022). Using an extended technology acceptance model to predict enterprise architecture adoption in making cities smarter. *Environment Systems and Decisions*, *43*(1), 36–53. https://doi.org/10.1007/s10669-022-09867-x
- Bonina, C., Koskinen, K., Eaton, B., & Gawer, A. (2021). Digital platforms for development: Foundations and research agenda. *Information Systems Journal*, 31(6), 869–902. https: //doi.org/10.1111/isj.12326
- Chen, A., Yuan, C.-W., Ma, N. F., Hsu, C., & Hanrahan, B. V. (2019). Navigating ride-sharing regulations: How regulations changed the "gig" of ride-sharing for drivers in Taiwan. *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, 1–12. https://doi.org/10.1145/3290605.3300366
- Choudhary, P. K., Kaushik, A., & Bharadwaj, S. S. (2021). Societal digital platforms: Architecture and design. *Vision: The Journal of Business Perspective*, *29*(2), 151–165. https://doi .org/10.1177/09722629211056706
- Dai, Q. (2023). Understanding how platform modularity enhances network effects. *Electronic Markets*, *33*(1). https://doi.org/10.1007/s12525-023-00660-5
- Daradkeh, M. (2023). Exploring the boundaries of success: A literature review and research agenda on resource, complementary, and ecological boundaries in digital platform business model innovation. *Informatics*, *10*(2), 41. https://doi.org/10.3390/informatics10020041
- Dash, G., & Paul, J. (2021). CB-SEM vs PLS-SEM methods for research in social sciences and technology forecasting. *Technological Forecasting and Social Change*, *173*, 121092. https://doi.org/10.1016/j.techfore.2021.121092
- de Reuver, M., Sørensen, C., & Basole, R. C. (2018). The digital platform: A research agenda. *Journal of Information Technology*, *33*(2), 124–135. https://doi.org/10.1057/s41265-016-0033-3
- Deryabin, S., Temkin, I., Rzazade, U., & Kondratev, E. (2023). Models and methods of designing data-centric microservice architectures of digital enterprises. *Informatics*, *10*(1), 4. http s://doi.org/10.3390/informatics10010004
- DiBella, J., Forrest, N., Burch, S., Rao-Williams, J., Ninomiya, S. M., Hermelingmeier, V., & Chisholm, K. (2022). Exploring the potential of SMEs to build individual, organizational, and community resilience through sustainability-oriented business practices. *Business Strategy and the Environment*, *32*(1), 721–735. https://doi.org/10.1002/bse.3171
- Eisape, D. A. (2022). Transforming pipelines into digital platforms: An illustrative case study transforming a traditional pipeline business model in the standardization industry into a digital platform. *Journal of Open Innovation: Technology, Market, and Complexity*, *8*(4), 183. https://doi.org/10.3390/joitmc8040183

- Engert, M., Evers, J., Hein, A., & Krcmar, H. (2023). Sustaining complementor engagement in digital platform ecosystems: Antecedents, behaviours and engagement trajectories. *Information Systems Journal*, *33*(5), 1151–1185. https://doi.org/10.1111/isj.12438
- Engeström, Y. (1993). Developmental studies of work as a testbench of activity theory: The case of primary care medical practice. In *Understanding practice* (pp. 64–103). Cambridge University Press. https://doi.org/10.1017/cbo9780511625510.004
- Gastaldi, L., Appio, F. P., Trabucchi, D., Buganza, T., & Corso, M. (2023). From mutualism to commensalism: Assessing the evolving relationship between complementors and digital platforms. *Information Systems Journal*, *34*(4), 1217–1263. https://doi.org/10.1111/i sj.12491
- Gawer, A. (2021a). Digital platforms and ecosystems: Remarks on the dominant organizational forms of the digital age. *Innovation*, *24*(1), 110–124. https://doi.org/10.1080/14479 338.2021.1965888
- Gawer, A. (2021b). Digital platforms' boundaries: The interplay of firm scope, platform sides, and digital interfaces. *Long Range Planning*, *54*(5), 102045. https://doi.org/10.1016/j .lrp.2020.102045
- Ghobakhloo, M., & Ching, N. T. (2019). Adoption of digital technologies of smart manufacturing in SMEs. *Journal of Industrial Information Integration*, *16*, 100107. https://doi.org/1 0.1016/j.jii.2019.100107
- Govindan, K., Shankar, K. M., & Kannan, D. (2020). Achieving sustainable development goals through identifying and analyzing barriers to industrial sharing economy: A framework development. *International Journal of Production Economics*, 227, 107575. https://doi.o rg/10.1016/j.ijpe.2019.107575
- Gupta, L., & Shankar, R. (2022). Adoption of battery management system in utility grid: An empirical study using structural equation modeling. *Global Journal of Flexible Systems Management*, 23(4), 573–596. https://doi.org/10.1007/s40171-022-00319-8
- Hair, J. F., Howard, M. C., & Nitzl, C. (2020). Assessing measurement model quality in PLS-SEM using confirmatory composite analysis. *Journal of Business Research*, 109, 101–110. https://doi.org/10.1016/j.jbusres.2019.11.069
- Hein, A., Schreieck, M., Riasanow, T., Setzke, D. S., Wiesche, M., Böhm, M., & Krcmar, H. (2019). Digital platform ecosystems. *Electronic Markets*, 30(1), 87–98. https://doi.org /10.1007/s12525-019-00377-4
- Hirota, T., Kishita, Y., Tsunezawa, M., Sugiyama, K., Tasaka, K., & Umeda, Y. (2022). Developing architecture for platform-based circular economy business: An exploratory study. *Procedia CIRP*, *105*, 642–647. https://doi.org/10.1016/j.procir.2022.02.107
- Hou, H., & Shi, Y. (2021). Ecosystem-as-structure and ecosystem-as-coevolution: A constructive examination. *Technovation*, 100, 102193. https://doi.org/10.1016/j.technovation .2020.102193
- Hwang, H., Sarstedt, M., Cho, G., Choo, H., & Ringle, C. M. (2023). A primer on integrated generalized structured component analysis. *European Business Review*, *35*(3), 261–284. https://doi.org/10.1108/ebr-11-2022-0224
- Janowski, T. (2015). Digital government evolution: From transformation to contextualization. *Government Information Quarterly*, *32*(3), 221–236. https://doi.org/10.1016/j.giq.201 5.07.001

- Kamal, S. A., Shafiq, M., & Kakria, P. (2020). Investigating acceptance of telemedicine services through an extended technology acceptance model (TAM). *Technology in Society*, 60, 101212. https://doi.org/10.1016/j.techsoc.2019.101212
- Kaptelinin, V., & Nardi, B. (2017). Activity theory as a framework for human-technology interaction research. *Mind, Culture, and Activity*, 25(1), 3–5. https://doi.org/10.1080/1 0749039.2017.1393089
- Kaptelinin, V., & Nardi, B. A. (1997). Activity theory: Basic concepts and applications. *CHI* '97 Extended abstracts on Human factors in computing systems looking to the future, 158. https://doi.org/10.1145/1120212.1120321
- Kock, N., & Hadaya, P. (2016). Minimum sample size estimation in PLS-SEM: The inverse square root and gamma-exponential methods. *Information Systems Journal*, 28(1), 227– 261. https://doi.org/10.1111/isj.12131
- Köhler, J., Sönnichsen, S. D., & Beske-Jansen, P. (2022). Towards a collaboration framework for circular economy: The role of dynamic capabilities and open innovation. *Business Strategy and the Environment*, 31(6), 2700–2713. https://doi.org/10.1002/bse.3000
- Kouhizadeh, M., Saberi, S., & Sarkis, J. (2021). Blockchain technology and the sustainable supply chain: Theoretically exploring adoption barriers. *International Journal of Production Economics*, 231, 107831. https://doi.org/10.1016/j.ijpe.2020.107831
- Laigner, R., Zhou, Y., & Salles, M. A. V. (2021). A distributed database system for event-based microservices. Proceedings of the 15th ACM International Conference on Distributed and Event-based Systems, 25–30. https://doi.org/10.1145/3465480.3466919
- Legate, A. E., Hair, J. F., Chretien, J. L., & Risher, J. J. (2021). PLS-SEM: Prediction-oriented solutions for HRD researchers. *Human Resource Development Quarterly*, *34*(1), 91–109. https://doi.org/10.1002/hrdq.21466
- Mullarkey, M. T., & Hevner, A. R. (2018). An elaborated action design research process model. *European Journal of Information Systems*, 28(1), 6–20. https://doi.org/10.1080/0960 085x.2018.1451811
- Murer, M., Meschtscherjakov, A., Fuchsberger, V., Giuliani, M., Neureiter, K., Moser, C., Aslan, I., & Tscheligi, M. (2015). Contextual interaction design research: Enabling HCI. In *Human-computer interaction – INTERACT 2015* (pp. 621–623). Springer International Publishing. https://doi.org/10.1007/978-3-319-22723-8_75
- Myovella, G., Karacuka, M., & Haucap, J. (2020). Digitalization and economic growth: A comparative analysis of sub-saharan Africa and OECD economies. *Telecommunications Policy*, 44(2), 101856. https://doi.org/10.1016/j.telpol.2019.101856
- Naik, H. S., Fritzsche, A., & Moeslein, K. M. (2020). Modularity in making: Simplifying solution space for user innovation. *R&D Management*, *51*(1), 57–72. https://doi.org/10.1111/r adm.12427
- Ochinanwata, J., & Ochinanwata, N. (2023). The impact of digital platforms on SMEs' development and performance. In *Small business and entrepreneurial development in Africa* (pp. 179–194). Springer Nature Switzerland. https://doi.org/10.1007/978-3-031-3 7675-7_8
- Pankomera, R., & van Greunen, D. (2019). Opportunities, barriers, and adoption factors of mobile commerce for the informal sector in developing countries in Africa: A systematic

review. The Electronic Journal of Information Systems in Developing Countries, 85(5). https://doi.org/10.1002/isd2.12096

- Peng, X., Ali Babar, M., & Ebert, C. (2014). Collaborative software development platforms for crowdsourcing. *IEEE Software*, *31*(2), 30–36. https://doi.org/10.1109/ms.2014.31
- Rahman, K. S., & Thelen, K. (2019). The rise of the platform business model and the transformation of twenty-first-century capitalism. *Politics & Society*, *47*(2), 177–204. https://doi .org/10.1177/0032329219838932
- Rahman, S. A., Didarul Alam, M. M., & Taghizadeh, S. K. (2019). Do mobile financial services ensure the subjective well-being of micro-entrepreneurs? an investigation applying UTAUT2 model. *Information Technology for Development*, 26(2), 421–444. https://doi.org/10.1080/02681102.2019.1643278
- Ranjbari, M., Morales-Alonso, G., & Carrasco-Gallego, R. (2018). Conceptualizing the sharing economy through presenting a comprehensive framework. *Sustainability*, *10*(7), 2336. https://doi.org/10.3390/su10072336
- Rumanyika, J., Apiola, M., Mramba, N. R., Oyelere, S. S., & Tedre, M. (2021). Design and development of Machinga mobile trading application: A participatory and design science research. *African Journal of Science, Technology, Innovation and Development*, 14(5), 1196–1214. https://doi.org/10.1080/20421338.2021.1942411
- Shneor, R., & Munim, Z. H. (2019). Reward crowdfunding contribution as planned behaviour: An extended framework. *Journal of Business Research*, *103*, 56–70. https://doi.org/10 .1016/j.jbusres.2019.06.013
- Smidt, H. J., & Jokonya, O. (2021). Factors affecting digital technology adoption by small-scale farmers in agriculture value chains (AVCs) in South Africa. *Information Technology for Development*, 28(3), 558–584. https://doi.org/10.1080/02681102.2021.1975256
- Spagnoletti, P., Resca, A., & Lee, G. (2015). A design theory for digital platforms supporting online communities: A multiple case study. *Journal of Information Technology*, 30(4), 364–380. https://doi.org/10.1057/jit.2014.37
- Stamps, A. (2014). A protocol for evaluating contextual design principles. *Behavioral Sciences*, 4(4), 448–470. https://doi.org/10.3390/bs4040448
- Tiberius, V., & Hauptmeijer, R. (2021). Equity crowdfunding: Forecasting market development, platform evolution, and regulation. *Journal of Small Business Management*, *59*(2), 337–369. https://doi.org/10.1080/00472778.2020.1849714
- Tura, N., Kutvonen, A., & Ritala, P. (2017). Platform design framework: Conceptualisation and application. *Technology Analysis & Strategic Management*, 30(8), 881–894. https://doi.o rg/10.1080/09537325.2017.1390220
- Twinomurinzi, H., & Ayalew, S. A. (2022). Identifying business services from small and micro enterprises collaboration: The activity-based service identification framework. *International Journal of Process Management and Benchmarking*, 1(1), 1. https://doi.org/10.15 04/ijpmb.2022.10050446
- Upadhyay, C. K., Tewari, V., & Tiwari, V. (2021). Assessing the impact of sharing economy through adoption of ICT based crowdshipping platform for last-mile delivery in urban and semi-urban India. *Information Technology for Development*, *27*(4), 670–696. https://doi.org/10.1080/02681102.2021.1971147

- Vaske, J. J., Beaman, J., & Sponarski, C. C. (2016). Rethinking internal consistency in Cronbach's alpha. *Leisure Sciences*, 39(2), 163–173. https://doi.org/10.1080/01490400.2 015.1127189
- Weiss, N., Schreieck, M., Wiesche, M., & Krcmar, H. (2023). Lead complementor involvement in the design of platform boundary resources: A case study of BMW's onboard apps. *Information Systems Journal*, *33*(6), 1279–1311. https://doi.org/10.1111/isj.12449
- Wulfert, T., Woroch, R., Strobel, G., Seufert, S., & Möller, F. (2022). Developing design principles to standardize e-commerce ecosystems: A systematic literature review and multicase study of boundary resources. *Electronic Markets*, *32*(4), 1813–1842. https://doi.or g/10.1007/s12525-022-00558-8

The use of detective analytics for mitigating financial crimes: A South African perspective

Nontobeko Mlambo 💿, Tiko Iyamu 💿

Department of Information Technology, Cape Peninsula University of Technology, Cape Town, South Africa

ABSTRACT

South African organisations are increasingly exploring the use of detective analytics in mitigating financial crimes. However, many organisations are struggling with how to employ the tool in detecting and preventing financial crimes. This study aimed to conceptualise how detective analytics can be used to mitigate financial crimes in organisations. Qualitative data were gathered from different sources, from peer-reviewed to grey literature. Actor-network theory (ANT) was employed as a lens, through its mantra 'follow the actors', to gain insights on how activities can be identified, traced, and tracked, in mitigating financial crime in institutions. The interpretive approach was applied. The findings revealed seamless integration of incidents, cybersecurity detection, in-house fraud detection, external infiltrate detection, and image-matching data into one cohesive system. The study highlights the need for gaining a deeper understanding of networks of actors, following the actors, and obligatory passage points within an organisation. The findings have significant implications for improving the efficiency and effectiveness of the use of detective analytics, from both technical and non-technical perspectives.

Keywords Actor-network theory, Detective analytics, Financial crime

 $\label{eq:categories} \bullet Information \ systems \sim Cross-computing \ tools \ and \ techniques, \ Design$

Email	Article history		
Nontobeko Mlambo – mlambononto@gmail.com	Received:	30 May 2024	
Tiko Iyamu – iyamut@cput.ac.za (CORRESPONDING)	Accepted:	4 October 2024	
	Online:	2 July 2025	

1 INTRODUCTION

Financial institutions play a very vital role in the economy of any country. Financial institutions include banking agencies that assist individuals and organisations with carrying out transactions at both national and international levels (Cole, 2023). The transactions include the exchanging of forex and assets, from small to large volumes (Sunio & Mendejar, 2022). Some of the transactions have severe implications and consequences for the actors or the representing agents involved. For example, when a transaction goes wrong on a large scale, an entire organisation can be declared liquidated, which affects the livelihoods of the employees and others connected with the organisation. It is therefore critical, to always guide against

Mlambo, N., and Iyamu, T. (2025). The use of detective analytics for mitigating financial crimes: A South African perspective . *South African Computer Journal* 37(1), 45–62. https://doi.org/10.18489/sacj.v37i1.18697

Copyright © the author(s); published under a Creative Commons NonCommercial 4.0 License (Commons SACJ is a publication of SAICSIT. ISSN 1015-7999 (print) ISSN 2313-7835 (online)

the transactions by being precautionary with the enabling facets, primarily, which are people, data, and technology.

Financial institutions rely on data for the processing of millions of transactions daily (Hasan & Rizvi, 2022). The data is enabled and supported using information technology (IT) solutions (Bataev, 2018). Furthermore, the manipulation, use and management of the data and IT solutions are carried out by people (Andrade-Rojas et al., 2024). Thus, financial institutions continue to build the security and protection of their assets and finances around these three facets. Financial institutions analyse data, to gain a better understanding and make better financial decisions and help prevent processing of suspicious transactions (Li et al., 2020). Despite the preventative, detective, precautionary and security measures, processes, and transactions are often in danger because of fraudulent activities, from unprecedented circumstances (Yamen et al., 2019). Some of the activities are from internal and external entities and agents including humans' actions, consciously or unconsciously.

In the last ten years, South Africa has been one of the most hit countries in the world by financial crime (Kempen, 2020). It is argued that financial crime is double in low-income countries than it is in high-income countries (Achim et al., 2021). This could be attributed to the sophistication of preventative, detective, and other security measures in high-income countries, using IT solutions. Some financial crimes are detected by financial institutions (Gombiro et al., 2015). However, there are loopholes in the current methods and approaches, hence, the rate of financial crime in South Africa is increasing (Macdonald, 2019). Thus, a different mechanism is required, to advance protection and security against financial crime in the country. This should allow and enable early detection of the crime before and as it happens, using the most appropriate mechanism such as detective analytics.

Detective analytics is classified in the group of data analytics, which includes diagnostics: descriptive, predictive, and prescriptive (Vanani & Shaabani, 2021). Data analytics are widely used to combat financial crimes and the most used analytics are predictive, prescriptive, and detective analytics (Menezes et al., 2019). Although there is closeness and a bit of overlap among the analytics tools (Lee et al., 2022), detective analytics uniquely focuses on identifying a problem in data, as and when it occurs (Poornima & Pushpalatha, 2020). In its advancement, detective analytics focuses on performing diagnostics on big data and small data (normal), uncovers and rectifies infeasible events including occurrences (Aliguliyev et al., 2016; Empl & Pernul, 2023). The few organisations that are using detective analytics do so because it is considered advanced analytics (Al-Banna et al., 2023; Menezes et al., 2022).

However, there remain two fundamental issues. Firstly, not many financial institutions use or know how to fully utilise the capabilities of detective analytics (Liu et al., 2021). Secondly, despite its advancement, challenges persist. Thus, the research objective is to gain a deeper understanding on how detective analytics can be applied to mitigate financial crimes in organisations. This prompts the research question, which is: *How can detective analytics be employed for mitigating financial crimes in organisations?* Actor-network theory (ANT) was employed as a lens, to examine human and non-human roles in applying detective analytics, to trace, track, and prevent financial crime in financial organisations. This paper is organised into six main sections. The first and second sections introduce and problematise the study, respectively. A review of the literature focusing on the core aspects of the study is presented in the third section. In the fourth, the theory, ANT that underpins the study is discussed. The methodology applied in the study is covered in the fifth section. The analysis and discussion are presented in the sixth section. A conclusion is drawn in the final section.

2 PROBLEMATISING THE STUDY

Like many organisations, financial institutions depend on data for their strategic and operational activities. Unfortunately, the data includes processes which are continually infiltrated or manipulated, consciously by actors in criminal activities, and unconsciously by human errors (Akinbowale et al., 2020). Some of these crimes are so severe that the organisation is affected and may shut down, which inevitably has an impact on the livelihood of employees. Thus, institutions are constantly exploring and employing tools and approaches to mitigate financial crimes, which is prohibitive to business continuity. Thus, IT solutions are increasingly relied upon for remedy.

Despite the IT security solutions and preventative tools such as the Financial Intelligence Centre Act (FICA), 2001 (Republic of South Africa [RSA], 2001); Banks Act, 1990 (RSA, 1990); and Inspection of Financial Institutions Act, 1998 (RSA, 1998) that have been deployed in layers, for mitigation purposes, financial crimes are on the increase in South Africa (Chitimira & Ncube, 2021; Sutherland, 2017). Economic crime remains significantly higher than the global average rate of 49% (Thomson, 2024; White, 2018). Consequently, many institutions continue to lose income to criminal activities, which affects their sustainability and competitiveness. Another negative effect is that the affected institutions suffer reputation damage, which takes considerable time to recover from (Kshetri, 2019). These highlighted problems derail economic development and growth and affect individuals' job security in the country. Thus, it is critical to find a fresh and more sophisticated solution to mitigating financial crimes for South African institutions.

3 LITERATURE REVIEW

3.1 Financial crime in institutions

Most financial institutions rely on data, and the growth of data is drastic throughout the whole world (Bataev, 2018). Hundreds of millions of financial transactions occur in financial institutions each day and all these transactions lead to data creation (Hasan & Rizvi, 2022). In this age of innovation and machine learning, data is seen as one of the most vital contributors in decision-making for most financial institutions. Consequently, financial institutions have been targets for financial crimes both internally and externally (Yamen et al., 2019). Internally by individuals who have access to transaction data and externally by individuals or organisations

that target specific individuals' information to commit financial crimes. Financial institutions have seen a high rise in financial crimes over the past ten years which negatively impacts the development and reliability of information systems (Hope, 2020). Financial crime is double in low-income countries compared with high-income countries (Achim et al., 2021). South Africa has been one of the leading countries exposed to financial crime (Kempen, 2020).

Financial crime is a widespread problem and has been reported to be very aggressive in African countries due to the high rate of poverty. The economic development minister in South Africa has claimed that over 76000 jobs are lost every year due to financial crime (Hope, 2020). The South African government has tried their best to implement strategies to combat crime in general and these strategies also include the regulation of laws relating to financial crime (De Koker, 2007; Kshetri, 2019). Activities of financial crime seem to be increasing in South Africa. In recent years many financial institutions in South Africa have experienced an increased rate of financial crimes due to the high demand for online transactions (van Niekerk, 2017). Notwithstanding the mitigating approaches implemented by some institutions, technological enhancements are used to commit many financial crimes (Coetzee, 2018).

3.2 Detective analytics

The use of data analytics has grown exponentially in the financial sector (Cockcroft & Russell, 2018). This can be attributed to its strengths of accurate reporting, cost reduction, enhanced decision making and operational benefits. Data analytics in the financial sector creates opportunities to advance financial management for both customers and organisations (Giebe et al., 2019; Nobanee, 2021). With advances in solutions, most financial problems that exist can be solved by examining the available data, which can be done using data analytics (Andriosopoulos et al., 2019). However, some researchers argue that the advancement of data analytics in the financial sector has not been thoroughly explored (López-Robles et al., 2019). The most explored data analytics methods in the financial sector are descriptive analytics, diagnostic analytics, predictive analytics and prescriptive analytics. However, detective analytics has little to no research conducted in finance literature.

Detective analytics focuses on the analytics (or analysis) of data, like other tools such as descriptive analytics, diagnostic analytics, predictive analytics, and prescriptive analytics. Descriptive analytics is used to understand what has occurred in the past using historical data (Janakiraman & Ayyanathan, 2022). While diagnostic analytics focuses on historical data, to gain a deeper understanding of the reason behind certain outcomes (Balali et al., 2020), hence it was mostly used to build insights into why certain events occurred in the way they do (Deshpande et al., 2019). Predictive analytics is used to determine patterns and themes to understand what could happen in the future (Jeble et al., 2018; Selvan & Balasundaram, 2021). From an organisation's perspective, to predict is to forecast a problem or a solution (De Jesús Liriano & Sevillano, 2019).

For many years, financial institutions have been using data analytics to derive patterns that lead to financial criminal activities (Andriosopoulos et al., 2019; Köseoğlu et al., 2022).

Data analytics help to collect relevant data, and to access and integrate the data in providing reports of deeper insights into business operations and productions (Fosso Wamba, 2017). Thus, like other sectors, the use of data analytics has enhanced and enabled the financial sector to make better decisions (Ranjan & Jeyanthi, 2020). Increasingly, many organisations can use data analytics, to explore and visualise data to a simple representation that can be easily understood by both users and managers (Hoang & Bui, 2023).

Despite the benefits that data analytics offers organisations, the financial sector continues to experience an increase in financial crimes (Holzenthal, 2017; Yeoh, 2019). Detective analytics is used to diagnose and detect a problem immediately as and when the problem occurs (Vanani & Majidian, 2021). However, not many financial institutions use or know how to fully utilise the capabilities of detective analytics (Liu et al., 2021). Therefore, detective analytics should be critically explored in the financial sector to help combat criminal activities within the institutions.

Among other things, detective analytics is used by organisations, to detect fraudulent activities (Abdallah et al., 2016). To detect means to discover and identify a problem before it occurs. The use of detective analytics is not only advantageous for future purposes but could also be used to detect traces and track incidents using historical data (Sun et al., 2011). Detective analytics is often described as the combination of predictive analytics and prescriptive analytics in the sense that it forecasts and recommends solutions to problems as and when they occur (Menezes et al., 2019). Fraud detection is used mostly to detect unknowns and known unknowns (Thornton et al., 2013). This makes detective analytics critical for organisations. Corroboratively, there seems to be an emphasis on the need for financial institutions to adopt effective fraud detection techniques such as detective analytics to reduce the number of fraud instances. The use of detective analytics is a needed mechanism for financial institutions due to most processes being data-driven (West & Bhattacharya, 2016).

4 UNDERPINNING THEORY

This study is underpinned by a sociotechnical theory, actor-network theory (ANT). The theory is selected to underpin the study based on three main reasons. Firstly, the nature of the study requires a sociotechnical view because, from technology alone, the challenges remain with the phenomenon. Secondly, in financial fraudulent activities, negotiations occur between humans, technologies, or humans and technology actors. One of the main focuses of ANT is understanding negotiation shifts among actors, consciously or unconsciously (Callon, 1986). Thirdly, the mantra 'follow the actors' of the theory helps to connect (or link) non-humans (activities) with humans through rules and networks including the tools used.

The complementarity of ANT and detective analytics bridges any possible gap, either in the analysis of the data or interpretation of the findings. A comprehensive justification and a guide for complementarily employing theories have been provided in the literature (Iyamu, 2021).

The ANT is a sociotechnical theory that is used by researchers to explain the relationship

between humans and non-human objects (Iyamu, 2024). There are further explanations on how and why ANT mostly focuses on the formation of networks, the relationship between actors, and shifting negotiation within the actor-network (Sage et al., 2011). This helps to understand how human and non-human actors are involved in an activity or incident (Gao, 2005). An actor-network is formed immediately when the actors have aligned interests (Walsham, 1997), and the actor-networks are formed consciously or unconsciously, based on common understanding (Latour, 1996). In summation, the theory focuses on how networks are built and maintained including what makes the network dissolve (Michael, 2017; Shim & Shin, 2016).

Applying ANT allows the researcher to follow the actors in their heterogeneous networks. This is because the theory offers methodological steps in the activities, actions, and interactions between actors in a network (Callon, 1986; Heeks & Stanforth, 2015). "ANT provides a platform which allows for the analysis of both human and non-human interaction in a network" (Iyamu, 2021, p. 73). This means that ANT is a resource for understanding the actions of humans. Congruently, a researcher can draw on ANT to trace information pathways that enable humans to access actions and connect with needed resources (Lefkowitz, 2022). Thus, ANT was employed to provide an underutilised post-humanist lens to understand the creation of collaborative connections between action-based interactions (Kumar & Tissenbaum, 2022).

Another strength of ANT that is critically useful in this study is the concept of translation. The translation in ANT is the process of creating a relationship with things, which did not exist, previously (Lezaun, 2017). Translation occurs when the human actors' interests are aligned with their actor-network (Walsham, 1997). The use of ANT has gained popularity in areas of IS research, in areas such as the adoption of technology, examination of healthcare systems and assessment of IT solutions that support and enable engineering, finance and education activities. Also, ANT is increasingly applied to gain a deeper understanding of processes and innovation of technology (Shim & Shin, 2016).

5 METHODOLOGICAL APPROACH

Based on the focus of the study, the qualitative method was employed. This is because the qualitative method allows an in-depth understanding of the phenomenon being studied (Morgan, 2022). Such understanding is achieved because it allows subjective reasoning and views of the data (Akyildiz & Ahmed, 2021). Additionally, a qualitative method is described as a method that enables an understanding of the factors influencing human behaviours and attitudes in a setting (Iyamu, 2024).

This allows the many realities about human behaviour including events and technology to be explored. The events and realities of this study are the existence of financial institutions, financial crimes, and detective analytics. Ontologically, detective analytics is understood and viewed from different perspectives. For example, detective analytics was used with the Internet of Things (IoT) to generate new insights (Empl & Pernul, 2023). Another reality is that detective analytics has been used to obtain accurate predictions by organisations (Menezes

et al., 2019). Also, there are various ways or approaches to the implementation or adoption of detective analytics. This includes the use of frameworks, policies, and models, some of which have been applied by organisations across the world (Broeders et al., 2017).

Existing materials (documents) were gathered using the document analysis technique. The technique allows the rationalisation of available literature, in the forms of books, newspaper articles, peer-reviewed articles, and technical reports (Morgan, 2022). The data were collected using a set of criteria, to ensure appropriateness. As shown in Table 1, the criteria consist of three groupings:

- i. areas of focus,
- ii. publication dates, and
- iii. credible sources.

Each of the groups was further categorised. The areas of focus include crime in financial institutions, detective analytics and ANT, which are the core aspects of the study. The publication dates were vita, to imbibe context and relevance. Thus, articles published within ten years were considered, to gain insights into critical aspects such as the historical background and meanings associated with the phenomenon over time (Iyamu et al., 2016). For credibility purposes, the concentration was on peer-reviewed articles. Based on the newness of detective analytics, we could only gather a small-sized collection of literature that was most appropriate and relevant to the study. This type of circumstance is not new as it has been experienced and argued in many IS studies (Nyikana & Iyamu, 2023).

Materials published in journals, books, conference proceedings and the internet between 2013 and 2023 were gathered. A total of 255 articles were collected from databases such as Ebscohost, IEEE, AIS, and Emerald. The databases were used as sources because they host many computing articles and instil credibility and reliability in the data (Nyikana & Iyamu, 2023). Based on the focus and objective of the study, only 77 papers were relevant and used for the study. Table 1 presents a sample of the papers.

Guided by the objective of the study, which was to gain an understanding of how detective analytics can be applied to mitigate financial crimes in organisations, the data were processed, and meanings were associated in the context of the study. ANT was employed as a lens in the analysis of the data, to gain deeper insights into mitigating financial crimes in organisations. The analysis focuses on three fundamental areas:

- i. to advance an understanding of evidence in the literature on how humans interact with systems and processes that are obligatory to commit financial crimes;
- ii. by following the actors, to gain insights on how actor-networks are formed, consciously or unintended, that lead to financial crimes; and
- iii. a better understanding of actors' relationships and how they interact in committing financial crimes in organisations.

Focus	Source		Туре
	The impact of national culture on financial crime	(Yamen et al., 2019)	Journal
	Channels of corruption in Africa: analytical review of trends in financial crimes	(Hope, 2020)	Journal
financial institutions	Does technology matter for combating economic and financial crime? A panel data study	(Achim et al., 2021)	Journal
	The world of private investigators in South Africa	(Kempen, 2020)	Journal
	Financial crime in South Africa	(De Koker, 2007)	Journal
	Digital Transformation: Utilization of Analytics and Machine Learning in Smart Manufacturing	(Vanani & Shaabani, 2021)	Book chapter
	Fraud detection system: A survey	(Abdallah et al., 2016)	Journal
Detective analytics	Gross error detection and data reconciliation using historical data	(Sun et al., 2011)	Journal
	Predictive, prescriptive, and detective analytics for smart manufacturing in the information age	(Menezes et al., 2019)	Journal
	Intelligent financial fraud detection: a comprehensive review	(West & Bhattacharya, 2016)	Journal
	Actor-network theory: Trial, trails, and translations	(Kumar & Tissenbaum, 2022)	Book chapter
	How actor-network theories can help in understanding project complexities	(Sage et al., 2011)	Journal
Actor- network theory	Using actor-network theory to analyse strategy formulation	(Gao, 2005)	Journal
	Actor-network theory and IS research: current status and future prospects	(Walsham, 1997)	Conference proceedings
	On actor-network theory: A few clarifications	(Latour, 1996)	Journal
	Applying theories for information systems research	(Iyamu, 2021)	Book chapter

Table 1: Data classification

6 DATA ANALYSIS

As presented in the section that follows including Table 2 ANT is used to gain an understanding of how networks of actors (humans and non-humans) can be linked together in a crimerelated event. This is primarily because ANT proposes and allows actors to be followed, to establish their roles in the processes and activities in finance transactions within an organisation. Table 2 reveals how ANT helps to gain a deeper understanding of the relationship that exists between the actors in the process of executing financial transactions. This includes committing a financial crime or attempting to mitigate the financial crime. In Table 3, the implications of the factors are explained. This includes how activities of financial transactions are connected and lead to crime.

From ANT, a social context perspective, five factors were found to be crucial points in

ANT Tenet	ANT Translative view	Factor
Actor- network	Detective analytics integrates multiple operations, scales, scenarios, and layers (Al-Banna et al., 2023). The multiplexity consists of actor-networks, of humans and non-humans. Both humans and non-humans are interwoven, which means that they cannot be separated. For example, humans and non-humans play roles for a financial transaction to occur. The actor-networks leverage automation of processes based on centralised data, to detect crime patterns across enterprise-wide.	Seamlessly integrates incident
Follow the actors	Detective analytics enacts real-time and offline operations, in detecting compromising incid- ents. Also, detective analytics generates new insights as non-human actors transform. From a cybersecurity perspective, some non-human actors commonly transform are signatures, processes, and rules (Empl & Pernul, 2023). The operations link various compromises to an incident by following the actors. This enhances the power to connect disparate data across an enterprise.	Cybersecurity detection
Obligatory passage points	The concept of obligatory passage point (OPP) infuses tacit knowledge into members of a unit (an actor-network), to maintain a common translation of fraud-related actions and gain a better understanding. OPP is used to facilitate and filter the appropriateness of action that is congruent with the agreed goals. The OPP is the result of "translations", in which the actors have no choice but to accept the dedicates (Callon, 1986). It is thus, an essential glue that brings together various links, in mitigating financial crimes. Thus, OPP brings a different dimension that reveals the roles and identities including recognition of the actors. The translation of obligated processes and rules are required, in following the actors (Latour, 1996). This is a crucial dynamic in the use of detective analytics for the diagnosis of evens, to induce consistency and accuracy.	In-house fraud detection
Black box	By following the actors, it can lead to opening a black box, to gain new insights. Transaction of the new insights leads to discovering of the links and interactions between human and non-human actors, in a financial transaction. Following the actors over time and across a multitude is an important means of analysis, to gain insights. Also, opening the black boxes and examining the ontology of the actor-networks can lead to the construction of new dynamics.	External infiltrate detection
Actor- network	Detective analytics expands the accuracy capability of IT solutions. Detective analytics induces a level of preciseness in its operations on veracity, volume, variety, and velocity of data (Menezes et al., 2022). It advances detective analytics to detect inconsistencies including image recognition in big data and small data.	Image- matching data into one cohesive system

Table 2: Findings from the analysis

following the actors to mitigate financial crime in an organisation. The factors are:

- i. Seamless integration of incidents,
- ii. Cybersecurity detection,

- iii. In-house fraud detection,
- iv. External infiltrate detection, and
- v. Image-matching data into one cohesive system.

The discussion that follows should be read with Tables 2 and 3, to gain a better understanding.

Factor	Technical	Non-technical
Seamlessly integrates incident	It enables the automation of processes and resolves fragmentations. Also, it enables the synchronisation of processes in following the actors. By doing so, it removes manual monitoring and reactive response to fraudulent incidents.	Distinction of actors' roles and defining the commu- nication channel within the structure of the environ- ment. Incident management must be enabled using automated processes, to actively remediate reported incidents.
In-house fraud detection	It must promulgate a policy that allows connectivity to enable and support the centralisation of systems and processes. Thus, the detective approach can en- able enterprise-wide and connect processes and data- bases.	Employees and relevant authorities must understand how to follow the actors. Thus, operations and invest- igations of fraudulent transactions and activities can be fortified. Requirements must be defined and used to follow.
Cybersecurity detection	Cyber activities involve nodes and networks of actors. Thus, it requires an understanding of the fundament- als of detecting and responding to cybersecurity incid- ents by following the actors.	Employees and relevant authorities must understand how to link activities and associated events. There- fore, documentation and reporting of cybersecurity in- cidents require a special approach for communicating security breaches.
External infilt- rate detection	Employs tools for tracing external links that explore internal leaks. Response time finds millions of link connections per second and minute.	Social engineering attacks are human interactions used to manipulate security procedures. A process must be developed to mitigate this activity.
Image-matching data into one cohesive system	Following the actors early enough helps to detect when an actor transforms or impersonates himself or herself. Cohesiveness increases accuracy through its integrated approach to fraud prevention.	It requires the configuration of rules and intents within an organisation. Fraud categorisation-based rules help to ease processing and anomaly detection.

Table 3: Implication of the study

7 DISCUSSION OF THE FINDINGS

The factors revealed from the analysis: Seamless integration of incidents, Cybersecurity detection, In-house fraud detection, External infiltrate detection, and Image-matching data into one cohesive system are interrelated in committing or detecting financial crimes in organisations. Seamless integration refers to the smooth and efficient connection between processes and computer technology such as detective analytics. Seamless integration of technology has been demonstrated to eliminate insertion and mismatch of incidents, from internal and external intrusion to image matching. Cybersecurity detection is an aspect of IT solutions focusing on activities including images in cyberspace. The solution is intended to detect and recognise fraudulent activities as they occur within an organisation (Abdallah et al., 2016). Financial cybercrime could be orchestrated from in-house or externally. In-house fraud detection discovers and identifies fraudulent financial activities within an organisation. The detection system is used to follow and trace actors' financial activities as they try to enter an organisation's space and commit a crime.

Correspondingly, some activities and actions involved unregulated and unprecedented movements. Consequently, organisations are increasingly employing detective analytics to manage and promote their business efficacy (Iyamu, 2022). However, it is challenging to identify some hidden characters that manifest in many environments. Thus, the analysis of fraudulent or suspected incidents should not follow a fixed process. In our attempt to gain a

deeper understanding of how to employ detective analytics in the qualitative world for analysis purposes, we focus on the notion of 'actor-network', 'follow the actors', 'obligatory passage point' (OPP), and black box from ANT perspective.

Relationships between actors, humans and non-humans are informed by processes of association and translation that can be "material as well as social, physical as well as semiotic" (Michael, 2017, p. 41). ANT's mantra is "to follow the actors" (Boodhun & Jayabalan, 2018, p. 12) that are not exclusively human but non-human parts of interactions within networks. Actornetworks need to be unfolded, and the inherent black boxes offer insights into their ontologies and allow detailed accounts of these inner workings . The OPP characterises the dynamic convergence of the processes of network constitution and highlights crucial participatory roles and identities of various actors (Minniti & Magaudda, 2024).

Detective analytics focuses on performing the diagnostics on big data and small data (normal), uncovers and rectifies infeasible events and occurrences (Aliguliyev et al., 2016; Empl & Pernul, 2023). From this standpoint, OPP, in a detective analytics operation, helps to gain insights about actors' roles in activity by recognising their identities, how they transformed, and consolidating their alignments. Also, ANT proposes a relational ontology, to gain a deeper understanding of networks and existing syntax as enactments of associations between actors, human and non-human.

The syntax is discovered as we follow the actors . Complementarily, using detective analytics, explicitly, ensures the timeliness, consistency, and integration of big data, in recognising activities and traces of events in an environment (Menezes et al., 2022). In detective analytics, constraints are reduced, and the diagnosis eliminates and rectifies inappropriate values.

Across the world, governments of countries are trying various approaches to combat financial crimes in financial institutions. Yet, the situation has not eased, instead, it has increased. Revealed in the literature, there is evidence of financial crimes and it is argued that until corruption channels such as embezzlement, theft, bribes, kickbacks, money laundering, and illicit financial flow are thoroughly, investigated financial crime will continue to increase (Hope, 2020). The approach, "follow the actor" is consistently useful, as it spears the traces of the minds, processes, and occurrences, towards a fundamentally constructive decision. Among other things, this provides nuance and an in-depth assessment of the interrelatedness that exists between human actors and non-human actants along with their translation and participatory power. In 2021, during the lockdown caused by COVID-19, which resulted in more reliance on online transactions, financial crime in South Africa increased by 15.1%.

The perpetrators of financial crimes in organisations are either internal or external personnel. From the internal perspective, the crimes are usually intended or unconscious (human error) by employees who have access to internal procedures and data, while the external factors include fraudsters, phishing, and money laundering. The increase in financial crime is due to the emerging technologies that institutions adopt to enhance processes (West & Bhattacharya, 2016). Thus, there is a need to explore how the use of detective analytics as a mechanism can be used to detect financial crime.

8 IMPLICATION OF THE STUDY

The factors that manifest from using ANT to examine the use of detective analytics are Seamless integration of incidents, In-house fraud detection, Cybersecurity detection; External infiltrate detection; and Image-matching data into one cohesive system. Each of the factors has both technical and non-technical implications, as tabulated in Table 3. Fundamentally, the implication can provide guidelines on how humans interpret and apply rules in carrying out financial transactions in an organisation. This includes how detective analytics integrate with other tools or systems, to mitigate financial crime.

9 CONTRIBUTION OF THE STUDY

This study is intended to serve as a comprehensive work and extend detective analytics for mitigating financial crimes, in the literature. This study focuses on articles published between 2013 and 2023. In lieu of the other studies on detective analytics available thus far in the literature, this study makes the following three main contributions:

- i. It adds to the literature on detective analytics by providing a nuanced perspective on mitigating financial crimes, from both business and IT perspectives.
- ii. It demonstrates the advancement of ANT, in gaining a better understanding of complex organisational phenomena.
- iii. It reveals the need for a continual comprehensive investigation of advances in detecting and mitigating financial crimes in organisations. It thus suggests that organisations need to focus on advancing detection capability.

10 CONCLUSION

This study highlights the critical challenges faced by organisations in mitigating financial crime due to the constraint of following the actors. The findings reveal that Seamless integration of incidents, Cybersecurity detection, In-house fraud detection, External infiltrate detection, and Image-matching data into one cohesive system are major barriers to effectively mitigating financial crime in organisations. An understanding of these factors has the potential to address these challenges. However, the implementation of detective analytics would require significant efforts in terms of understanding the implications from both technical and non-technical perspectives. Nonetheless, the study also has some limitations, such as the focus on a non-empirical or case study approach. Future research should aim to validate the findings in organisational settings and explore the feasibility and impact of implementing detective analytics in a financial institution.

References

- Abdallah, A., Maarof, M. A., & Zainal, A. (2016). Fraud detection system: A survey. *Journal of Network and Computer Applications*, 68, 90–113. https://doi.org/10.1016/j.jnca.2016 .04.007
- Achim, M. V., Borlea, S. N., & Văidean, V. L. (2021). Does technology matter for combating economic and financial crime? A panel data study. *Technological and Economic Development* of Economy, 27(1), 223–261. https://doi.org/10.3846/tede.2021.13977
- Akinbowale, O. E., Klingelhöfer, H. E., & Zerihun, M. F. (2020). Analysis of cyber-crime effects on the banking sector using the balanced score card: A survey of literature. *Journal of Financial Crime*, 27(3), 945–958. https://doi.org/10.1108/jfc-03-2020-0037
- Akyildiz, S. T., & Ahmed, K. H. (2021). An overview of qualitative research and focus group discussion. *International Journal of Academic Research in Education*, 7(1), 1–15. https://doi.org/10.17985/ijare.866762
- Al-Banna, A., Menezes, B. C., Yaqot, M., & Kelly, J. D. (2023). Decision regression for modelling of supply chain resilience in interdependent networks: LNG case. In 33rd European symposium on computer aided process engineering (pp. 1113–1118). Elsevier. https://doi .org/10.1016/b978-0-443-15274-0.50178-5
- Aliguliyev, R., Imamverdiyev, Y., & Abdullayeva, F. (2016). The investigation of the opportunities of big data analytics as analytics-as-a-service in cloud computing for oil and gas industry. *Problems of Information Technology*, *07*(1), 9–22. https://doi.org/10.25045/j pit.v07.i1.02
- Andrade-Rojas, M. G., Saldanha, T. J. V., Kathuria, A., Khuntia, J., & Boh, W. (2024). How information technology overcomes deficiencies for innovation in small and mediumsized enterprises: Closed innovation vs. open innovation. *Information Systems Research*. https://doi.org/10.1287/ISRE.2021.0096
- Andriosopoulos, D., Doumpos, M., Pardalos, P. M., & Zopounidis, C. (2019). Computational approaches and data analytics in financial services: A literature review. *Journal of the Operational Research Society*, 70(10), 1581–1599. https://doi.org/10.1080/01605682 .2019.1595193
- Balali, F., Nouri, J., Nasiri, A., & Zhao, T. (2020). Data analytics. In *Data intensive industrial asset management* (pp. 105–113). Springer International Publishing. https://doi.org/10.1007/978-3-030-35930-0_7
- Bataev, A. V. (2018). Evaluation of using big data technologies in Russian financial institutions. Proceedings of the 2018 International Conference "Quality Management, Transport and Information Security, Information Technologies", 573–577. https://doi.org/10.1109 /ITMQIS.2018.8524938
- Boodhun, N., & Jayabalan, M. (2018). Risk prediction in life insurance industry using supervised learning algorithms. *Complex & Intelligent Systems*, 4(2), 145–154. https://doi.or g/10.1007/s40747-018-0072-1

- Broeders, D., Schrijvers, E., van der Sloot, B., van Brakel, R., de Hoog, J., & Hirsch Ballin, E. (2017). Big Data and security policies: Towards a framework for regulating the phases of analytics and use of Big Data. *Computer Law & Security Review*, *33*(3), 309–323. http s://doi.org/10.1016/j.clsr.2017.03.002
- Callon, M. (1986). The sociology of an actor-network: The case of the electric vehicle. In *Mapping the dynamics of science and technology* (pp. 19–34). Palgrave Macmillan UK. https://doi.org/10.1007/978-1-349-07408-2_2
- Chitimira, H., & Ncube, P. (2021). The regulation and use of artificial intelligence and 5G technology to combat cybercrime and financial crime in South African banks. *Potchefstroom Electronic Law Journal*, 24, 1–33. https://doi.org/10.17159/1727-3781/2021/v24i0 a10742
- Cockcroft, S., & Russell, M. (2018). Big data opportunities for accounting and finance practice and research. *Australian Accounting Review*, *28*(3), 323–333. https://doi.org/10.1111 /auar.12218
- Coetzee, J. (2018). Strategic implications of Fintech on South African retail banks. South African Journal of Economic and Management Sciences, 21(1). https://doi.org/10.4 102/sajems.v21i1.2455
- Cole, T. (2023). How are financial institutions enabling online fraud? A developmental online financial fraud policy review. *Journal of Financial Crime*, *30*(6), 1458–1473. https://doi.org/10.1108/JFC-10-2022-0261
- De Jesús Liriano, R., & Sevillano, M. C. (2019). Improving the learning process in the higher education through the use of a predictive tool (dashboard). *Distance Learning*, *16*(1), 33–40. https://eric.ed.gov/?id=EJ1299603
- De Koker, L. (2007). Financial crime in South Africa. *Economic Affairs*, 27(1), 34–38. https://doi.org/10.1111/j.1468-0270.2007.00707.x
- Deshpande, P. S., Sharma, S. C., & Peddoju, S. K. (2019). Security and data storage aspect in cloud computing. Springer Singapore. https://doi.org/10.1007/978-981-13-6089-3
- Empl, P., & Pernul, G. (2023). Digital-twin-based security analytics for the internet of things. *Information*, *14*(2), 95. https://doi.org/10.3390/info14020095
- Fosso Wamba, S. (2017). Big data analytics and business process innovation. *Business Process* Management Journal, 23(3), 470–476. https://doi.org/10.1108/bpmj-02-2017-0046
- Gao, P. (2005). Using actor-network theory to analyse strategy formulation. *Information Systems Journal*, *15*(3), 255–275. https://doi.org/10.1111/j.1365-2575.2005.00197.x
- Giebe, C., Hammerström, L., & Zwerenz, D. (2019). Big data & analytics as a sustainable customer loyalty instrument in banking and finance. *Financial Markets, Institutions and Risks*, 3(4), 74–88. https://doi.org/10.21272/fmir.3(4).74-88.2019
- Gombiro, C., Jantjies, M., & Mavetera, N. (2015). A conceptual framework for detecting financial crime in mobile money transactions. *Journal of Governance and Regulation*, *4*(4), 727–734. https://doi.org/10.22495/jgr_v4_i4_c6_p8

- Hasan, I., & Rizvi, S. (2022). AI-driven fraud detection and mitigation in e-commerce transactions. *Lecture Notes on Data Engineering and Communications Technologies*, *90*, 403–414. https://doi.org/10.1007/978-981-16-6289-8_34
- Heeks, R., & Stanforth, C. (2015). Technological change in developing countries: Opening the black box of process using actor–network theory. *Development Studies Research*, 2(1), 33–50. https://doi.org/10.1080/21665095.2015.1026610
- Hoang, T. G., & Bui, M. L. (2023). Business intelligence and analytic (BIA) stage-of-practice in micro-, small- and medium-sized enterprises (MSMEs). *Journal of Enterprise Information Management*, 36(4), 1080–1104. https://doi.org/10.1108/jeim-01-2022-0037
- Holzenthal, F. (2017). Five trends shaping the fight against financial crime. *Computer Fraud & Security*, 2017(3), 5–9. https://doi.org/10.1016/s1361-3723(17)30022-2
- Hope, K. R. (2020). Channels of corruption in Africa: Analytical review of trends in financial crimes. *Journal of Financial Crime*, *27*(1), 294–306. https://doi.org/10.1108/jfc-05-2 019-0053
- Iyamu, T. (2024). *The application of sociotechnical theories in information systems research*. London: Cambridge Scholar Press.
- Iyamu, T. (2021). Applying theories for information systems research. Routledge. https://doi.or g/10.4324/9781003184119
- Iyamu, T. (2022). Advancing big data analytics for healthcare service delivery. Routledge. https://doi.org/10.4324/9781003251064
- Iyamu, T., Nehemia-Maletzky, M., & Shaanika, I. (2016). The overlapping nature of business analysis and business architecture: What we need to know. *Electronic Journal of Information Systems Evaluation*, 19(3), 169–179. https://academic-publishing.org/index.php /ejise/article/view/159
- Janakiraman, S., & Ayyanathan, N. (2022). Design and development of big data framework using NoSQL–MongoDB and descriptive analytics of Indian green coffee export demand modeling. In *Disruptive technologies for big data and cloud applications* (pp. 777–785). Springer Nature Singapore. https://doi.org/10.1007/978-981-19-2177-3_72
- Jeble, S., Dubey, R., Childe, S. J., Papadopoulos, T., Roubaud, D., & Prakash, A. (2018). Impact of big data and predictive analytics capability on supply chain sustainability. *The International Journal of Logistics Management*, 29(2), 513–538. https://doi.org/10.110 8/ijlm-05-2017-0134
- Kempen, A. (2020). The world of private investigators in South Africa. Servamus Communitybased Safety and Security Magazine, 11(113), 34–37. https://hdl.handle.net/10520/ejc -servamus-v113-n11-a12
- Köseoğlu, S. D., Ead, W. M., & Abbassy, M. M. (2022). Basics of financial data analytics. In *Financial data analytics* (pp. 23–57). Springer International Publishing. https://doi.org /10.1007/978-3-030-83799-0_2
- Kshetri, N. (2019). Cybercrime and cybersecurity in Africa. *Journal of Global Information Technology Management*, 22(2), 77–81. https://doi.org/10.1080/1097198x.2019.160352 7

- Kumar, V., & Tissenbaum, M. (2022). Supporting collaborative classroom networks through technology: An actor network theory approach to understanding social behaviours and design. *British Journal of Educational Technology*, *53*(6), 1549–1570. https://doi.org/1 0.1111/bjet.13274
- Latour, B. (1996). On actor-network theory: A few clarifications. *Soziale Welt*, *47*, 369–381. https://www.jstor.org/stable/40878163
- Lee, C. S., Cheang, P. Y. S., & Moslehpour, M. (2022). Predictive analytics in business analytics: Decision tree. *Advances in Decision Sciences*, *26*, 1–29. https://doi.org/10.47654/V26 Y2022I1P1-30
- Lefkowitz, D. (2022). Black boxes and information pathways: An actor-network theory approach to breast cancer survivorship care. *Social Science & Medicine*, *307*, 115184. http s://doi.org/10.1016/j.socscimed.2022.115184
- Lezaun, J. (2017). Actor-network theory. In Social theory now. The University of Chicago Press.
- Li, J., Li, J., Zhu, X., Yao, Y., & Casu, B. (2020). Risk spillovers between FinTech and traditional financial institutions: Evidence from the U.S. *International Review of Financial Analysis*, 71, 101544. https://doi.org/10.1016/J.IRFA.2020.101544
- Liu, X., Shin, H., & Burns, A. C. (2021). Examining the impact of luxury brand's social media marketing on customer engagement: Using big data analytics and natural language processing. *Journal of Business Research*, 125, 815–826. https://doi.org/10.1016/j .jbusres.2019.04.042
- López-Robles, J. R., Rodríguez-Salvador, M., Gamboa-Rosales, N. K., Ramirez-Rosales, S., & Cobo, M. J. (2019). The last five years of big data research in economics, econometrics and finance: Identification and conceptual analysis. *Procedia Computer Science*, 162, 729–736. https://doi.org/10.1016/j.procs.2019.12.044
- Macdonald, D. (2019). Barriers to reconstruction and development: A brief view of corruption and economic crime in Southern Africa. In *Structural adjustment, reconstruction and development in Africa* (pp. 172–178). Routledge. https://doi.org/10.4324/9780429401 749-15
- Menezes, B., Kelly, J. D., Leal, A. G., & le Roux, G. C. (2019). Predictive, prescriptive and detective analytics for smart manufacturing in the information age. *IFAC-PapersOnLine*, 52(1), 568–573. https://doi.org/10.1016/j.ifacol.2019.06.123
- Menezes, B., Yaqot, M., Hassaan, S., Franzoi, R., AlQashouti, N., & Al-Banna, A. (2022). Digital transformation in the era of Industry 4.0 and Society 5.0: A perspective. 2022 2nd International Conference on Emerging Smart Technologies and Applications (eSmarTA), 1–6. https://doi.org/10.1109/esmarta56775.2022.9935399
- Michael, M. (2017). Actor-network theory: Trials, trails and translations. SAGE Publications Ltd. https://doi.org/10.4135/9781473983045
- Minniti, S., & Magaudda, P. (2024). The 'obligatory passage point' in knowledge co-production: Italy's participatory environmental monitoring platform. *Science as Culture*, 34(1), 13– 30. https://doi.org/10.1080/09505431.2024.2317236

- Morgan, H. (2022). Conducting a qualitative document analysis. The Qualitative Report. https:// //doi.org/10.46743/2160-3715/2022.5044
- Nobanee, H. (2021). A bibliometric review of big data in finance. *Big Data*, 9(2), 73–78. http s://doi.org/10.1089/big.2021.29044.edi
- Nyikana, W., & Iyamu, T. (2023). A formulaic approach for selecting Big Data analytics tools for organizational purposes. In Handbook of research on driving socioeconomic development with big data (pp. 224–242). IGI Global. https://doi.org/10.4018/978-1-6684-5959-1.ch010
- Poornima, S., & Pushpalatha, M. (2020). A survey on various applications of prescriptive analytics. International Journal of Intelligent Networks, 1, 76–84. https://doi.org/10.1016/j .ijin.2020.07.001
- Ranjan, J., & Jeyanthi, M. P. (2020). Big data analytics in the healthcare industry. In Global business leadership development for the fourth industrial revolution (pp. 134–154). IGI Global. https://doi.org/10.4018/978-1-7998-4861-5.ch006
- Republic of South Africa. (1990). Banks Act (previously known as deposit-taking institutions Act) 94 of 1990 [Retrieved June 13, 2025]. https://www.gov.za/documents/deposit-takin g-institutions-act-6-mar-2015-1030
- Republic of South Africa. (1998). Inspection of financial institutions Act 80 of 1998 [Retrieved June 13, 2025]. https://www.gov.za/documents/inspection-financial-institutions-act
- Republic of South Africa. (2001). Financial intelligence centre Act 38 of 2001 [Retrieved June 13, 2025]. https://www.gov.za/documents/financial-intelligence-centre-act
- Sage, D., Dainty, A., & Brookes, N. (2011). How actor-network theories can help in understanding project complexities. International Journal of Managing Projects in Business, 4(2), 274-293. https://doi.org/10.1108/17538371111120243
- Selvan, C., & Balasundaram, S. R. (2021). Data analysis in context-based statistical modeling in predictive analytics. In Handbook of research on engineering, business, and healthcare applications of data science and analytics (pp. 96–114). IGI Global. https://doi.org/10.4 018/978-1-7998-3053-5.ch006
- Shim, Y., & Shin, D.-H. (2016). Analyzing China's Fintech industry from the perspective of actor-network theory. Telecommunications Policy, 40(2-3), 168–181. https://doi.org/1 0.1016/j.telpol.2015.11.005
- Sun, S., Huang, D., & Gong, Y. (2011). Gross error detection and data reconciliation using historical data. Procedia Engineering, 15, 55–59. https://doi.org/10.1016/j.proeng.20 11.08.012
- Sunio, V., & Mendejar, J. (2022). Financing low-carbon transport transition in the Philippines: Mapping financing sources, gaps and directionality of innovation. Transportation Research Interdisciplinary Perspectives, 14, 100590. https://doi.org/10.1016/J.TRIP.2022 .100590
- Sutherland, E. (2017). Governance of cybersecurity The case of South Africa. The African Journal of Information and Communication (AJIC), (20), 83–112. https://doi.org/10.23 962/10539/23574

- Thomson, D. (2024). Navigating the new era of financial resilience: Preparing for Dora [Retrieved 4 August 2024]. https://www.bizcommunity.com/article/navigating-the-newera-of-financial-resilience-preparing-for-dora-903064a
- Thornton, D., Mueller, R. M., Schoutsen, P., & van Hillegersberg, J. (2013). Predicting healthcare fraud in Medicaid: A multidimensional data model and analysis techniques for fraud detection. *Procedia Technology*, *9*, 1252–1264. https://doi.org/10.1016/j.protc y.2013.12.140
- van Niekerk, B. (2017). An analysis of cyber-incidents in South Africa. *The African Journal of Information and Communication (AJIC)*, (20). https://doi.org/10.23962/10539/23573
- Vanani, I. R., & Majidian, S. (2021). Prescriptive analytics in internet of things with concentration on deep learning. In *Introduction to internet of things in management science and operations research* (pp. 31–54). Springer International Publishing. https://doi.org/10.1007/978-3-030-74644-5_2
- Vanani, I. R., & Shaabani, A. (2021). Digital transformation. In Artificial intelligence, machine learning, and data science technologies (pp. 269–281). CRC Press. https://doi.org/10.12 01/9781003153405-14
- Walsham, G. (1997). Actor-network theory and IS research: Current status and future prospects. In *Information systems and qualitative research* (pp. 466–480). Springer US. https://doi .org/10.1007/978-0-387-35309-8_23
- West, J., & Bhattacharya, M. (2016). Intelligent financial fraud detection: A comprehensive review. *Computers & Security*, 57, 47–66. https://doi.org/10.1016/j.cose.2015.09.005
- White, T. (2018). Reported economic crime in South Africa hits record levels; cost and accountability concerns rising [Retrieved June 2022]. https://pricewaterhousecooperspwc.africa-newsroom.com/press/reported-economic-crime-in-south-africa-hits-rec ord-levels-cost-and-accountability-concerns-rising?lang=en
- Yamen, A., Al Qudah, A., Badawi, A., & Bani-Mustafa, A. (2019). The impact of national culture on financial crime. *Journal of Money Laundering Control*, 22(2), 373–387. https://doi.o rg/10.1108/jmlc-01-2018-0004
- Yeoh, P. (2019). Artificial intelligence: Accelerator or panacea for financial crime? *Journal of Financial Crime*, *26*(2), 634–646. https://doi.org/10.1108/jfc-08-2018-0077

An investigation into the feasibility of using virtual environments as an induction method in $SHIP^{\mathbb{R}}$ therapy

Adolf Weich Malan^(D), Isak de Villiers Bosman^(D), Theo JD Bothma^(D)

Department of Information Science, University of Pretoria, Pretoria, South Africa

ABSTRACT

Virtual reality contributes to the successful treatment of patients by assisting those who have difficulty with the process of imagining the required visual images needed during psychotherapy. SHIP® is a form of psychotherapy that suggests that spontaneous healing is a natural tendency that emerges from within a person. It identifies certain activator images as essential pathways for accessing unconscious trauma material that needs healing. The purpose of this study was to examine whether virtual reality can be used as a medium to induce memories through the utilisation of neutral images based on the SHIP[®] Frame. Two groups of participants were gathered: one group underwent a traditional SHIP[®] session while the other group underwent the virtual induction with the aid of a head mounted display. A random clinical trial was used to determine the level of induction and identify the helpful aspects that contributed to the induction. The results indicated that virtual reality was able to assist as a cognitive stimulus as well as a cognitive proxy in the overall process of SHIP[®].

Keywords virtual reality, psychotherapy, exposure therapy, virtual environment, SHIP®

 $\textbf{Categories} \quad \textbf{• Human-centred computing} \sim \textbf{Human computer interaction} \quad \textbf{• Applied computing} \sim \textit{Life and medical sciences}$

Email		Article history		
Adolf Weich Malan – weich14.malan@gmail.com (CORRESPONDING)	Received:	20 July 2024		
Isak de Villiers Bosman – isak.bosman@up.ac.za	Accepted:	15 January 2025		
Theo JD Bothma – theo.bothma@up.ac.za	Online:	2 July 2025		

1 INTRODUCTION

Exposure therapy (ET) and spontaneous healing intra-systemic process (SHIP[®]) are both psychotherapeutic methods that are effective in treating individuals for various psychological disorders and/or trauma-spectrum manifestations (TSMs) (Bryant et al., 2019; Levinson et al., 2020; Steenkamp, 2018). If the patient is unable to visualise what the therapist requests them to, the treatment may be less effective, or it might not succeed at all (Friedrich, 2016; Steenkamp, 2018). Virtual reality (VR) could provide a solution to this problem by helping a patient see the requested stimuli.

Malan, A.W., Bosman, I.de V., and Bothma, T.J.D. (2025). An investigation into the feasibility of using virtual environments as an induction method in SHIP[®] therapy. *South African Computer Journal* 37(1), 63–93. https://doi.org/10.18489/sacj.v37i1.19286

Copyright © the author(s); published under a Creative Commons NonCommercial 4.0 License © () (S *SACJ* is a publication of *SAICSIT*. ISSN 1015-7999 (print) ISSN 2313-7835 (online)

VR has already contributed to the successful treatment of psychotherapeutic patients when used in tandem with exposure therapy (Donker et al., 2019; Hodges et al., 1995). This success can be partly attributed to VR's ability to simulate virtual environments (VEs) within which to immerse patients (Vienne et al., 2020). This enables the user to feel present or have the sense of 'being there' (Høeg et al., 2021; Riva, 2022) which makes it a useful tool for psychotherapy, particularly in exposure-based treatments (Albakri et al., 2022; Wilson & Soranzo, 2015). However, exposure therapy and, by extension, virtual reality exposure therapy (VRET) are both based on systematic desensitisation therapy (Markowitz & Fanselow, 2020) and many post-traumatic stress disorder (PTSD) patients still do not benefit from it (Foa & Mclean, 2015; Markowitz & Fanselow, 2020). Another PTSD treatment method, SHIP[®], differs from this approach and attempts to heal the client through completing a frozen traumatic memory (Steenkamp, 2018).

In the realm of VR as a psychotherapeutic tool, research tends to focus on VR exposure therapy and not on other forms of treatment. Through a random clinical trial (RCT) (Hariton & Locascio, 2018) this study attempteds to address this by investigating the efficacy of VR induction as a medium for facilitating the SHIP[®] process. By comparing two groups, where one underwent a standard SHIP[®] intervention and the other a VR intervention, data were collected and analysed to determine if the VR intervention was as assistive to the overall SHIP[®] process as the standard intervention. The analysis of the data was used to answer one main research question:

How can a simulated virtual environment stimulate and enhance the participant's imagecreative neural-visual facility – i.e. the participant's imagination – to assist participants who find it difficult to imagine the requested images in SHIP[®]?

The remainder of this paper answers the above research question through literature and empirical data to determine whether VR can be used in tandem with SHIP[®] to assist clients who have difficulty with imagining requested visual stimuli.

2 REVIEW OF LITERATURE

VR aims to create environments and situations that resemble reality or any fictitious reality (Albakri et al., 2022) and generate an experience that is immersive and that feels as engaging as reality when an individual is placed within it (Garrett et al., 2018). A VR experience can be understood with reference to Sherman and Craig's (2003) four main factors of a VR experience, with the addition of Muhanna's (2015) fifth, and are as follows:

- 1. Virtual world: The environment simulated through computer software (Albakri et al., 2022).
- 2. Sensory feedback: Input and output devices such as headphones and controllers (de Regt et al., 2020).
- 3. Interactivity: Having the ability to change a virtual world through manipulating it, such as picking up objects or turning them around (Muhanna, 2015; Sherman & Craig, 2003).
- 4. Participants: Having at least one person being present in the virtual environment and experiencing it (Muhanna, 2015).
- 5. Immersion: An objective account of its ability to simulate reality with high levels of fidelity (Wilson & Soranzo, 2015).

Because of the importance of immersion and how it is often confused with presence, it will be discussed in the next section in more detail.

2.1 Immersion and presence

When VR is able to supplant a user's senses with the sensory stimuli of a simulated world and metaphorically remove users from the real world, the VR is considered to be immersive (Riva, 2022). Immersion within a virtual world has been described in several ways, including system immersion (Lønne et al., 2023), mental immersion, physical immersion (Albakri et al., 2022; Sherman & Craig, 2003), and narrative immersion (Nilsson et al., 2016). Immersion can be quantified to a large extent and altered with regards to the level to which a user can explore the virtual world, how realistic it appears, and more technical aspects such as graphic frame rate (Kelly et al., 2023). Presence, on the other hand, comes from the degree to which an individual can respond naturally to the VE (Kober et al., 2012; Slater et al., 2009; Wilson & Soranzo, 2015) and makes the user feel like they are "there" in reality (Albakri et al., 2022; Botella et al., 2017; Wilson & Soranzo, 2015). The psychological response of a user to a simulated environment can be seen as the perceived presence by the user (Kelly et al., 2023; Slater et al., 2009; Wilson & Soranzo, 2015) and differs from immersion (Slater et al., 2009). In other words, depending on the user's specific state of mind, as well as other factors, different users might experience the same degree of immersion but not the same degree of presence within a VE (Slater et al., 2009; Wilson & Soranzo, 2015). If a user is self-aware within a VE and able to respond and manipulate their surroundings, they should be considered to be highly present within the VE (Kelly et al., 2023). Although immersion and presence differ from one another, a recent study suggests that immersive VR was more successful in instilling a sense of presence within learners than desktop VR because the participants felt as if they were "actually there" in the VE (Lønne et al., 2023).

VR generates a simulated environment within which a user can explore and interact as if it were the real world, forecasting the sensory inputs and experiences one might receive in reality (Riva et al., 2019). This is similar to predictive coding, a popular framework for comprehending how messages are passed around in our brains (Friston, 2012). The feeling of presence or "being there" is the characteristic that separates VR from other traditional media (Garrett et al., 2018; Høeg et al., 2021; Riva, 2022; Rizzo et al., 2015) and can help psychology in many different ways such as exposing a veteran who suffers from PTSD to a traumatic event in a safe and controlled environment (Albakri et al., 2022; Friedrich, 2016).

2.2 Virtual reality exposure therapy (VRET)

Exposure therapy, which is a particular form of cognitive behavioural therapy (CBT), is a psychotherapeutic treatment that has been proven to be effective in the treatment of various psychological disorders (Levinson et al., 2020; Olatunji et al., 2010; Opriş et al., 2012) and PTSD (Bryant et al., 2019). Exposure therapy focuses on changing pathological fear structures by repeatedly confronting the patient with the stimuli that trigger these fear structures and incorporating information that goes directly against the pathological elements inside the fear structure (Rauch & Foa, 2006). This is achieved either through real-life in vivo exposure, where the patient is immersed in real life within the environment that first caused the trauma (if possible), or imaginal exposure, where a patient imagines the events where fear and trauma are experienced (Friedrich, 2016). There are some shortfalls to exposure therapy (Beidel et al., 2019) such as some patients having difficulty with the process of imagining traumatic events (Friedrich, 2016) which motivates other approaches such as virtual reality exposure therapy (VRET) where the feared stimuli are generated through moving imagery, sounds, and sometimes through simulating other senses such as smell within VR (Eshuis et al., 2021). VRET offers an alternative to traditional exposure therapy, and has shown greater success in treating patients with PTSD who did not respond to prior therapies (Volovik et al., 2023).

VRET systems have been as successful as in vivo therapy in the treatment of various phobias (Donker et al., 2019; Suied et al., 2013), and PTSD in veterans of war (Volovik et al., 2023) or survivors of the World Trade Centre attacks in 2001 (Beidel et al., 2019; Friedrich, 2016). More novel interventions have also shown promise in treating PTSD in UK military veterans through multi-modular motion-assisted memory desensitisation and reconsolidation therapy (3MDR) (Hannigan et al., 2023). Where previous research has investigated the use of VR with the treatment of PTSD in war veterans such as the *Bravemind* system (Friedrich, 2016), more recent investigations have also attempted to enhance such systems with the introduction of smells (Freedman et al., 2024), and noninvasive electrical cranial stimulation (van't Wout-Frank et al., 2024).

The main problem with exposure therapy and, by extension, VRET, is that during this treatment, attempts are made to replace a reaction to feared stimuli with a more relaxed reaction by slowly exposing the patient to higher degrees of the feared stimuli while responding with the taught relaxation techniques (Davison, 1968; Markowitz & Fanselow, 2020). The treatment targets only the symptoms (Olatunji et al., 2010) or the elements of the trauma that are directly connected to it (Markowitz & Fanselow, 2020) and not the fundamental cause of the disorder. The efficacy of VRET thus falters because of the context within which the exposure takes place, i.e., the therapist's office, and could lead to returning fear reactions when facing the feared stimuli in a different setting (Markowitz & Fanselow, 2020). Other phobias such as social phobias also don't respond well to exposure therapy, because it assumes in most cases that the only cause of phobias is learning through association (Davison, 1968; McLeod, 2024).

2.3 Spontaneous Healing Intra-systemic Process

Spontaneous Healing Intra-systemic Process (SHIP[®]) is a psychotherapy for the healing of the above-mentioned pathological fear structures or, as referred to in SHIP[®], for trauma-spectrum manifestations (TSMs) (Steenkamp, 2018). TSM is an integrated SHIP[®] diagnosis related to trauma (Steenkamp, 2018), which encompasses manifestations such as PTSD (Steenkamp, 2018). SHIP[®] concerns itself with generating awareness of internal processes within a client to facilitate change by validating those processes (Hoffman & Steyn, 2010). The role of the psychotherapist is thus to facilitate this internal spontaneous healing dialogue for it to move from a state of dysregulation (out of the space of tolerance) to a state of flow, i.e. being able to effectively regulate one's autonomic responses to the external world in an orderly manner (Steenkamp, 2018).

Trauma does not disappear and can resurface because of trauma-activating associations which are normally connected with the original traumatic event in some manner (Steenkamp et al., 2012). If the trauma-activating associations are triggered again, they could stimulate the unfreezing of trauma and present as spontaneous healing reactions (SHRs) (Steenkamp et al., 2012). SHRs are psychobiological responses that are seen as interactive, interconnected energy patterns consisting of the physical, mental, spiritual, and emotional (Kieser-Muller, 2016). SHRs are described as trauma-unfreezing and are a core part of the autonomic regulatory process in an individual's day-to-day experiences and reactions to external events (Steenkamp, 2018). Through trauma-unfreezing, the healing and integration of trauma-material can begin to help an individual to effectively regulate their autonomic responses to the external world (Steenkamp, 2018). The healing and the integration of trauma material have been experientially proven to result in flow and the reciprocal relief from TSM (Steenkamp, 2018).

The key differences between exposure therapy and SHIP[®] are what motivate this research. VRET leads to unsuccessful treatment in some cases because of its approach to treatment (as discussed in Section 2.2). SHIP[®], however, does not aim to habituate the client to the trauma-associative triggers, but rather to allow the trauma to become unstuck and complete its natural process (Steenkamp, 2018). The similarities between exposure therapy and SHIP[®] motivate investigation into whether VR can be used alongside SHIP[®] to assist those who do not benefit from VRET.

3 THE SHIP[®] FRAME: THE DOOR

During the process of SHIP[®], the facilitator can select appropriate trauma-activating associations, or activators, to induce opening of the trauma memory and subsequent motion completion/healing thereof (Steenkamp, 2018). The SHIP[®] Frame consists of a variety of inductivelyderived images and contains a wealth of associative information relating to trauma-material, awhich isnd are used primarily used as an associative activators (Steenkamp, 2010). The SHIP[®] Frame is a medium to transport clients to their internal process of spontaneous healing and makes the process of psychotherapy faster (Steenkamp, 2010). One such image is known as "the door", which involves a client being asked to lie on a bed with their eyes closed and tasked with imagining a door with different words written on it such as the words "*Emotions*" or "*Identity*" (Steenkamp, 2010). This door image is used most frequently and the words written on the door can be substituted with any word that might help them reconnect with the disconnected emotions (Steenkamp, 2010). The client should experience all the emotions and sensations the image evokes, as well as any and all associations, whether it be from the past, present or future must also be identified and acknowledged (Steenkamp, 2010). In doing so, the imagined environment soon corrects itself in a way that the client can experience it in a positive light (Steenkamp, 2010). However, in some cases the client is unable to visualise the images requested during the phases (Steenkamp, 2010) which means that the feelings that were denied expression by the trauma-activating event are kept from completion leading to unsuccessful or ineffective treatment (Steenkamp, 2018). A simulated environment could thus be used as a tool for igniting subsequent layers of spontaneous healing when clients keep struggling to visualise the requested associative images.

3.1 Designing a VR system to assist in the SHIP[®] process

The first objective of a VR system to assist in the SHIP[®] process would be to successfully immerse a user into a VE and control the simulated environment remotely. VRET systems such as the *Bravemind* system (Friedrich, 2016), immerse patients into a controlled, simulated environment through the use of a head-mounted display (HMD) to expose them to their trauma. Because SHIP[®] relies on the visualisation of images from the SHIP[®] Frame (Kieser-Muller, 2016), a SHIP[®] VR system would also be able to utilise an HMD to immerse the client within a VE where the required stimuli can effectively be simulated. A simple, user-friendly interface similar to the sandbox-type public speaking anxiety treatment system (Lindner et al., 2021) would also allow a psychotherapist to control this VE remotely. The door image, as discussed above, can be a predefined scenario such as those designed for the Iraq/Afghanistan system (Rizzo et al., 2015).

Such a VR system would then have to successfully induce physiological and psychological responses from the user. Both SHIP[®] and VRET use visual associative triggers within a simulated/imagined environment to serve as disrupting activators in order to induce activation of the trauma state of dissociation of the client (Steenkamp, 2018). VRET systems such as the system for the treatment of acrophobia (Nabukenya et al., 2021) exposed users to disrupting activators, i.e., various height scenarios, to treat their fear of heights. For a SHIP[®] VR system to successfully induce such activation as mentioned above, images that fall under the client's current phase in the SHIP[®] Frame need to be simulated. For example, simulating something unique to the user on the door, such as their name, could successfully induce activation of frozen trauma (Steenkamp, 2010, 2018).

For a SHIP[®] VR system it is not enough to simply trigger certain experiences of the client, but the VE should also disrupt the existing TSMs. Various VRET systems are able to disrupt fear structures effectively and even change or replace them (da Costa et al., 2018; Gromer et al.,

2018). These disruptions have led to responses such as increased heart rate, and self-described discomfort (da Costa et al., 2018), suggesting that the existing fear structures were triggered. With one VRET system these symptoms became less severe as they progressed through more sessions suggesting that the fear structures were also altered so that the experienced anxiety became less severe (da Costa et al., 2018). Responses such as the ones described above are similar to SHRs experienced by clients as they undergo emotional processing during SHIP[®] sessions (as discussed in Section 2.3). This suggests that a similar VR system might be able to trigger the trauma to induce SHRs to "unfreeze" the trauma within the SHIP[®] process.

Because both SHIP[®] and exposure therapy rely on an individual's ability to imagine the necessary stimuli, it is reasonable to assume that SHIP[®] could share other limitations with exposure therapy. VR has been used in tandem with exposure therapy to effectively overcome some of these limitations, therefore this study set out to investigate whether VR can also assist in overcoming similar limitations in SHIP[®].

4 RESEARCH METHODOLOGY

To determine the induction efficacy of VR within the process of SHIP[®], a RCT design (Hariton & Locascio, 2018; Thiese, 2014) was followed with data gathered of each group's respective experiences through observation, giving this study a qualitative method approach (Pickard, 2013). A total of 28 Participants participants were randomly divided into two groups of 14 participants each, with one group receiving the VR SHIP[®] intervention and a Control group receiving the standard SHIP[®] intervention (Machin et al., 2021) and is, as discussed in detail in Section 4.1.

A SHIP[®] psychotherapist was recruited as facilitator to guide the participants through their sessions using their in-depth SHIP[®] training in the field of image activation in SHIP[®] clients. The facilitator was best suited to observe the participant responses during the sessions and in the unlikely event that the participant had an adverse reaction to the stimuli, the facilitator was capable of guiding the participant away from the stimuli and could assist the participant by referring them to the appropriate professional.

4.1 Participants

Purposive sampling was used to gather participants and was implemented twice: once for the initial group of participants, and again to select the low level of imaginative potential group from the initial group. For the initial group, the participants had to adhere to the following criteria to be considered:

- 1. no prior knowledge of SHIP[®],
- 2. no history of undergoing SHIP[®], and
- 3. no prior trauma or mental disorders.

This sample was used to reduce the differences between the participants to ensure that the data were more significant (Hariton & Locascio, 2018; Thiese, 2014) and to exclude any potential participants who might experience adverse reactions during their sessions. All participants provided their informed consent before the commencement of the study and the study was approved by the institutional ethics review committee (protocol number: EBIT/30/2022). Clearance was not granted to collect demographic information such as gender and age.

The desired population consisted of participants who exhibited the lowest level of imaginative potential of the initial group, i.e., individuals who struggled to imagine requested stimuli the most. To determine the imaginative potential for each of the participants, an imagination exercise was developed and administered by the facilitator for this specific study using their in-depth knowledge of SHIP[®]. Because this method was designed specifically for this study, no benchmark existed for what was to be considered a low imaginative potential score. Therefore the facilitator and researcher decided to recruit the 20 participants from the initial group with the lowest levels of imaginative potential for the remainder of the study. During the imagination exercise, each participant was given a score out of 10 for three different criteria deemed necessary by the facilitator to be viable SHIP[®] candidates:

- 1. level of engagement,
- 2. level of authenticity, and
- 3. level of consistency.

In total, 28 individuals took part in the preliminary imagination exercise. The overall process of the imagination exercise administered to the initial group by the facilitator can be seen in Appendix A. Some of the participants were uncomfortable with lying down flat on the air mattress provided, therefore, the participants were given the option to either lie down on the air mattress flat, sit on the air mattress with their backs against the wall, or sit in the office chair opposite the facilitator. They were also allowed to keep their eyes open during the exercise if they preferred it. This allowed for the participants to feel more in control of the session to encourage higher engagement with the SHIP[®] activities.

Six participants from the initial group were excluded from consideration to take further part in the study. These exclusions were as a result of previous exposure to SHIP[®], unavailability during the designated time frame for the RCT, or exhibiting signs of past or present trauma. Thus, 22 participants were considered for the remainder of the study. The 20 participants who exhibited the lowest levels of visualisation capability were randomly divided up into two groups, namely a 'standard' group – who partook in a standard SHIP[®] session – and a 'VR' group – who partook in a VR SHIP[®] session. The randomisation of the groups was repeated until the mean level of the imaginative potential ratings of the groups were within an acceptable threshold of each other, to ensure that no known difference between the groups existed except for the intervention itself (Thiese, 2014). These groups can be seen in Table 1:

70

Control Grou	up	•	VR Group	
Participant	LIP Rating	_	Participant	LIP Rating
P31 (C01)	8.6		P29 (V01)	7.6
P21 (C02)	4.6		P02 (V02)	9.3
P33 (C03)	8.0		P24 (V03)	8.3
P04 (C04)	9.3		P18 (V04)	8.6
P28 (C05)	8.0		P32 (V05)	8.6
P08 (C06)	9.3		P06 (V06)	6.3
P05 (C07)	8.3		P09 (V07)	9.7
P30 (C08)	9.3		P13 (V08)	2.3
P22 (C09)	5.0		P20 (V09)	9.3
P19 (C10)	8.0		P11 (V10)	6.6
Average	7.84	-	Average	7.66

Table 1:	The final group divided into two groups based o	n
	level of imaginative potential (LIP) rating	

4.2 Materials

The PENSIEVE prototype, designed and developed for this study, consisted of a Meta Quest 2 HMD which ran a VR application using Unreal Engine 4 (UE4) via a Windows PC and was controlled using a simple graphical user interface (GUI) as shown in Figure 1.

PENSIEVE	Client name rendered. Simulation viewable in cast window.	Logout
Please select a participa	 Once the name has faded into view you may begin guiding the client through the environment. When the client reaches a neutral state, click the button below to end the simulation. 	rticipant B
Participan	End Simulation	rticipant D

Figure 1: GUI used by the facilitator to start and end the simulation

The facilitator would click on a participant's name via the GUI, and control the VE through a series of prompts, culminating in the door with the participant's name on it fading from view and into black.

The VE was designed using a VR template and starter content, i.e., a door 3D model, provided by UE4. The application was passive (Vergara et al., 2017) in nature and did not allow for any interaction with the simulated world and did not include audio output.

The main structure of the VE consisted of four walls, a ceiling and a floor that were differentiated through different materials. The main goal of the VE was to instil a high level of presence so as to have the participant respond naturally to it. This was achieved by simulating a realistic 3D model of a door with the participant's name rendered on it and can be seen in Figure 2.



Figure 2: Realistic 3D door with primary investigator's name rendered in large letters

4.3 Procedure

Data were collected via the following procedure: each participant in the initial group was given a written consent form to sign, denoting the potential risks associated with the imagination exercise. The final 20 participants for the RCT trial were recruited after the imagination exercise was given to the initial group. They were given another consent form to sign also explaining the potential risks, and gave permission for the recording of various data throughout the RCT. The participants were informed about the main goal of the RCT. It is important to note that no form of psychotherapeutic treatment was administered to any of the participants.

For the sessions, only a single image out of the SHIP[®] frame was chosen, namely the door image with the participant's name either imagined or simulated on it. Other images in the SHIP[®] Frame were considered to be too ambiguous and might not trigger unique experiences within every participant. The general process followed by the facilitator during the sessions once the participant was seated or lying down was as follows:

1. The facilitator asked them to close their eyes and imagine a door.

- 2. They were then asked if they could see the door, or the door was faded into view. They were then asked if they were able to identify any experienced emotions.
- 3. The facilitator then requested that they imagine their name on the door or rendered a simulation of their name on the door again the facilitator asked if they were able to identify any specific emotion.
- 4. If they were able to identify an emotion, they were asked where in their bodies they felt it.
- 5. If they were able to pinpoint the feeling, they were tasked with focusing on that specific feeling.
- 6. During this the facilitator would periodically ask what thoughts, memories, or sensations they became aware of.
- 7. If they were unable to identify any emotions or sensations, they were asked to just keep focusing on the visualised door or the simulation.

Sessions tended to last between 20 to 30 minutes and varied between participants.

The VR group sessions followed the same process as the Control group's session differing only with regards to the door simulation fading in from black once asked to imagine a door, and fading in the participant's name on the door once asked to imagine their name on the door. The rest of the process as outlined above was then followed.

During each participant's session the facilitator would observe all the responses to fill in a helpful aspects of therapy form (HAT). This questionnaire typically consists of seven openended questions given to the client after a therapy session (Elliott, 2002, 2008) and was used, with incorporated VR terminology, to identify assistive or hindering events to the SHIP[®] process that occurred during therapy sessions. The HAT form used for each session can be found in Appendix B.

A month after the RCT sessions were completed, the first author conducted a semi-structured interview with the facilitator, henceforth referred to as the change interview. A month was chosen to allow the researcher to examine all the HAT forms and transcribe all the audio recordings of the sessions prior to the change interview. The purpose of this interview was to gather insights about the assistive or hindering events observed, what led to these events, and what characteristics were considered to be helpful or hindering (Elliott, 2002). The question list used during the change interview can be found in Appendix C.

The collected data were analysed using thematic analysis by repeatedly going through the collected data, familiarisation with it, generating initial codes, and searching and reviewing recurring themes that appear within the data (Kiger & Varpio, 2020; Nowell et al., 2017; Terry et al., 2017). The full details of how the data were collected and analysed can be found at (Malan, 2024).

5 RESULTS

Within both the control group and the VR group it was observed that all of the participants experienced some form of emotional, physical, or mental response towards the imagined/simulated stimuli. The emotional responses ranged from feelings of euphoria and empowerment to anxiety and fear, the physical responses included sensations felt in the chest, shoulders, hands, and head, and the mental responses included unique memories and thoughts as well as other unique sensory experiences such as sound. Each participant was kept anonymous and denoted with a letter corresponding to their group and a number. For example C01 refers to the first participant in the control group. Some of the most notable responses of the two groups are illustrated in Tables 2 and 3.

Participant	Response	Quote
Emotional R	esponses	
C02	Euphoria	"I guess happy or excited, this particular door's from when I was on holiday, so it was an exciting time"
C09	Fear	"I think one thought that comes to mind is, like, unsure and scared to open the door"
Mental Resp	onses	
C02	Memories	"Well this particular door was a church door and I loved visiting churches when we were in Paris. Um, but this one was in Dusseldorf"
C01	Sensory	"Hearing crackling sounds of the wood"
Physical Res	ponses	
C01	Chest: comfort	"You know, just sort of like in the chest area"
C08	Hands: fear	"It's almost more in the hands now"

Table 2: Notable responses	from	Control	group
----------------------------	------	---------	-------

Table 3: Notable responses fi	from VR group
-------------------------------	---------------

Participant	Response	Quote
Emotional Re	esponses	
V09	Euphoria	"joy"
V05	Fear	<i>"fear, uncertainty. I'm too hesitant. Cautious, worried that it might make it worse"</i>
Mental Respo	onses	
V07	Memories	"My dad passed away, like, two years ago …"
V04	Sensory	"I think of, like, the sounds that you hear, like the birds, but also of the mosquitoes, I was thinking of the itching now"
Physical Resp	oonses	
V10 V02	Shoulders: unease Arms: empowered	<i>"I think in my shoulders and in my chest"</i> <i>"In my chest, and my upper arms"</i>

It is important to note that even if a participant was able to identify specific emotions, or physical sensations, or experience unique mental responses, it does not necessarily imply an effective SHIP[®] session. To determine which responses were assistive to the overall process of SHIP[®] and which ones were hindering, thematic analysis was applied to the HAT forms as well as to the data gathered from the change interview. With regards to the HAT forms, the facilitator scored each session according to the following guidelines:

- 1–4 was considered to be not a useful session at all.
- 5–7 was considered to be a neutral session that didn't advance nor hindered the process of SHIP[®].
- 8–10 was considered to be an extremely useful session to the process of SHIP[®].

The results of the HAT forms are illustrated in Table 4.

Control Grou Participant	ıp Score	_	VR Group Participant	Score
C01	7		V01	7
C02	6		V02	8
C03	6		V03	8
C04	8		V04	7
C05	5		V05	9
C06	7		V06	7
C07	8		V07	7
C08	9		V08	9
C09	9		V09	10
C10	7		V10	6
Average	7.2		Average	7.8

Table 4: HAT session usefulness rating
--

Table 4 illustrates that the facilitator experienced both treatment methods as being assistive to the overall process of SHIP[®] with no one method clearly perceived as superior by the facilitator. However, the HAT scores also provide insight into the individual experiences had by participants, what aspects of VR and the traditional SHIP[®] process led to these experiences and why the facilitator perceived them to be assistive or not.

5.1 Assistive aspects of VR

The assistive events observed by the facilitator were categorised into two main categories, i.e., induced responses and VR attributes. Induced responses included all the responses directly

associated with the assistive events and were categorised as emotional, physical, and mental responses. Attributes of the VR system that were directly associated with assistive events were categorised as simulation and headset. The induced responses and specified VR attributes are seen as assistive aspects of the VR method and are illustrated as all connecting to assist the process of SHIP[®] in Figure 3



Figure 3: Assistive aspects of the VR group

In the sessions of participants V05, V07, and V09, emotional processing (or disruption as discussed in Section 3.1) took place because the initial emotion that was experienced gradually changed or faded as the session progressed. In the case of participant V05, the facilitator noted that: "the door visual (VR) stimulated or triggered a memory of a rugby injury that still brings up fear, disappointment, uncertainty … We were able to use SHIP techniques to process the fear which actualised as restlessness in their legs".

Participant V09 "had a limited experience …" during their imagination exercise, however during their SHIP[®] session they "had a significantly richer and deeper experience". The facilitator explained that "The VR visual seemed to serve as a cognitive stimulus to bring up emotion, memories, or reflections that were less accessible to this particular client in the first round of sessions". Within the HAT form, the facilitator stated the following: "I have a sense that the VR headset or visual does not shape the entire SHIP[®] process but rather acts as a stimulus to initiate a chain of thought, emotions, or reflections …". The facilitator explained this visual stimulation during the change interview as being one of the most helpful aspects of the VR group sessions and that it allowed the VR to function as a tool that led to deeper engagement and an enhanced experience

During the imagination exercise of participant CO3, the facilitator noted that their description was "authentic" and "detailed", although "rather to the point" meaning that they "[might] not have a high visual processing ability". This participant received a high level of imaginative potential rating as their description was "authentic" and it allowed them to engage the SHIP[®] activities completely. However, during the Control group sessions, the facilitator noted that participant CO3 was "not expressive by nature" and that this "limits the SHIP[®] process because we

can express deeply only if we allow ourselves to be vulnerable". Although an inexpressive nature is not directly linked to the success of a traditional SHIP[®] process, it is worth noting that an individual who is unable to effectively express themselves, as well as being unable to formulate more vivid visualisations, could lead to a very ineffective SHIP[®] session. Participant V08 was also inexpressive by nature and struggled with visualising the requested stimuli during the imagination exercise. Participant V08's HAT form reads: "This participant could not visualise a door the first time, this time they could" referring to the participant's VR SHIP® session. They were able to identify a physical sensation towards the end of their session noting that it "Actually feels as if my hands and feet are getting numb". Participant V08 had a richer experience with the VR intervention than during the imagination exercise. The participant mentioned to the facilitator: "... I'm not searching for an image, you have the image and you just focus on that" and during the change interview the facilitator explained that VR "cut out the physical world" ... and it forced your conscious mind to get immersed to be present in the virtual reality world ... and then your energy becomes more concentrated on the task in front of you ... going through all the SHIP processes basically". This is mainly because there was no need for the participant to focus their full concentration on imagining the requested stimuli, but only focus and engage with the SHIP[®] activities posed by the facilitator.

5.2 Hindering aspects

Similar to the assistive events that were discussed above, only the responses and/or attributes of the VR system that were identified to be directly associated with the hindering events, were categorised into induced responses and VR attributes. These categories in turn are what hinder the overall SHIP[®] process, making it less effective. The outcome of this process of identification and categorisation is illustrated in Figure 4. The main hindering aspects of the VR treatment method for SHIP[®], in some cases, was that of the low fidelity of the VE, the point-of-view of the simulation, and the actual VR technology being a distraction for some participants.

Participant V02 noted during their session that "when I open my eyes and I see this picture, it's, like, such a boring door …" and agreed with the facilitator that when they closed their eyes, the door they were able to visualise was much more "vibrant". The facilitator noted in the HAT form that the participant was "often tempted to close their eyes". This event was only described as slightly hindering by the facilitator, but nonetheless the lack of a vibrant, more visually appealing simulation reduced the immersion levels, reducing presence, and could result in their responses to the SHIP[®] activities being less authentic and spontaneous. This could have led to less effective emotional processing.

Additionally, the actual VR hardware seemed to be a distraction in some cases. Throughout the session of participant V07, the weight of the headset started to loosen the straps and caused the headset to sag down on the participant's face. This resulted in the participant having to physically hold up the headset to see the simulation properly. Within the participant's HAT form the facilitator wrote the following: *"Participant preferred to sit up on a chair. The headset*

77





Figure 4: Hindering aspects of the VR group

needed to physically be held in place, because of its weight, to keep the door image orientated. This may have distracted them from the SHIP activities by shifting their focus". Other distractions related to the VR technology that were noted by the facilitator included a power outage experienced by participant V04, the "anxious anticipation" towards VR technology in general by participant V10, and the point-of-view (POV) of participant V05 where they experienced the door as being skewed in the VE instead of upright.

During the change interview the facilitator explained that although the distractions were hindering, they weren't hindering to such an extent that it would completely nullify a SHIP[®] session. The facilitator said that in practice "What I try to do with clients is identify the distractor … Whether it's a headset being too heavy that you need to prop up or it's unexpected and avoidable things … I think in real life as within VR you need to be prepared to cope with unexpected things because that is the nature of the human experience".

However, the most hindering aspect observed by the facilitator, on a psychological level, was considered to be the height of the air mattress the participants were asked to lie down or sit on. During the change interview the facilitator stated that: *"if you think of very caveman, fight-or-flight- type of thinking [sic], if you are on the floor, you're not safe. Because if you need to respond to a risk or a threat, it takes a long time to stand up before you can fight".* With regards to the control group the facilitator noted that some participants *"feel a little bit self-conscious about lying there with their eyes closed … It's a vulnerable position"*. This hindering aspect was however not the case with the VR group according to the facilitator as the VR headset was effective at removing the participant from their immediate reality. VR's ability to 'cut-off' participants from the external world, was beneficial to the SHIP[®] process and is discussed in more detail in the following section.

5.3 Immersion, presence, and engagement

The facilitator suggested that the VR group experienced higher levels of immersion, and felt that it was the VE simulated by the VR headset that contributed to this. They noted that they "...found that the VR group was able to go a little bit deeper ... that is accounted for by the additional brain stimulation that the VR headset provided". Thus, when considering the most hindering aspect noted by the facilitator across both groups, i.e. the height of the air mattress, it suggests that the hindering aspects of both groups were, in a sense, experienced more vividly by the Control group than the VR group. The VR group was able to escape those hindrances more effectively because of the visual simulation and the nature of the VR headset. Because of the physical nature of the HMD, it effectively "cuts off" the user's peripheral vision, removing them from their immediate surroundings.

One implication of 'removing the participant from the real world' was that it empowers an individual to become fully present in the VE. As discussed in Section 5.1, it allows one to focus all one's energy onto the task at hand, in this case the SHIP[®] activities. According to the facilitator, the VR headset and the simulation were the most helpful contributors to the SHIP[®] process: "I think the VR goggles served as almost a healthy distraction because it allowed them to immerse themselves into the VR world". The facilitator also noted that "reducing your consciousness to the VR world, I think that was helpful in enhancing presence". In Section 2.1, it was discussed that an individual's level of presence experienced comes from the degree to which they can respond naturally in virtual environments. It stands to reason then that if an individual experiences a greater sense of presence, they are more likely to respond more naturally to the simulated environment. Being "cut-off" from the distractions of the physical world leads to a greater feeling of "being there" within the simulated environment. In the context of SHIP[®] (as discussed in Section 3), this heightened sense of presence would allow the participants to respond more naturally to the SHIP® activities. It is important to note that successful emotional processing in the context of SHIP® is not solely dependent on the level of the engagement with the imagined stimuli, but also on the authenticity of the spontaneous responses to the SHIP[®] activities administered in tandem with a client's experience with the imagined stimuli. Thus, the more naturally a client can respond to their environment, the more authentic their responses are towards the SHIP[®] activities which could lead to successful emotional processing.

5.4 VR as therapeutic tool

According to the facilitator, both groups engaged the SHIP[®] activities to "the best of their ability" and that the level of engagement was good overall. This is largely due to the role of therapeutic alliance in psychotherapy. The facilitator said during the change interview that "Therapy requires vulnerability, the therapeutic alliance facilitates the vulnerability … therapeutic alliance, you'll find that it's a massive contribution to absolutely any therapeutic process …". This contribution was noted during participant CO4's session, where the participant was very honest about not experiencing any emotions or sensations at the start of the session. According to

the facilitator, this admission allowed them to *"build a good therapeutic alliance"*. This enabled the facilitator to engage the participant more honestly, and the participant was then able to engage in the SHIP[®] activities better as the session progressed, which allowed them to effectively process their emotions that eventually came up.

This therapeutic alliance existed between the facilitator and the participants in the VR group as well. For example, in the session with participant V01, the facilitator noted that the participant was *"very much at ease or safe during the session"* within the VE and therapeutic space. This feeling, in combination with the therapeutic alliance, led to the participant sharing *"personal information or thoughts and reflections"*. This therapeutic alliance could also have been enhanced by the physical nature of VR *"cutting off"* the external world. The facilitator elaborated that *"... they don't know what I'm seeing and it's not a shared experience, and that maybe allows them to feel a sense of, of psychological safety"*. The facilitator is alluding to the possibility that a client might experience images and thoughts that are private, and being removed from their immediate environment makes them feel as though their private thoughts stay private. Therapeutic alliance existed between the facilitator and the participants in the VR group similar to the control group, despite the added unfamiliarity of the VR technology.

During this section it was clearly outlined how the VR intervention was able to foster the same level of therapeutic alliance as the traditional SHIP[®] method. Section 5.2 revealed that the most hindering aspect of both methods was psychological in nature, but that the impact of this hindering aspect was lessened by the technological and physical characteristics of VR as examined in Section 5.1. Why these characteristics were able to reduce the impact was then explored in Section 5.3 and it was found that the physical nature of the HMD and the VE simulated were directly responsible for reducing the impact of the psychological hindrances. VR was thus able to overcome a psychological obstacle faced by the traditional SHIP[®] method. This, along with the proven induction efficacy of unique memories of the VR group as stated by the facilitator, namely *"the VR induction was able to allow us to stimulate very significant memories"* leads to the notion of VR potentially being used as a tool to assist the existing SHIP[®] process rather than shaping it.

6 LIMITATIONS

While the prototype successfully achieved its goal of inducing unique memories, and physiological and psychological responses, the sample for the current study, the simplistic nature of the imagination exercise, and the chosen SHIP[®] Frame limited the results of the study somewhat from preventing to determineby making it difficult to determine which of the two intervention methods were more assistive to the overall SHIP[®] process. These limitations impacted the results as follows:

• *Small sample size*: the final group of 20 participants that took part in the RCT were recruited from a small initial group of 22 individuals. This prevented the researcher from recruiting only participants who truly had difficulty with imagining requested stimuli.

80

As a result mostly participants who did not really have any visualisation deficiencies took part in the RCT which lessened the credibility of the results.

- *Simplistic imagination exercise*: the researcher determined that the oversimplified imagination exercise is what led to most of the participants obtaining high levels of imaginative potential scores. As a result the *"low"* level of imaginative potential group consisted mostly of participants who did not have trouble visualising the requested stimuli. This lessened the credibility of the results as we could not effectively measure an improvement between a participant's ability to visualise before and after the RCT.
- *Shallow SHIP*[®] *Frame*: the door image chosen for the current study is quite general, even with the addition of the participant's name on the door. Thus, it is this researcher's belief that a more personalised VE would have garnered even richer data.

These limitations outlined above assist in identifying the key areas that warrant further research and are discussed in the following section.

7 CONCLUSION AND FUTURE WORK

This paper describes the PENSIEVE prototype, a software intervention that allowed a SHIP[®] facilitator to immerse participants into a VE to experience specific stimuli based on the SHIP[®] Frame. Qualitative data were gathered of participants' physiological and psychological responses to the imagined or simulated stimuli during a SHIP[®] session. The results of this study suggests that VR has the potential to stimulate and enhance the visualisation capabilities of the client or act as a proxy for a client's imagination. In other words, VR can aid participants who have difficulty visualising requested images and in doing so aid them in engaging with the SHIP[®] activities and successfully progressing through the process of SHIP[®]. Thus, a key difference to consider between the two methods is that the induction of unique memories would be impossible for SHIP[®] clients with visualisation deficiencies, an obstacle that was overcome by the PENSIEVE prototype.

Simulating a virtual world through the use of an HMD effectively "*cut off*" participants from their immediate environment. Distractions and other hindering aspects identified were reduced or completely removed for participants. This enhanced the existing therapeutic alliance between the facilitator and the participant, leading to high levels of presence, and allowing participants to experience authentic responses towards the stimuli and SHIP[®] activities. In addition to this, the prototype was able to replicate every aspect of the traditional SHIP[®] process that is needed for successful treatment. Thus, the prototype was successful in simulating a VE that was able to induce unique memories and responses necessary for the SHIP[®] process.

Further research might include investigating the extent of VR's role as cognitive proxy for a SHIP[®] client's imagination by recruiting a larger sample via a more complex imagination exercise to ensure only participants who cannot visualise effectively take part in the study. Other research avenues might include, but are not limited to, a purely experimental approach based

on the induction potential of both interventions and compare them directly with one another. A quantitative study such as this can also take the form of a pre- and post-test comparison to determine if there was any improvement in the imaginative state of a participant. Lastly, a more extensive study could also be carried out focusing on the improvement of a client's disorder by observing the client over the span of multiple SHIP[®] sessions, while each client is experiencing a personalised virtual world developed specifically for said client. A study such as this would provide valuable insights into the extent to which VR is able to improve a client's specific mental disorder. Another consideration for such an investigation is to determine to what extent collaboration would be necessary between the psychologist and the VR developer/ operator. The integration limitations such as the therapeutic space i.e. the psychologist's office (or wherever they administer treatment) and the psychologist's technical capabilities (with regards to operating a VR system) would also warrant thorough investigation.

References

- Albakri, G., Bouaziz, R., Alharthi, W., Kammoun, S., Al-Sarem, M., Saeed, F., & Hadwan, M. (2022). Phobia exposure therapy using virtual and augmented reality: A systematic review. *Applied Sciences*, 12(3), 1672. https://doi.org/10.3390/app12031672
- Beidel, D. C., Frueh, B. C., Neer, S. M., Bowers, C. A., Trachik, B., Uhde, T. W., & Grubaugh, A. (2019). Trauma management therapy with virtual-reality augmented exposure therapy for combat-related PTSD: A randomized controlled trial. *Journal of Anxiety Disorders*, 61, 64–74. https://doi.org/10.1016/j.janxdis.2017.08.005
- Botella, C., Fernández-Álvarez, J., Guillén, V., García-Palacios, A., & Baños, R. (2017). Recent progress in virtual reality exposure therapy for phobias: A systematic review. *Current Psychiatry Reports*, *19*(7), 42. https://doi.org/10.1007/s11920-017-0788-4
- Bryant, R. A., Kenny, L., Rawson, N., Cahill, C., Joscelyne, A., Garber, B., Tockar, J., Dawson, K., & Nickerson, A. (2019). Efficacy of exposure-based cognitive behaviour therapy for post-traumatic stress disorder in emergency service personnel: A randomised clinical trial. *Psychological Medicine*, 49(9), 1565–1573. https://doi.org/10.1017/S00332917 18002234
- da Costa, R. T., de Carvalho, M. R., Ribeiro, P., & Nardi, A. E. (2018). Virtual reality exposure therapy for fear of driving: Analysis of clinical characteristics, physiological response, and sense of presence. *Brazilian Journal of Psychiatry*, *40*, 192–199. https://doi.org/10.1590/1516-4446-2017-2270
- Davison, G. C. (1968). Systematic desensitization as a counterconditioning process. *Journal of Abnormal Psychology*, *73*(2), 91–99. https://doi.org/10.1037/h0025501
- de Regt, A., Barnes, S. J., & Plangger, K. (2020). The virtual reality value chain. *Business Horizons*, *63*(6), 737–748. https://doi.org/10.1016/j.bushor.2020.08.002
- Donker, T., Cornelisz, I., van Klaveren, C., van Straten, A., Carlbring, P., Cuijpers, P., & van Gelder, J. L. (2019). Effectiveness of Self-guided app-based virtual reality cognitive

behavior therapy for acrophobia: A randomized clinical trial. *JAMA Psychiatry*, *76*(7), 682–690. https://doi.org/10.1001/jamapsychiatry.2019.0219

- Elliott, R. (2002). Hermeneutic single-case efficacy design. *Psychotherapy research : Journal of the Society for Psychotherapy Research*, *12*, 1–21. https://doi.org/10.1080/713869614
- Elliott, R. (2008). Helpful aspects of therapy form [Accessed: 1 December 2021]. https://ww w.experiential-researchers.org/instruments/elliott/hat.html
- Eshuis, L. V., van Gelderen, M. J., van Zuiden, M., Nijdam, M. J., Vermetten, E., Olff, M., & Bakker, A. (2021). Efficacy of immersive ptsd treatments: A systematic review of virtual and augmented reality exposure therapy and a meta-analysis of virtual reality exposure therapy. *Journal of Psychiatric Research*, *143*, 516–527. https://doi.org/10.1016/j.jps ychires.2020.11.030
- Foa, E., & Mclean, C. (2015). The efficacy of exposure therapy for anxiety-related disorders and its underlying mechanisms: The case of OCD and PTSD. *Annual review of clinical psychology*, 12. https://doi.org/10.1146/annurev-clinpsy-021815-093533
- Freedman, S. A., Dayan, E., Senitsky, M., Bellin, E., Attias, Y., Richman, T., Eitan, R., & Tatsa-Laur, L. (2024). Case report: The addition of olfaction to virtual reality enhanced exposure therapy for PTSD. *Frontiers in Virtual Reality*, 5. https://doi.org/10.3389/frvir .2024.1284299
- Friedrich, K. (2016). Therapeutic Media: Treating PTSD with virtual reality exposure therapy. *MediaTropes*, 6(1), 86–113. https://mediatropes.com/index.php/Mediatropes/article /view/27102
- Friston, K. (2012). Embodied inference and spatial cognition. *Cognitive Processing*, *13*(1), 171–177. https://doi.org/10.1007/s10339-012-0519-z
- Garrett, B., Taverner, T., Gromala, D., Tao, G., Cordingley, E., & Sun, C. (2018). Virtual reality clinical research: Promises and challenges. *JMIR Serious Games*, 6(4). https://doi.org /10.2196/10839
- Gromer, D., Madeira, O., Gast, P., Nehfischer, M., Jost, M., Müller, M., Mühlberger, A., & Pauli, P. (2018). Height simulation in a virtual reality CAVE system: Validity of fear responses and effects of an immersion manipulation. *Frontiers in Human Neuroscience*, 12. https://doi.org/10.3389/fnhum.2018.00372
- Hannigan, B., van Deursen, R., Barawi, K., Kitchiner, N., & Bisson, J. I. (2023). Factors associated with the outcomes of a novel virtual reality therapy for military veterans with PTSD: Theory development using a mixed methods analysis. *PLOS ONE*, 18(5), e0285763. https://doi.org/10.1371/journal.pone.0285763
- Hariton, E., & Locascio, J. J. (2018). Randomised controlled trials—the gold standard for effectiveness research. *BJOG: An international journal of obstetrics and gynaecology*, 125(13), 1716. https://doi.org/10.1111/1471-0528.15199
- Hodges, L., Kooper, R., Meyer, T., Rothbaum, B., Opdyke, D., de Graaff, J., Williford, J., & North, M. (1995). Virtual environments for treating the fear of heights. *Computer*, 28(7), 27–34. https://doi.org/10.1109/2.391038

- Høeg, E. R., Povlsen, T. M., Bruun-Pedersen, J. R., Lange, B., Nilsson, N. C., Haugaard, K. B., Faber, S. M., Hansen, S. W., Kimby, C. K., & Serafin, S. (2021). System immersion in virtual reality-based rehabilitation of motor function in older adults: A systematic review and meta-analysis. Frontiers in Virtual Reality, 2, 30. https://doi.org/10.3389/f rvir.2021.647993
- Hoffman, D., & Steyn, B. (2010). The effect of SHIP[®] (Spontaneous Healing Intrasystemic Process) on adolescent tennis players. African Journal for Physical, Health Education, Recreation and Dance, 16(1). https://doi.org/10.4314/ajpherd.v16i1.53315
- Kelly, N. J., Hallam, J., & Bignell, S. (2023). Using interpretative phenomenological analysis to gain a qualitative understanding of presence in virtual reality. Virtual Reality, 27(2), 1173-1185. https://doi.org/10.1007/s10055-022-00719-2
- Kieser-Muller, C. (2016). Facilitation of spontaneous healing in an ageing individual presenting with complex PTSD: A SHIP[®] perspective [Doctoral dissertation, North-West University]. https://repository.nwu.ac.za/handle/10394/25192
- Kiger, M., & Varpio, L. (2020). Thematic analysis of qualitative data: Amee guide no. 131. Medical teacher, 42(8), 846–854. https://doi.org/10.1080/0142159X.2020.1755030
- Kober, S. E., Kurzmann, J., & Neuper, C. (2012). Cortical correlate of spatial presence in 2d and 3d interactive virtual reality: An eeg study. International Journal of Psychophysiology, 83(3), 365-374. https://doi.org/10.1016/j.jpsycho.2011.12.003
- Levinson, C. A., Christian, C., Ram, S. S., Vanzhula, I., Brosof, L. C., Michelson, L. P., & Williams, B. M. (2020). Eating disorder symptoms and core eating disorder fears decrease during online imaginal exposure therapy for eating disorders. Journal of Affective Disorders, 276, 585–591. https://doi.org/10.1016/j.jad.2020.07.075
- Lindner, P., Dagöö, J., Hamilton, W., Miloff, A., Andersson, G., Schill, A., & Carlbring, P. (2021). Virtual reality exposure therapy for public speaking anxiety in routine care: A singlesubject effectiveness trial. Cognitive Behaviour Therapy, 50(1), 67–87. https://doi.org /10.1080/16506073.2020.1795240
- Lønne, T. F., Karlsen, H. R., Langvik, E., & Saksvik-Lehouillier, I. (2023). The effect of immersion on sense of presence and affect when experiencing an educational scenario in virtual reality: A randomized controlled study. *Heliyon*, 9(6), e17196. https://doi.org /10.1016/j.heliyon.2023.e17196
- Machin, D., Fayers, P. M., & Tai, B. C. (2021, April). Randomised clinical trials: Design, practice and reporting (2nd ed.). John Wiley & Sons.
- Malan, A. W. (2024, June). The PENSIEVE : Psychological quasi-experimental neural study of induction efficacy of virtual environments [Data set]. https://doi.org/10.25403/UPre searchdata.26490982
- Markowitz, S., & Fanselow, M. (2020). Exposure therapy for post-traumatic stress disorder: Factors of limited success and possible alternative treatment. Brain Sciences, 10(3), 167. https://doi.org/10.3390/brainsci10030167
- McLeod, S. (2024). Classical conditioning: How it works with examples [Accessed: 5 May 2025]. https://www.simplypsychology.org/classical-conditioning.html

84

- Muhanna, M. A. (2015). Virtual reality and the cave: Taxonomy, interaction challenges and research directions. *Journal of King Saud University Computer and Information Sciences*, 27(3), 344–361. https://doi.org/10.1016/j.jksuci.2014.03.023
- Nabukenya, P., Zuanni, F., Jadidi, O., & Brazzoli, A. (2021, January). Multisensory interactive systems combined with vr gradual exposure therapy to treat patients with acrophobia: Potentialities and challenges (tech. rep.). University of Trento, Italy. https://doi.org/10.131 40/RG.2.2.27917.26084
- Nilsson, N. C., Nordahl, R., & Serafin, S. (2016). Immersion revisited: A review of existing definitions of immersion and their relation to different theories of presence. *Human Technology*, 12(2), 108–134. https://doi.org/http://dx.doi.org/10.17011/ht/urn.201 611174652
- Nowell, L., Norris, J., White, D., & Moules, N. (2017). Thematic analysis: Striving to meet the trustworthiness criteria. *International journal of qualitative methods*, *16*(1), 1–13. https://doi.org/10.1177/1609406917733847
- Olatunji, B. O., Cisler, J. M., & Deacon, B. J. (2010). Efficacy of cognitive behavioral therapy for anxiety disorders: A review of meta-analytic findings. *Psychiatric Clinics*, *33*(3), 557–577. https://doi.org/10.1016/j.psc.2010.04.002
- Opriș, D., Pintea, S., García-Palacios, A., Botella, C., Szamosközi, Ş., & David, D. (2012). Virtual reality exposure therapy in anxiety disorders: A quantitative meta-analysis. *Depression and Anxiety*, *29*(2), 85–93. https://doi.org/10.1002/da.20910
- Pickard, A. J. (2013). Research methods in information (2nd ed.). Facet.
- Rauch, S., & Foa, E. (2006). Emotional processing theory (ept) and exposure therapy for PTSD. *Journal of Contemporary Psychotherapy*, *36*(2), 61–65. https://doi.org/10.1007/s108 79-006-9008-y
- Riva, G. (2022). Virtual reality in clinical psychology. *Reference Module in Neuroscience and Biobehavioral Psychology*, B978-0-12-818697-8.00006–6. https://doi.org/10.1016/B9 78-0-12-818697-8.00006-6
- Riva, G., Wiederhold, B. K., & Mantovani, F. (2019). Neuroscience of virtual reality: From virtual exposure to embodied medicine. *Cyberpsychology, Behavior, and Social Networking*, 22(1), 82–96. https://doi.org/10.1089/cyber.2017.29099.gri
- Rizzo, A., Cukor, J., Gerardi, M., Alley, S., Reist, C., Roy, M., Rothbaum, B. O., & Difede, J. (2015). Virtual reality exposure for PTSD due to military combat and terrorist attacks. *Journal of Contemporary Psychotherapy*, 45(4), 255–264. https://doi.org/10.1007/s10 879-015-9306-3
- Sherman, W. R., & Craig, A. B. (2003). Understanding virtual reality : Interface, application, and design. Morgan Kaufmann.
- Slater, M., Lotto, B., Arnold, M. M., & Sanchez-Vives, M. V. (2009). How we experience immersive virtual environments: The concept of presence and its measurement. *Anuario de Psicología*, 40(2), 193–210. https://hdl.handle.net/2445/49643
- Steenkamp, J. (2010, January). SHIP (Spontaneous Healing Intrasystemic Healing): The age-old art of facilitating healing. JOS.

- Steenkamp, J. (2018). An integrated theory & psychotherapy for trauma-spectrum manifestations (1st ed.). JOS: Pretoria.
- Steenkamp, J., van der Walt, M. J., Schoeman-Steenkamp, E. M., & Strydom, I. (2012). Introducing SHIP[®] as a psychotherapeutic model to access the body memory of traumatised clients: Depathologising expressions of trauma. *South African Journal of Psychology*, 42(2), 202–213. https://doi.org/10.1177/008124631204200207
- Suied, C., Drettakis, G., Warusfel, O., & Viaud-Delmon, I. (2013). Auditory-visual virtual reality as a diagnostic and therapeutic tool for cynophobia. *Cyberpsychology, Behavior, and Social Networking*, *16*(2), 145–152. https://doi.org/10.1089/cyber.2012.1568
- Terry, G., Hayfield, N., Clarke, V., & Braun, V. (2017). Thematic analysis. In *The SAGE hand-book of qualitative research in psychology* (2nd ed., pp. 17–37). SAGE. https://doi.org/https://doi.org/10.4135/9781526405555.n2
- Thiese, M. S. (2014). Observational and interventional study design types; an overview. *Biochemia Medica*, 24(2), 199–210. https://doi.org/10.11613/BM.2014.022
- van't Wout-Frank, M., Arulpragasam, A. R., Faucher, C., Aiken, E., Shea, M. T., Jones, R. N., Greenberg, B. D., & Philip, N. S. (2024). Virtual reality and transcranial direct current stimulation for posttraumatic stress disorder: A randomized clinical trial. *JAMA Psychi atry*, *81*(5), 437–446. https://doi.org/10.1001/jamapsychiatry.2023.5661
- Vergara, D., Rubio, M. P., & Lorenzo, M. (2017). On the design of virtual reality learning environments in engineering. *Multimodal Technologies and Interaction*, 1(2), 11. https: //doi.org/10.3390/mti1020011
- Vienne, C., Masfrand, S., Bourdin, C., & Vercher, J. L. (2020). Depth perception in virtual reality systems: Effect of screen distance, environment richness and display factors. *IEEE Access*, 8, 29099–29110. https://doi.org/10.1109/ACCESS.2020.2972122
- Volovik, M. G., Belova, A. N., Kuznetsov, A. N., Polevaia, A. V., Vorobyova, O. V., & Khalak, M. E. (2023). Use of virtual reality techniques to rehabilitate military veterans with post-traumatic stress disorder (review). *Sovremennye Tekhnologii v Meditsine*, 15(1), 74–85. https://doi.org/10.17691/STM2023.15.1.08
- Wilson, C. J., & Soranzo, A. (2015). The use of virtual reality in psychology: A case study in visual perception. *Computational and Mathematical Methods in Medicine*, 2015, 1–7. https://doi.org/10.1155/2015/151702

IMAGINATION EXERCISE PROCESS Α

- Step 1: The facilitator asked the participant to lie down on an air mattress and relax their bodies and to try and become aware of themselves within the space.
- Step 2: Once the participant was relaxed, the facilitator asked them to imagine a door.
- Step 3: After a couple of minutes, the facilitator asked the participant to describe the door in as much detail as possible.
- Step 4: Once the participant had finished with their description, they were asked to just relax and keep imagining their door.
- Step 5: After about 10 minutes the facilitator asked the participant to describe the door again.

https://doi.org/10.18489/sacj.v37i1.19286

87

B HAT FORM

We are interested in determining whether virtual environments are able to induce memories in the same manner as a traditional, imaginal SHIP® session. The questions are applicable to the type of session you just facilitated. This means that if you just facilitated a VR session, all the questions will refer to that specific method of treatment and ONLY that method of treatment.

We use the term "EVENT(S)" to refer to anything that happened during the session with the participant, be it a physiological or psychological response, something the participant said, or anything else they might do that is a result of the specific treatment being administered. It will be up to you to use your insights as a trained SHIP® therapist to observe and identify any and all of these events that might be considered useful or hindering.

There is a single Likert scale question at the beginning of the questionnaire, followed by 6 open ended questions. There are no right or wrong answers. Any event that you consider to be useful or hindering must be listed and described in as much detail as possible. This questionnaire should not be shared with the participant and should only be seen and filled in by you, the facilitator.

1. To what extent was the session useful to the process of SHIP[®]? Rate it on the following scale. (Put an "X" at the appropriate point)

Not at a	Not at all useful		Neutral					remely us	eful
1	2	3	4	5	6	7	8	9	10

- 2. Of the events which occurred in this session, which one do you feel was the most important or helpful with regards to facilitating the session? (By "event" we mean something that happened in the session related to the type of treatment administered i.e., VR or standard. It might be something the participant said or did, or something you said or did.)
- 3. Please describe what made this event important/helpful and what it helped achieve in the context of the SHIP® session.
- 4. During which section of the session did this event occur?
- 5. How long did the event last?
- 6. Did anything else particularly helpful happen during this session that was related to the treatment method administered i.e. VR or standard? (Put an "X" at the appropriate box)

YES	
NO	

- (a) If yes, please rate how helpful this event was: (Put an "X" at the appropriate box)

 Slightly
 Moderately
 Greatly
 Extremely
- (b) Please describe the event briefly and explain why you consider it helpful.
- 7. Did anything happen during the session which might have been hindering that was related to the treatment method administered i.e. VR or standard? (Put an "X" at the appropriate box)

YES	
NO	

- (a) If yes, please rate how hindering this event was: (Put an "X" at the appropriate box)

 Slightly
 Moderately
 Greatly
 Extremely
- (b) Please describe the event briefly and explain why you consider it hindering.

C APPENDIX: CHANGE INTERVIEW QUESTIONS

Note:

- The interview will be semi-structured with the questions listed below.
- The interview will only be conducted once with the facilitator that facilitated all the sessions.
- The interview will take place in the SHIP® facilitator's office and the audio of the interview will be recorded.
- Refreshments will be given to the facilitator and researcher before the interview commences.
- Information regarding the helpful aspects of the two treatments will be gathered from the HAT forms filled in by the facilitator after each session to assist in the interview and to help guide it to uncover new information and to compare the two treatments effectively.

Instructions to be read to the interviewee before the interview starts:

Just to reiterate, we are interested in the overall experience of each of the treatment methods and which aspects of each lead to your experience. The focus is on comparing the helpful and hindering aspects of both treatments to determine the viability of utilizing a virtual environment in the context of the SHIP[®] process. There are no incorrect answers during this interview. The only correct answers are your experiences of each treatment based on your professional opinion as a trained SHIP[®] therapist.

Section A: General experience

Question 1

The HAT forms you filled in for the standard induction indicated that you experienced the sessions as X*. Can you please explain this overall experience of the standard induction?

*X will refer to the overall perceived usefulness of the standard induction as indicated on the HAT forms Likert scale

Question 2

The HAT forms also indicated that you experienced the VR induction sessions as Y*. Can you please explain this overall experience of the VR induction?

*Y will refer to the overall perceived usefulness of the VR induction as indicated on the HAT forms Likert scale

Section B: Attributions

Question 3

What events occurred during the standard induction sessions that attributed to this overall experience?

Question 4

What events occurred during the VR induction sessions that attributed to your overall experience of the VR sessions?

Section C: Helpful aspects

Question 5

What were the most helpful aspects of the standard induction? Why were they helpful?

Question 6

What were the most helpful aspects of the VR induction? Why were they helpful?

Section D: Hindering aspects or lacking aspects

Question 7

What were the most hindering aspects of the standard induction, or did you find any aspects missing during the standard induction?

Question 8

What were the most hindering aspects of the VR induction, or did you find any aspects missing during the VR induction?

Question 9

What aspects of the standard induction made the process more difficult, but that you still perceived as helpful or "OK"?

Question 10

What aspects of the VR induction made the process more difficult, but that you still perceived as helpful or "OK"?

Section E: Immersion, presence, and level of descriptive detail

Question 11

What differences did you experience between the standard and VR induction in terms of the level of immersion of the participants?

Question 12

During which treatment do you think the participants were more involved/engaged with the imagined/simulated stimuli during the sessions?

Question 13

What aspects of the chosen treatment at Q12 do you think contributed to this?

Question 14

On a scale of 1 to 10, how well did the participants describe their emotional or physiological responses experienced during the standard induction? Please explain your answer.

Question 15

On a scale of 1 to 10, how well did the participants describe their emotional or physiological responses experienced during the VR induction? Please explain your answer.

Section F: Effects on the induction of memories

Question 16

What effect on the induction of unique memories did the standard induction have on participants?

Question 17

What effect on the induction of unique memories did the VR induction have on participants?

Section G: Final thoughts

Question 18

Do you have any suggestions regarding the research or the VR treatment as to how to improve it?

- a) Research:
 - The participant sessions.
 - The virtual environment.
 - The sample size and recruitment criteria.
- b) The VR induction

Question 19

In your professional opinion, how successful do you think the VR induction is compared to the standard induction?

Question 20

Is there anything else that you would like to add?

A multilevel analysis of digital technology to support teachers to improve their professional practice

Reuben Dlamini^(D), Brahm Fleisch^(D)

Wits School of Education, University of the Witwatersrand, South Africa

ABSTRACT

Within the education sector, digital technology has increasingly entered into school subjects, curricula, assessment and teaching methods. The common features of technology-orientated and supported programmes are the independence from geographic boundaries and time constraints. Covid-19 pandemic accelerated the debates about digitalisation in the education sector. Therefore, the purpose of this study was to conduct a multilevel analysis of evidence from a large-scale randomised control trial complemented with the systematic collection of implementation information to compare the impact of a conventional structured pedagogic programme to that of a pedagogically similar programme built with a digital technology overlay. The technology acceptance model framed our analysis. Results from the multilevel analysis provided an opportunity to advance knowledge about teachers' acceptance of digital technology to improve student learning and their professional practice. The evidence pointed to two hypotheses: While much of the literature on teachers' acceptance of digital technology concentrated on the subjective element of teachers' behavioural intentions, the current study suggests the extent to which professional accountability may be a key factor in successful digital technology interventions for teachers; and the results points to substantial differentiated uptake that is not linked to either perceived usefulness or perceived ease of use.

Keywords Digital Technology Intervention, Randomised Control Trial, Structured Pedagogical Programme, Technology Acceptance Model, Virtual Coaching

 $\textbf{Categories} \quad \textbf{•} \textbf{ Applied computing} \sim \textbf{Education} \quad \textbf{•} \textbf{ Applied computing} \sim \textbf{Digitalisation}$

Email	Article history	
Reuben Dlamini – reuben.dlamini@wits.ac.za (CORRESPONDING)	Received:	5 September 2023
Brahm Fleisch – brahm.fleisch@wits.ac.za	Accepted:	7 October 2024
	Online:	2 July 2025

1 INTRODUCTION

Computers, and more generally digital technology, permeate all aspects of our contemporary world. Within the education sector, information and communication technology (ICT) has increasingly entered into school subjects, curricula, assessment, teaching methods, and pedagogy. Specifically, few education systems can avoid the pressure to include computer literacy

Copyright © the author(s); published under a Creative Commons NonCommercial 4.0 License (3) (SACJ is a publication of SAICSIT. ISSN 1015-7999 (print) ISSN 2313-7835 (online)

Dlamini, R. and Fleisch, B. (2025). A multilevel analysis of digital technology to support teachers to improve their professional practice. *South African Computer Journal* 37(1), 94–119. https://doi.org/10.18489/sacj.v3 7i1.20602

as a new aim of schooling. The Covid-19 pandemic accelerated the debates about the use of digital technology to connect learners at a distance and providing support to teachers to improve their practice. Digital technology is *"viewed as a solution to systemic inequalities"* (Mhlongo & Dlamini, 2022, p. 1). Within the field of computers and education, one of the main research concerns is the challenges that teachers face adopting technology in their classrooms (Dlamini, 2022). Within this field, the conceptual framework, the technology acceptance model (TAM), has gained considerable ground. TAM research is focused on better understanding how perceived usefulness and ease of use influence the adoption of technology in the workplace. Systematic review and meta-analysis of TAM for teachers largely confirmed earlier individual studies of the usefulness of an analysis of the TAM variables to explain the uneven uptake in the education sector (Scherer et al., 2019).

This paper contributes to the knowledge base of the conditions under which teachers teaching in poor communities in the Global South take up digital technology as a means to improve learning outcomes. The uneven take-up of technology is often attributed to issues of access in digital resource-constrained conditions (Brown & Czerniewicz, 2010). Using evidence from a large-scale randomised control trial (RCT), the aim of which was to compare the impact of a conventional structured pedagogic programme (CSPP) to a pedagogically similar programme built with a digital technology overlay, the study provided an opportunity to advance knowledge about teachers' acceptance of digital technology as a means to improve student learning. The original RCT study found that after three years, average student performance was significantly higher in the schools that received the CSPP relative to the control schools, but that average student performance in the digital technology intervention schools was only marginally higher than the average performance in the control schools (Cilliers et al., 2021). Given the initial positive findings in the first year and the considerable interest in the use of digital technology to improve teaching, it is not sufficient to know that the intervention had minimal impact; it is critical to know why and to better understand why the digital technology component did not add value. Deaton and Cartwright (2018, p. 2) noted the following:

RCTs can play a role in building scientific knowledge and useful predictions but they can only do so as part of a cumulative programme, combining with other methods, including conceptual and theoretical development, to discover not 'what works', but 'why things work.'

The purpose of this paper was thus to analyse the evidence that may explain why the digital technology intervention, which showed initial promise, came up short. Two hypotheses are offered in this paper. Firstly, while much of the literature on teachers' acceptance of digital technology concentrated on the subjective element of their behavioural intentions, the evidence in this study pointed to the centrality of professional accountability in the digital technology interventions. The second hypothesis points to the need to better understand factors associated with differential uptake not linked to either perceived usefulness or ease of use.

The paper is divided into four sections. The first section reviews the literature of both structured pedagogic programme models and the emerging literature on the use of digital technology in instructional coaching. This literature provides a useful context against which the study is framed. The second section of the paper describes the original research study itself, particularly the context and the methodology used in the design and implementation of the study. Although the main study focused on comparing learning outcomes in the three research groups, the current paper uses extensive implementation data to better understand teachers' engagement with digital technology. The third section revisits the main findings but concentrates on an analysis of both the quantitative and qualitative secondary findings on the uptake of technology and links to changes in instructional practices, and, ultimately, to learning outcomes. This is followed by the fourth section, which discusses these findings and hypothesis building.

2 LITERATURE REVIEW

While the scale of the early grade reading crisis was highlighted recently (Azevedo et al., 2019), significant advances have been made in understanding the problem of early learning in lowand lower middle-income country contexts. Recent reviews of research consistently and convincingly demonstrated that structured pedagogic programmes are likely to be the most effective approaches to improving early grade reading outcomes and possibly early mathematics as well (Evans & Popova, 2016; Hoadley, 2024; Snilstveit et al., 2016). Current research in South Africa, Kenya, and Haiti confirmed these indings (Angrist et al., 2023; Cilliers, Fleisch, Kotze et al., 2019; Fleisch, 2018; Guzmán et al., 2021; Piper et al., 2018). Within the structured pedagogic approach, studies pointed to the importance of CSPP as the optimal method for teacher in-service professional development (Kraft et al., 2018; Popova et al., 2022). Cilliers, Fleisch, Kotze et al. (2019) showed that when traditional training in a centralised venue is compared to training plus onsite instructional coaching, which is consistently more expensive, the latter is nonetheless more cost effective. This is evidenced not only by effects immediately after the end of the intervention but also by the persisting effects (Cilliers et al., 2022). Majerowicz and Montero (2018) found similar persistence in their study in Peru. These findings expand on Piper and Zuilkowski's (2015) coaching study that provided valuable evidence on the relative cost-effectiveness of different coach-teacher ratios.

However, it must be asked whether other forms or modes of structured pedagogy, particularly structured pedagogy programmes that use digital technology, could be effective. As part of a structured pedagogic model, could e-coaching be as effective as conventional coaching? There is a small but growing body of research that suggested this may be true. Although the study by Piper et al. (2016) on ICT interventions designed to improve early grade reading in Kenya provided an important baseline, it did not really engage with the idea that technology could be harnessed for virtual coaching. The literature on using digital technology for coaching has been around for the past decade, primarily in developed system contexts. Rock et al. (2014) were among the pioneers in e-coaching, or virtual coaching, particularly in relation to bug-in-ear technology. Their earlier study (Rock et al., 2014) focused on using virtual coaching with pre-service student teachers. Rock et al. (2013) tested Skype and how teachers could develop their classroom management skills through virtual coaching. Geissler et al. (2014) presented original empirical findings on virtual coaching programmes that combined telephone coaching with an internet-based coaching platform, and although the study was small in scale (14 participants who received three coaching sessions), they found positive results.

Stapleton et al. (2017) tested virtual coaching software to assess the potential of linking principal candidates with teacher candidates to ascertain whether the technology would allow principals to give teachers live feedback on their classroom practices. A new direction that this research is taking is to tackle the problem of scaling up. Hennessy et al. (2022) and Rodriguez-Segura (2022) in a research synthesis showed benefits of ICT for teachers, but there is limited evidence that the technology is sustainable, cost-effective or impactful on student outcome. Uribe-Banda et al. (2023) in their comparative study of teacher professional development found few outcome differences between technology-based versus blended equivalent, save that teachers preferred the blended version. Shal et al. (2025), exploring professional development not with direct training but via webinars, adds important insights into wider approaches to teacher education using technology. El-Serafy et al. (2022) have extended this literature into emergency settings where the use of technology for teacher professional development has a distinct advantage. Layton's (2023) work is important as it begins to explore possible mechanism involved in technology in effective e-coaching for teachers.

While this research provided an important point of departure, given the substantive difference in the context of teachers' work in relatively affluent systems compared to teachers' work in low-income contexts in the Global South, the evidence that emerged speaks to the realities of resource-constraint contexts. Bruns et al.'s (2018) study was one of the first of its kind that described the impact of a Skype coaching intervention on secondary school teachers in Brazil. Possibly the most important is the study by Jukes et al. (2017). They used a combined model with multiple components, including training workshops, semi-scripted lesson plans, and weekly text support for teachers, and conducted a cluster randomised trial involving 51 primary schools in Kenya.

The results of this experiment showed that the combined intervention model, which included text message support, improved Grade 2 student literacy outcomes. While Jukes et al. (2017) provided strong support for the viability of using an ICT type intervention to support coaching teachers in their classrooms, they did not answer the following key questions: Is an ICT/digital technology bundled intervention with virtual or e-coaching (digital technology structured pedagogical programme [DTSPP]) as effective as a CSPP? If so, what are the impact mechanisms of the digital technology component on early grade reading outcomes?

2.1 Conceptual Framework

TAM (Davis, 1989) framed our analysis of the mechanisms of DTSPP. In terms of the framework, acceptance refers to the extent to which new digital technology is perceived to be useful and easy to use. According to Marangunić and Granić (2015, p. 81), the TAM *"has evolved to become a key model in understanding predictors of human behaviour toward potential acceptance or* *rejection of the technology*". However, our use of the conceptual framework was less concerned with the specific operational aspects of the model and more concerned with how it differentiates between potential users' views of its usefulness on the one hand, ease of use on the other, and the relationship between these two perceptions and actual system use in practice. Figure 1 presents the core variables of the TAM, namely perceived usefulness, perceived ease of use, attitude, and behavioural intention, which could contribute to attitude, and by extension, to teachers' behavioural intentions to accept or reject technology. This study was premised on the TAM and the DeLone and McLean (2003) information systems success model. The interplay between the TAM and the DeLone and McLean information systems success model provided a lens to completely understand the complex dependent variables of technology-orientated operations in schools.



Figure 1: Modified TAM^a

^a Davis (1989)

Accordingly, the DeLone and McLean information systems success model allowed us to explore complex dependent variables in the positioning of digital technologies in the teaching profession in order to draw meaningful conclusions. DeLone and McLean defined "six distinct dimensions of information systems success: system quality, information quality, use, user satisfaction, individual impact, and organisational impact" (Urbach & Müller, 2012, p. 3). The six dimensions of information systems success are consistent with the TAM's construct, and their complementarity provided an integrated view on technology-orientated support of teachers to improve their professional practice. While much of the literature on teachers' acceptance of digital technology concentrated on the subjective element of the findings, the current study suggests the extent to which professional accountability may be a key factor in successful digital technology interventions for teachers and points to substantial differentiated uptake that is not linked to either perceived usefulness or perceived ease of use. Although systems quality contributes to behavioural intention and user satisfaction, the current study extended the theoretical framework with the additional dimension of professional accountability.

The increased affordances offered by digital technologies have become an attractive trend in the education sector to improve teachers' professional practices, but the quality of the support teachers receive is also a factor. Therefore, it is important for education leaders to pursue digital strategic plans to support teachers and enhance their professional practices. The TAM has been used to understand acceptance of new technologies by users in developed system contexts (Marangunić & Granić, 2015; Wingo et al., 2017), but the current study positioned the TAM and the DeLone and McLean information systems success model core variables in a developing and resource-constrained context. It was also important to consider the individual impact and organisational impact beyond the perceived usefulness and perceived ease of use constructs. The interplay between the TAM and the DeLone and McLean information systems success model provided us with the lens to explore the multidimensional relationship and interdependencies among the different constructs in developing and resource-constrained contexts.

3 RESEARCH STUDY

Over the past ten years, the Early Grade Reading Study has been advancing knowledge on system-wide improvement of early grade reading in both African languages and English as a second language in South Africa (Cilliers, Fleisch, Kotze et al., 2019; Cilliers, Fleisch, Prinsloo & Taylor, 2019; Fleisch, 2018). Much of this new knowledge centres on the effectiveness and the mechanisms of a basic structured pedagogic change model, which is a combination of detailed daily lesson plans, high-quality educational materials, and centralised training, otherwise referred to as the CSPP. The current research showed that the approach is a cost effective and sustainable way to improve early grade reading teaching system-wide. The South African research confirmed new insights from similar research programmes in India (Banerji & Chavan, 2016) and Kenya (Piper et al., 2018). These unique government and university research programmes are committed to building evidence-based knowledge through long-term cumulative research that integrates experimental and quasi-experimental research (RCTs and regression discontinuity designs), large-scale classroom observations, and in-depth qualitative case studies.

While there is a general agreement that structured pedagogy programme models are effective, concerns have been raised about the relatively high cost of the onsite instructional coaching component and the size of the pool of high-quality coaches available, particularly for remote rural areas in the Global South. To address these concerns, the original Early Grade Reading Study II investigated the viability and cost-effectiveness of an alternative model that involved a combination of components using digital technology. In this instance, the digital technology involved a mixture of low-cost Android tablets, a custom-built lesson plan application, a library of short video-clips, cell phone voice calls, text messaging, and virtual competitions. Preliminary findings from the first year of the intervention suggested that both conventional and digital technology versions were equally effective in improving early grade reading outcomes in English as a second language (Kotze et al., 2019). However, the learning outcome results at the end of the third year of the interventions revealed very different findings.

3.1 The Intervention Models

Both intervention models consisted of the following three core components:

- 1. detailed daily lesson plans;
- 2. integrated learning and teaching materials; and
- 3. professional development, which included both coaching and centralised training.

The first intervention model consisted of printed paper versions of the lesson plans, one-onone onsite coaching, and cluster needs-based workshops after school. In the digital technology model, the teachers received a low-cost 10-inch Android tablet with electronic versions of the lesson plans in the form of a simple application. The tablet also included various audio-visual resources. Coaching was provided either on the tablet or on the teachers' smart phones (or both) via cell phone calls and WhatsApp messaging, which included text, voice, and video messages. The electronic lesson plan application was custom-developed for the study. This and all other electronic resources were available offline to ensure functionality without Wi-Fi or mobile data.

The additional electronic resources included short videos demonstrating good classroom practice; sound clips of the phonics sounds and songs and rhymes that appear in the learning and teaching materials; and PDF examples of learners' work. Teachers in both interventions were trained in a residential setting at the start of each term. Those in the conventional intervention group were trained for two days, and those in the digital technology group received an additional day to orientate themselves to the tablet and Android applications. In addition, both groups of teachers attended one-day cluster training sessions in small groups. If any teachers did not attend a training session, the onsite coaches organised a makeup session to ensure that the teachers had the learning and teaching materials and understood the instructional practices. Teachers in the conventional version of the intervention received visits from specialist reading coaches about once a month for the duration of the school year.

During these visits, the coaches modelled, observed, supported, and evaluated teachers' classroom practices and monitored learners' exercise books. Coaching in digital interventions involved a phone conversation with each teacher once every two weeks, regular text messages, and the establishment of virtual communities of practice. In the conventional version of the intervention, the specialist reading coaches modelled professional practice to educators during their visits, but the virtual reading coaches used WhatsApp for weekly motivational messages and also sent teaching tips to answer questions on the lessons. While the virtual coach also ran biweekly competitions for teachers to showcase their work to the wider group, the absence of embodied simulation in the digital intervention could be the reason why the digital technology did not improve early grade reading. Embodied cognition theory shows that simulation (modelling) works best when it is face-to-face because of all the bodily-based moments of recognition that occur (Soylu et al., 2014). Table 1 is a comparison between the conventional version of the intervention and the digital intervention.
	Conventional Paper-Based	Digital Technology Electronic application on a tablet
Coaching	Coaches visit teachers in their classroom once every three weeks.	Coach contacts the teacher via telephone calls and WhatsApp once every two weeks.
Training	Initial training: Two-day block training.Quarterly training: One day at the start of each term.Needs-based training: Biweekly competitions.	<i>Initial training</i> : Three-day block training. <i>Quarterly training</i> : One day at the start of each term.
Core Methodologies	Paper-based instructional manual	Application-based instructions that include videos, sounds clips, and photos of example writing.

Table 1: Comparison of conventional and digital interventions

3.2 The Randomised Control Trial and Complementary Implementation Data

The original Early Grade Reading Study II (a cluster RCT with two intervention arms, in the Mpumalanga province, South Africa) was designed to test the viability and cost-effectiveness of two different intervention models. Alongside the core baseline endline data collection, the study team collected extensive information about teachers' use of the technology and their beliefs about and attitudes towards digital technology in the classroom. The core instrument to assess the change in the average levels of learning was a modified version of the Early Grade Reading Assessment (Dubeck & Gove, 2015). Learner data were collected at the beginning of Grade 1, end of Grade 2, and end of Grade 3 (each was regarded as a data collection wave). Implementation data were collected at various points through the three years of implementation, particularly during the teacher training points and during onsite and virtual coaching sessions. Data on teacher use of the tablets were collected annually. The original sampling frame included 180 primary schools in two of the four districts in the Mpumalanga province, South Africa. Schools were included if they were designated in the three poorest quintiles on the national index and used either of two local languages, isiZulu or siSwati, as the home language in Grades 1–3.

The 2 684 learners who were assessed with the oral instrument and the slightly small number who were assessed with the written instrument were the same learners assessed at the original baseline assessment at the beginning of Grade 1. These learners included repeaters, mostly in Grade 2, and a handful of learners that appeared to be in Grade 4. Each successive set of instruments was designed in collaboration with language experts with the intention that the instruments would provide core information about learners' literacy skills in the home

language and their language and literacy skills in English, as the additional language. The school groups were mostly balanced on the various subtests at baseline. There were only two imbalances out of the 10 subtests, and both of these were on items that had either strong ceiling or floor effects. In the 2019 testing, there was balance in terms of learner age, gender, and principal component analysis (PCA) sub-score standardised to control (Table 2).

Variables	(1)	(2)	(3)
	Best possible learner age based on	Learner Gender:	PCA: Sub-Scores
	response and date of birth	Boy=1	Standardised to Control
Onsite Coach	0.0330	0.0158	-0.000198
	(±0.0536)	(±0.0204)	(±0.070800)
Virtual Coach	0.0248	0.0133	0.1100
	(±0.0594)	(±0.0241)	(±0.0709)
Observations	2684	2684	2684
R-Squared	0.016	0.002	0.025
Control Mean	6.093	0.535	0.00525

Table 2: Balance on learner	age, gender,	and PCA sub-sco	ore standard
-----------------------------	--------------	-----------------	--------------

Of the 3 327 learners that were tested at baseline, 2 684 were retested at the end of Grade 3, yielding an attrition rate of about 19.3%. When we regressed the intervention assignment on the probability of attrition and controlling for the gender and language dummies, the results showed that there was no bias in attrition in any of the intervention groups. As part of the pilot study in 2015, the research team worked with a group of eight teachers in schools that were not part of the full trial to field test the tablet and the application, and the team also observed classrooms and did extensive in-depth interviews to ascertain potential weaknesses or flaws in the tablets, virtual coaching processes, and the lesson plan application. As such, the qualitative data generated in the pilot phase were incorporated into the analysis. Administrative data collected as part of the implementation of the large-scale trial provided extensive information about attendance at face-to-face training sessions and the condition of tablets at the end of the intervention. One unanticipated source of key evidence emerged from an analysis of the data downloaded from the applications at the end of Term 3, Year 3.

The extraction of data was originally designed to help the application designers better understand usage patterns. Data from 56 Grade 3 teachers' tablets were accessed, and this data were derived from records of the frequency, time, and types of files accessed on the tablets. Several variables were prepared after cleaning the data and anatomizing the records. These included average time spent on the videos, audio files, examples, and PDF files of explanations. This data provided invaluable information on the patterns and prevalence of tablet use. Special attention was paid to the pattern of use of the various lesson plan pages on the tablet application. An inverse cumulative density function opened for at least five seconds was calculated, which allowed individual slides to be counted every time they were accessed for more than five seconds. The data on tablet use provided insights into actual ease of use and the distribution of use between teachers.

4 FINDINGS

The main findings provide a picture of the impact of the interventions, relative to the control group, on learning outcomes after a full three years of implementation. In Table 3, we only provide the results for the overall impact of the interventions for learners who were enrolled in Grade 3 (for the results for the full sample, which does not differ significantly, see Cilliers et al. (2021). Table 3 shows that learners in the conventional intervention model group performed better on both the oral proficiency and the decoding components of the assessment. This was not the case for learners in the digital technology group. The finding is congruent with other studies (Dlamini, 2022; Dlamini & Rafiki, 2022; Mhlongo & Dlamini, 2022) on digital skills deficit among teachers to pedagogically integrate technology in their professional practice. Although these learners did a little better in the oral English tasks, they were substantially below the oral language proficiency levels of learners in the conventional intervention group and did no better than the learners in the control group in English reading aggregate statistics (PCA).

Variables	(1) Oral Prof.	(2) Reading	(3) Vocab	(4) L. Compr.	(5) Word Recog.	(6) Read Fluency	(7) Read Compr.	(8) W. Compr.
Conventional M	odel							
	0.356* (±0.078)	$0.179^{\dagger}\ (\pm 0.072)$	0.258* (±0.064)	0.351* (±0.084)	0.176* (±0.067)	0.122 [‡] (±0.071)	0.271* (±0.085)	0.104 (±0.075)
Digital Technolo	ogy Model							
	0.149 [‡] (±0.078)	-0.015 (±0.076)	0.086 (±0.068)	0.170 [‡] (±0.082)	-0.031 (±0.072)	0.016 (±0.075)	0.106 (±0.078)	-0.033 (±0.069)
Observations	2148	2109	2148	2148	2148	2148	2148	2109
R-squared	0.27	0.273	0.248	0.213	0.24	0.241	0.248	0.183
P-value	0.0223	0.0208	0.0185	0.0614	0.00786	0.187	0.0835	0.1
Control Mean	0.139	0.17	0.13	0.108	0.159	0.152	0.126	0.161

Table 3: Main results for English First Additional Language, Grade 3 only

p < 0.01

$^{\dagger}_{*} p < 0.05$

 $p^{*} < 0.1$

4.1 Why did the digital technology model fail to help students make gains?

Why would researchers be concerned with or pay close attention to the reasons for the failure of a digital technology-based innovation designed for early grade teachers and their students? Is it not sufficient that we know that it had limited impact? The evidence of the failure of a promising innovation is always likely to add to the wider body of knowledge, but if we take Deaton and Cartwright's (2018) insight about the 'why' of innovation, we discover that the mechanisms of both success and failure are of paramount importance in our efforts to better

understand how digital technology can contribute to improving learning outcomes. This was not in anyway failure of the digital technology, however digital skills gap among in-service teachers is a reality (Dlamini & Mbatha, 2018; Dlamini & Rafiki, 2022). The RCT was never designed explicitly to furnish rigorous evidence of possible reasons for innovation failure. As such, an excavation of the various layers of the intervention would at best yield hypotheses or working theories that future work could corroborate. The methodology of the excavation began with the most straightforward surface questions: Was the digital technology ever actually delivered to the teachers? Was the technology operational? Once we have established this, we can proceed with questions that go deeper into the implementation process.

4.2 First layer: Functionality

Did the digital intervention fail for simple reasons, such as the teachers did not get the tablets; they did not get training on how to use the tablets; they did not have Wi-Fi or power to charge the tablets; or they did not have data to communicate effectively with the virtual coach? We hazarded a guess that in many digital technology interventions in schools in the Global South one needs to look no further than the first layer to discover the causes of intervention failure.

According to the administrative data from the main service provider, the teachers all reported receiving functioning tablets, and they did not seem to have any problems with either access to data or electricity to recharge the devices. The teachers all received training on both the technology and the content of the material on the tablet. In other words, there was close to comprehensive delivery of the basic technical infrastructure and training in the project (Table 4).

		#Teachers	#Teachers Fully Trained	#Teachers Fully Resourced
Term 1	CSPP	86	83 (97%)	86 (100%)
	DTSPP	85	84 (98%)	85 (100%)
Term 2	CSPP	86	85 (99%)	86 (100%)
	DTSPP	83	83 (100%)	83 (100%)
Term 3	CSPP	86	85 (99%)	86 (100%)
	DTSPP	82	82 (100%)	82 (100%)
Term 4	CSPP	86	79 (92%)	86 (100%)
	DTSPP	82	80 (98%)	82 (100%)

Table 4: Implementation data: Teacher attendance at training and resources received for July–November 2019

In terms of the virtual coaching itself, the administrative data suggested that in addition to the regularly scheduled contacts, there were additional virtual coach-initiated coaching sessions. For example, in Term 3, 2019, teachers received, on average, two individual coaching

session each. Table 5 reveals that the Android tablet was remarkably robust and that teachers did not have serious problems with damage or theft. All teachers in all three cohorts had operational tablets at the end of the intervention year, and 88% of Grade 1 teachers had working tablets 24 months after their participation in the intervention had ended. Two possible interpretations could be offered for the durability of the tablets. Firstly, the teachers valued the devices, and thus, kept them in safe, good working order, or secondly, they seldom used them.

Issues	DTSPP (Tablet Provision)
No. of tablets distributed to Grade 1 teachers in 2017	90
No. of Grade 1 tablets operational at end 2017	90
No. of Grade 1 tablets operational at end 2019	79
No. of tablets distributed to Grade 2 teachers in 2018	80
No. of Grade 2 tablets operational at end 2018	80
No. of Grade 2 tablets operational at end 2019	77
No. of tablets distributed to Grade 3 teachers in 2019	82
No. of Grade 3 tablets operational at end 2019	82
Total no. of tablets distributed	252
Total no. of tablets operational at end 2019	238

Table 5: Tablet status	at the	end of the	intervention	period
------------------------	--------	------------	--------------	--------

4.3 Second Layer: Technology Acceptance Model

The second layer relates to subjective aspects. Even if the digital technology was delivered, teachers were trained in how to use it, and it remained in working order for the duration of the project, emotional responses to the technology can either encourage or inhibit it being used effectively. The basic framework to understand these subjective factors is the TAM. The model assumes that if individuals perceive the technology not to be useful in the workplace and as difficult to use, the technology is unlikely to be accepted and used. Conversely, if the technology is viewed as useful and easy to use, then take-up is likely to be high.

In the pilot study in preparation for the larger trial, the research team tested out the prototype tablet and related application with 11 schools that did not form part of the larger study. The focus of the pilot was on teachers' perceptions, their self-reported competencies with digital technology, the quality of the training, and detailed elements of the working of the application as it related to early grade reading methodologies. The key findings of the pilot were that most of the teachers owned smart phones but few had any experience with computers. All but one teacher reported that it was the first time they used a tablet. Despite the limited familiarity or experience with digital technology in general, five weeks into using the tablet and application almost all interviewed teachers reported agreeing or strongly agreeing that the tablet and the application were easy to use and that they were useful in their teaching. Hence, there was no question on the pedagogical affordances of tablets and the application in their professional practice (Mhlongo et al., 2023). While, the pedagogical affordances of digital technologies have been well researched and documented, the education sector in South Africa continues to struggle in the adoption and leveraging of digital education innovations (Dlamini & Ndzinisa, 2020; Ndibalema, 2022; Ndzinisa & Dlamini, 2022).

Overall, most teachers found the instructions on how to use the application clear and said that the training enabled ease of navigation. They found the font size to be adequate and the colour coding of activities clear and easy to follow. A few teachers reported being confused by the colour coding on the activities and word charts. Overall, the teachers in the pilot study reported a preference for the tablet and the lesson plan application to a conventional paper version. The reasons given where that the tablet is easier to take home, portable, and more convenient than paper versions (it fits into a handbag). They described the tablet as user-friendly and used phrases such as *"everything is in"*. As one teacher suggested, *"We are tired of paper!"* Another teacher preferred the digital technology, saying, *"The tablet, because the paper can be lost, torn easier"*. Another teacher said the following:

I don't carry a big file. At home, I open it and I know I have prepared for the next day. It is so easy. I don't have to read and have many books. It is easy to take home. Before, I was intimidated by the technology, but now it has opened my mind.

There were some less sanguine observations. One teacher observed, "I only use in preparation. We don't need it in the classroom, just in prep mode." That said, this teacher indicated that she did use the tablet "when I teach." Another teacher described reviewing the tablet application in preparation mode in the evening before the lesson. A number of teachers questioned the sound quality. As noted below, a number of teachers used the audio directly with the whole class, notwithstanding the fact that the tablet speaker was not intended for this purpose. One teacher said, "Mostly we used preparation mode. It is easy to understand and present to learners. The learners can listen to the songs; the learners understand the meaning of the song".

One teacher identified the songs and the flashcards as the two most valuable components of the digital programme, saying, "I use all audios, because they help to learn the tunes of the songs". On the resources in the preparation mode, some teachers indicated that they particularly liked specific features such as the videos. As one teacher put it, "It helped me set up my charts and with questions." This particular teacher indicated that she watched the video numerous times. Other teachers reported only watching the videos once, and one teacher never found the video button. One teacher found the questions of the day video most helpful: "I don't know how to question learners, but through the videos I am able to do it." One teacher singled out the video guide to improving classroom management, which demonstrated how to manage children while teaching. Unlike most teachers, one teacher insisted that "there are no videos." Another teacher admitted, "I haven't checked the videos as yet. I forgot about them", and a third said,

"Never used them." On the WhatsApp group, one teacher said, "I used the WhatsApp group, but the data finished at the end of October." Another teacher noted, "I don't receive any messages from the group." One teacher raised a concern that "the tablet consumes lots of airtime [data]."

The researchers observed teachers using the tablet and the application in the classroom, and none of them appeared to be using them for the first time. Most had the tablet on at the start of the lesson. While some used in-class mode, most used preparation mode. A few teachers started the tablet from an off position. There was considerable variation about where teachers placed the tablet. Some placed it on their desks and moved back and forth to consult it between activities. Others held the tablet in their hands and read from it consistently throughout the lesson. One teacher struggled to find the reading 'Lucky's Taxi', and the researcher helped the teacher locate the story. There was considerable variation between teachers in the types of activities where the application was used intensively. Some teachers used the application to help guide the questions of the day, and others used it for specific sections of the shared reading activity. Some used it only to read aloud. Although we did not observe this, it was evident from the interviews that a significant portion of the teachers used the audio clips with the learners, which explains why they suggested stronger speakers for the tablets. With a few exceptions, teachers using the tablet and application were either confident and competent or very confident and very competent. One teacher got stuck navigating between preparation mode and in-class mode. Another played a song and wished to play it again but was unable to navigate to it a second time. In an informal interview with a principal, it emerged that a teacher did not understand the time allocation on each of the pages and that the total number of minutes was not for the screen but for the entire activity, which may have multiple screens.

In Term 3, 2019 teachers accessed slides between July and September for lesson planning and preparation. There is evidence that at the beginning of the term the digital technology group accessed slides to guide their teaching, however towards the end of the term there is evidence that less teacher referred to the slides. This is an indication that the lesson plan is a roadmap of what should be learned and allow teachers to develop appropriate pedagogical activities. All the teachers in the digital technology group opened at least one of the lesson plan slides; about 65% reached at least 40% slide coverage; and 27% covered more than 60% of the term's slides. This data indicates that some teachers treat lesson plans as add-ons instead of a critical resource for their instructional planning.

The breakdown of slide coverage of Term 3, 2019, week-by-week provides evidence that slides access dependent on the term period. Earlier in the term about 64% of the slides were covered, while towards the end of the term, in week 10 about 19% of the slides were covered. Week 7 was particularly well covered, which was almost certainly because this was the week in which the official curriculum prescribed assessment task took place. Teachers were expected to upload assessment results onto the government-wide school management system into which teachers have to upload various learner assessment data. It is also interesting that aside from Week 7, there seems to be a pattern of better coverage earlier in the term (Weeks 1, 2 and 3) with a steady decline in coverage until Weeks 9 and 10, which had the lowest levels of coverage. This indicates that some topics were not comprehensively covered and is a concern to us that

teachers lack understanding of the importance of lesson plans. Yet, lesson plans needs to be checked continuously as a roadmap, resource and historical document for teachers to reflect on their teaching and content coverage.

Again, from the perspective of ease of use of the digitized lesson plans, teachers are encouraged to always think and reflect on the slides weekly instead of using the application when there is a bureaucratic requirement to post assessment results. At the outset, teachers are encouraged to think of instructional activities appropriate for their classroom and the slides are appropriate application for their planning and reflection.

The second year case study report noted that "virtual coaches and tablets are a successful mode of delivery for the EFAL [English First Additional Language] programme and teachers in this study were unequivocally positive about it." The second year case study concentrated on interviews and observations of eight teachers, so the claim was based on a very small sample of teachers in the study. Botha and Schollar (2018) identified the specific reason teachers were "unequivocally positive": The teachers who got the tablets and the application particularly liked the audio files that gave second language teachers access to standard English pronunciation of both sounds and words.

4.4 Third Layer: Differentiated Uptake

To answer the question about the failure of the digital technology intervention to improve learning outcomes, we explored the possibilities that the technology and related training were either not delivered at the outset; the technology and related training were of uneven quality; or that during the life of the project, the digital technology failed to function, got damaged, or was stolen. The available evidence suggests that none of these applied. The second layer of evidence related to the subjective or affective responses of teachers to the digital technology intervention. Following the TAM, we examined the available qualitative and quantitative evidence on the digital technology's ease of use and usefulness in daily work. Although a TAM survey was not conducted, an analysis of existing evidence suggested that, as a sample, the teachers in the intervention found the digital technology deployed in this experiment (the tablet and the application) relatively easy to learn to use and easy to use. They found the tablet and the virtual coaching useful in their work as teachers, particularly for some core functions.

Could it then be that differences between types of teachers could explain the lack of impact? Put another way, could it be that a small groups of teachers really benefitted from the digital technology component of the intervention programme (as suggested by the case study teachers in the Year 2 evaluation) and improved their practice, which had a strong positive impact on their learners' learning outcomes, but that most teachers, who could use the technology and did so but to a very limited extent, failed to get ignited by the technology, and therefore, it had little effect on their practice, leading to the absence of gains in learning outcomes. A differential teacher effect hypothesis raised three questions. Firstly, are there clearly discernible initial characteristics associated with teachers who responded to the technology in a particular way? Then, the more complex question: What are the mechanisms associated with the digital technology component? What works for some types of teachers but not for many others? Lastly, why would the conventional intervention model work for different types of teachers?

One of the common assumptions is that age could explain different levels of engagement with a digital technology intervention. It is often assumed that digital natives (adults who grew up with digital technology) are substantially more open to using digital technology in a reading teaching intervention than digital immigrants (see Prensky (2001) for definitions of the categories). The most obvious factor that differentiates digital natives from digital immigrants is age. However, Brown and Czerniewicz (2010) clearly showed that this idea has limited validity in the South African context. The evidence from the Early Grade Reading Study II does not support this hypothesis. As the statistics in Table 6 show, in the subsample of teachers (over 55) were more likely to spend time on the tablets than teachers in the middle age band. In terms of the total number of slides covered during the term, the numbers were very similar across the three age bands, even though the middle age band group of teachers covered less of the prescribed work in Term 3 on their tablets.

		_		
	#Teachers	Average Hours	Total Slides Covered	Average Coverage of Term 3
Younger than 45	9	17.6	567.5	50.1
45–54 years old	26	13.6	550.1	43.7
Older than 55	10	22.9	559.9	53.4

Table 6: Tablet usage by teacher age

A potentially useful avenue via which to explore differential teacher engagement was analysing the teachers' responses to the virtual coach's competition. To get a better sense of how teachers were implementing the core methodologies, the virtual coach introduced small competitions around specific themes. For instance, in Term 3, 2019, she focused on phonics and asked all her teachers to submit a photo via WhatsApp video of one of their phonics activities. She then chose the best teacher based on the image or video in each of the teacher groups. The teachers won airtime as a prize. The competitions allowed the virtual coach to see into the classroom and to gather evidence about what teachers considered their best practice in their classrooms to be. Table 7 shows the variability of teacher participation in the competitions. Just under a quarter of the teachers entered every competition. At the other end of the spectrum, 22% of teachers were completely inactive. Given that teachers were allowed discretion to enter or not, the pattern of participation may be a proxy indicator of motivation and commitment. The assumption is that those who did not enter at all were overall less likely to be engaging meaningfully with the digital technology and those who participated in every competition thrived in the new technology space.

#Competitions	#Teachers	% Teachers
0	18	21.69
1	18	21.69
2	16	19.28
3	12	14.46
4	19	22.89
Total	83	100%

Table 7: Participation in virtual coach competitions

Teachers in both the conventional and the digital technology interventions attended the early grade reading training at roughly the same rate: close to full participation. While all school heads of department and other members of the school leadership teams were invited to the training sessions, attendance at these was substantially lower for the digital technology schools, especially in the second half of the year (Table 8). One possible explanation is that the conventional approach (paper-based and onsite one-on-one coaching) involved regular visits to schools during which the coach interacted with both other foundation phase teachers and school managers, thus strengthening the school communities of practice. In contrast, the technology intervention tended to focus more directly on the relationship of the virtual coach with the individual teachers and did not trigger conversations between teachers and management team members at school level.

 Table 8: Implementation data:
 Senior Management Team (SMT)

 member attendance at training

		SMT members trained
Term 1	CSPP DTSPP	38 (76%) 31 (63%)
Term 2	CSPP DTSPP	85 (99%) 25 (51%)
Term 3	CSPP	36 (72%) 10 (30%)
Term 4	CSPP DTSPP	32 (64%) 14 (29%)

The following observation that the virtual coach made regarding the lower levels of enthusiasm on the part of some school leadership teams and links to teacher engagement confirmed this insight:

Yeah, it's really difficult. A lot of the time I have to like phone the SMT and say like, "Listen, I've been trying to get a hold of your teacher, please make sure that, you know, they've got

their tablet switched on and they check their WhatsApp's and things like that". A lot of the time you'll find that if a teacher is one of those teachers, they come from a school where the SMT isn't very active, so it's difficult.

In the final report from the implementing agents, the evidence that *"innovative programs that allow meaningful support to teachers at a large scale must continue"* (Cilliers et al., 2022, p. 29). In the second year school case studies, Botha and Schollar (2018) reported distinct advantages and disadvantages of the conventional model compared to the digital model. They found that three of the teachers in the digital technology case studies were not well prepared, were more dependent on the lesson plans on the tablets during teaching, and did not pay as close attention to pacing in the lessons. The 2019 case study researcher (Fleisch & Alsofrom, 2022) gave the following analysis:

The tech-based intervention inherently includes distance, the barrier to success may be selfmotivation. Teachers still feel supported, but there is not a strong enough accountability mechanism. ...it was successful primarily for teachers who were motivated enough to drive their own development process. Basically, it is teachers who are self-motivated (or are perhaps in an already functional school environment where accountability is provided through principal or colleagues interactions) where the technological intervention seems likely to be most impactful.

A pattern emerged when we compare and contrast the diverse evidence. There was substantial variability in the uptake of the digital technology intervention among teachers in the study sample, with roughly a quarter of them fully engaged and another quarter not engaged at all. Unlike an onsite coaching approach that involves visiting schools and talking to teachers face-to-face, the virtual coach struggled to hold teachers accountable and mobilise the wider school management team to support and monitor compliance with the intervention requirements. More broadly, the virtual coaching model was less successful with relationship building and creating professional accountability linked to the new instructional practices associated with improved reading outcomes.

Figure 2 presents the distribution of average school Oral Reading Fluency (ORF) from lowest to highest for the conventional (blue) compared to the digital technology (red) intervention schools. There is a high degree of overlap of the box-and-whisker plots, indicating that there is no significant difference between the two interventions. Although no causal inferences can be made from this graph, it is suggestive that both interventions appear to have a roughly equal proportion of successful schools, but that for average and below average schools, receiving the conventional intervention model benefitted these schools more than the digital technology intervention.

5 DISCUSSION

Given the limitations associated with intervention administrative data, purposively chosen case studies, and information from the tablet logs, the insights presented here should be viewed





Figure 2: Distribution of Average School ORF, English First Additional Language

as hypothesis building. Nested in a cluster randomised study of two similarly structured pedagogic programmes of which one used a conventional model of paper lesson plans and onsite coaching and the other used tablets and an application with virtual coaching, the study provided an opportunity to excavate the multiple layers required to understand how digital technology can become a tool to improve learning outcomes at scale. By a process of elimination, we were able to establish that it is unlikely that the digital technology hardware or software could explain the slow improvement. The evidence suggests that the devices were durable and that the custom-developed software had few glitches and was easy to master. Using the TAM framework and interpreting available evidence showed that there is reason to believe the slow progress should not be attributed to teachers' negative perceptions of its usefulness and ease of use.

We suggest the following alternative hypothesis: The teachers were not a homogeneous group but were diverse in their beliefs, attitudes, and take-up of the digital innovation. Smagorinsky et al. (2002) had this insight about various positions and responses imposed by top-down curriculum reforms. In this study, teachers adopted one of three diverse responses to reform/innovation: open resistance, accommodation, and acquiescence. For our purposes, we think an alternative set of positions or responses is more appropriate: technology avoiders, minimalists, and technology enthusiasts (Zur & Walker, 2011). Technology avoiders are teachers who may not actively resist the reform innovation but who do not engage with the technology and associated tasks required of them. Minimalists, on the other hand, make an effort to appear to be engaged but only do so to save face or avoid sanction. At the other end of the spectrum, technology enthusiasts engage fully with the innovations and find ways to adapt and internalise them. These three positions/responses must be seen as heuristic devices rather than sharp, well-defined, and empirically proven categories. Although these are suitable categories for the purposes of better understanding mechanisms of change, teachers' responses exist on a spectrum rather than as distinct, clearly bounded groups.

The second assumption we make is that these groups/categories are likely present in any reform intervention, whether conventional or digital. The critical difference, however, between conventional and digital interventions in this case was that the conventional onsite coaching intervention did not allow minimalist teachers to get away without showing change in their classroom practices. Teachers accustomed to minimalist approaches in the conventional reform needed to save face, avoid humiliation or embarrassment, and preserve their dignity by showing some of the reform work. This explains why the average school performance was more or less similar at the top end and different at the middle and lower ends with conventional intervention schools on average outperforming comparable digital technology intervention schools.

This observation adds credibility to the hypothesis that the slow progress of the digital technology intervention was: (a) on the interpersonal and emotional side of the change process, and (b) due to the difficulty that the virtual coaches had accessing actual teacher work in the classroom. In other words, face-to-face conventional interactions are more likely to improve trust and allow for professional accountability.

6 CONCLUSION

One of the key issues for discussion is not whether the digital technology intervention model works (i.e. that it provides consistent and statistically significant evidence of cost-effectiveness) but whether the intervention model helps reduce what the World Bank (Azevedo et al., 2019) called *"learning poverty"* fast enough to achieve the 2030 Sustainable Development Goal in education. Although eliminating learning poverty is generally associated with the ability to read by the end of primary school, the benchmark of reading for meaning in English at the end of Grade 3 is an appropriate benchmark in South Africa because English is the language of instruction from Grade 4 onward for the majority of learners and it is the de facto language of government and the economy. If escaping learning poverty is associated with the CSPP model works. However, more research is needed to understand how a digital technology version of the structured pedagogic programme could be more successful.

The starting point of this paper was the following question: Why did the ICT intervention fail to improve learning outcomes in South Africa's Early Grade Reading Study? At its most basic, the TAM suggests that acceptance of digital technology is a function of a combination of individuals' perceptions of the usefulness of the technology and of the relative ease of use. If individuals perceived that the technology is useful in their work and/or personal lives and that the technology is easy to learn and use, they are likely to accept and adopt the new technology. In accordance with Zhang et al. (2019, p. 208), the perceived usefulness, perceived ease of use, and attitude constructs *"are antecedents of technology acceptance."* However, the evidence in this study points to the following key findings:

- The centrality of professional accountability in successful digital technology interventions for teachers; and
- There is substantial differentiated uptake not linked to either perceived usefulness or perceived ease of use.

This is significant for developing and resource-constrained contexts. Both attitude and behavioural intention constructs are dependent variables in Figure 1, and in this case the differentiated uptake is dependent on professional accountability. According to UNESCO (2017, p. 1), accountability is "*a process, aimed at helping actors meet responsibilities and …goals.*" Although attitude influences users' behavioural intention to adopt and appropriate digital technologies in their professional practice, in this study there was no evidence associated with the differentiated uptake as a means to improve learning outcomes. The converse applied: If individuals come to think of the technology as not particularly helpful or useful and difficult to learn and use, once they have mastered how it works, they are unlikely to accept and ultimately adopt the new technology. Acceptance exists on a spectrum, and high usefulness and high ease of use are on the one end of the spectrum and low usefulness and low ease of use are on the other. Most individuals and groups fall at different points along the continuum.

We did not do a survey of teachers' perception of the digital technology intervention, which would have told us where they fit on the TAM spectrum. The evidence suggests three critical insights. Firstly, there was considerable variability in teachers' engagement with the digital technology intervention. Secondly, this variability was not driven by teachers' perception of the usefulness or ease of use of the various components of the digital technology intervention. Thirdly, the digital technology intervention did not allow the intervention into the classroom nor did it mobilise communities of practice at school level. The latter two insights suggest that trust and technology may be critical to understanding successful digital technology interventions in schools.

References

- Angrist, N., Aurino, E., Patrinos, H. A., Psacharopoulos, G., Vegas, E., Nordjo, R., & Wong, B. (2023). Improving learning in low- and lower-middle-income countries. *Journal of Benefit-Cost Analysis*, 14(S1), 55–80. https://doi.org/10.1017/BCA.2023.26
- Azevedo, J. P., Crawford, M., Nayar, R., Rogers, H., Rodriguez, M. R. B., Ding, E., Bernal, M. G., Dixon, A., Saavedra, J., & Arias, O. (2019). Ending learning poverty: What will it take? [Accessed: 2 April 2025]. https://documents.worldbank.org/en/publication/docume nts-reports/documentdetail/395151571251399043/ending-learning-poverty-whatwill-it-take
- Banerji, R., & Chavan, M. (2016). Improving literacy and math instruction at scale in India's primary schools: The case of Pratham's Read India program. *Journal of Educational Change*, 17(4), 453–475. https://doi.org/10.1007/S10833-016-9285-5
- Botha, D., & Schollar, E. (2018, December). Case studies in EGRS II schools 2018 [Accessed: 1 December 2022]. https://www.education.gov.za/Portals/0/Documents/Publications /EGRS/EGRS%20II%20Website%20Upload/Reports/4_EGRS%20II%20Case%20 Study%202018_Final_WEB.pdf

- Brown, C., & Czerniewicz, L. (2010). Debunking the 'digital native': Beyond digital apartheid, towards digital democracy. *Journal of Computer Assisted Learning*, *26*(5), 357–369. https://doi.org/10.1111/J.1365-2729.2010.00369.X
- Bruns, B., Costa, L., & Cunha, N. (2018). Through the looking glass: Can classroom observation and coaching improve teacher performance in Brazil? *Economics of Education Review*, 64, 214–250. https://doi.org/10.1016/J.ECONEDUREV.2018.03.003
- Cilliers, J., Fleisch, B., Kotze, J., Mohohlwane, M., & Taylor, S. (2019). *The sustainability of early grade education interventions: Do learning gains and improved teacher practices persist?* (Tech. rep.). RISE. https://riseprogramme.org/sites/default/files/inline-files/Cilliers v2.pdf
- Cilliers, J., Fleisch, B., Kotze, J., Mohohlwane, N., Taylor, S., & Thulare, T. (2022). Can virtual replace in-person coaching? experimental evidence on teacher professional development and student learning. *Journal of Development Economics*, *155*, 102815. https://doi.org/10.1016/J.JDEVEC0.2021.102815
- Cilliers, J., Fleisch, B., Prinsloo, C., & Taylor, S. (2019). How to improve teaching practice? an experimental comparison of centralized training and in-classroom coaching. *Journal of Human Resources*, *55*(3), 926–962. https://doi.org/10.3368/JHR.55.3.0618-9538R1
- Cilliers, J., Mbiti, I. M., & Zeitlin, A. (2021). Can public rankings improve school performance?: Evidence from a nationwide reform in Tanzania. *Journal of Human Resources*, *56*(3), 655–685. https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3390159
- Davis, F. D. (1989). Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS Quarterly: Management Information Systems*, *13*(3), 319–339. https://doi.org/10.2307/249008
- Deaton, A., & Cartwright, N. (2018). Understanding and misunderstanding randomized controlled trials. *Social Science & Medicine*, *210*, 2–21. https://doi.org/10.1016/J.SOCSC IMED.2017.12.005
- DeLone, W. H., & McLean, E. R. (2003). The DeLone and McLean model of information systems success: A ten-year update. *Journal of Management Information Systems*, *19*(4), 9–30. ht tps://doi.org/10.1080/07421222.2003.11045748
- Dlamini, R., & Ndzinisa, N. (2020). Universities trailing behind: Unquestioned epistemological foundations constraining the transition to online instructional delivery and learning. *South African Journal of Higher Education*, *34*(6), 52–64. https://doi.org/10.20853/34 -6-4073
- Dlamini, R. (2022). Digital equity in schools: A multilevel analysis of in-service teachers' technological knowledge competencies. *Journal of Educational Studies*, *21*(2), 40–60. https://hdl.handle.net/10520/ejc-jeds_v21_n2_a4
- Dlamini, R., & Mbatha, K. (2018). The discourse on ICT teacher professional development needs: The case of a South African teachers' union. *International Journal of Education and Development using ICT*, 14(2). https://eric.ed.gov/?id=EJ1190045

Dlamini, R. and Fleisch, B.: A multilevel analysis of digital technology to support teachers to improve ... 116

- Dlamini, R., & Rafiki, M. (2022). Teachers' perspectives on the integration of information and communication technology: The case of a teachers' union. *Africa Education Review*, *19*(1), 34–55. https://doi.org/10.1080/18146627.2023.2181728
- Dubeck, M. M., & Gove, A. (2015). The early grade reading assessment (EGRA): Its theoretical foundation, purpose, and limitations. *International Journal of Educational Development*, 40, 315–322. https://doi.org/10.1016/J.IJEDUDEV.2014.11.004
- El-Serafy, Y., Adam, T., & Hassler, B. (2022). The effectiveness of technology-supported teacher professional learning communities in emergency settings. In *Future-proofing teacher education* (pp. 145–157). Routledge. https://www.routledge.com/Future-Proofing-Teach er-Education-Voices-from-South-Africa-and-Beyond/Gravett-Petersen/p/book/978 1032028514
- Evans, D. K., & Popova, A. (2016). Cost-effectiveness analysis in development: Accounting for local costs and noisy impacts. World Development, 77, 262–276. https://doi.org/10.10 16/J.WORLDDEV.2015.08.020
- Fleisch, B. (2018). *The education triple cocktail: System-wide instructional reform in South Africa*. UCT Press/Juta; Company (Pty) Ltd.
- Fleisch, B., & Alsofrom, K. (2022). Coaching research in the early grade reading studies in South Africa. In N. Spaull & S. Taylor (Eds.), *Early grade reading and mathematics interventions in South Africa, Interventions* (pp. 48–63). Oxford University Press Southern Africa. https://en.calameo.com/oxford-university-press-south-africa/read/006710 753e7d90d4a4755
- Geissler, H., Hasenbein, M., Kanatouri, S., & Wegener, R. (2014). E-coaching: Conceptual and empirical findings of a virtual coaching programme. *International Journal of Evidence Based Coaching and Mentoring*, 12(2), 165–187. https://radar.brookes.ac.uk/radar/ite ms/585eb4f9-19ce-49e1-b600-509fde1e18c0/1/
- Guzmán, J. C., Schuenke-Lucien, K., D'Agostino, A. J., Berends, M., & Elliot, A. J. (2021). Improving reading instruction and students' reading skills in the early grades: Evidence from a randomized evaluation in Haiti. *Reading Research Quarterly*, 56(1), 173–193. https://doi.org/10.1002/RRQ.297
- Hennessy, S., D'Angelo, S., McIntyre, N., Koomar, S., Kreimeia, A., Cao, L., Brugha, M., & Zubairi, A. (2022). Technology use for teacher professional development in low- and middle-income countries: A systematic review. *Computers and Education Open*, *3*, 100080. https://doi.org/10.1016/J.CAE0.2022.100080
- Hoadley, U. (2024). How do structured pedagogy programmes affect reading instruction in African early grade classrooms? *International Journal of Educational Development*, *107*, 103023. https://doi.org/10.1016/J.IJEDUDEV.2024.103023
- Jukes, M. C., Turner, E. L., Dubeck, M. M., Halliday, K. E., Inyega, H. N., Wolf, S., Zuilkowski, S. S., & Brooker, S. J. (2017). Improving literacy instruction in Kenya through teacher professional development and text messages support: A cluster randomized trial. *Journal* of Research on Educational Effectiveness, 10(3), 449–481. https://doi.org/10.1080/19 345747.2016.1221487

- Kotze, J., Fleisch, B., & Taylor, S. (2019). Alternative forms of early grade instructional coaching: Emerging evidence from field experiments in South Africa. *International Journal of Educational Development*, 66, 203–213. https://doi.org/10.1016/J.IJEDUDEV.2018.09 .004
- Kraft, M. A., Blazar, D., & Hogan, D. (2018). The effect of teacher coaching on instruction and achievement: A meta-analysis of the causal evidence. *Review of Educational Research*, 88(4), 547–588. https://doi.org/10.3102/0034654318759268
- Layton, T. D. (2023). Can you see me now? The perceived impact of a virtual instructional coaching partnership applied through the lens of the partnership principles on first-year teacher professional growth: An explanatory sequential mixed-methods study [Doctoral dissertation, Baylor University]. https://eric.ed.gov/?id=ED633849
- Majerowicz, S., & Montero, R. (2018). Can teaching be taught? experimental evidence from a teacher coaching program in Peru (tech. rep.). Harvard. https://scholar.harvard.edu/fil es/smajerowicz/files/majerowicz_latest_jmp.pdf
- Marangunić, N., & Granić, A. (2015). Technology acceptance model: A literature review from 1986 to 2013. Universal Access in the Information Society, 14(1), 81–95. https://doi.org/10.1007/S10209-014-0348-1/TABLES/3
- Mhlongo, S., & Dlamini, R. (2022). Digital inequities and societal context: Digital transformation as a conduit to achieve social and epistemic justice: Digital transformation as a conduit to achieve social and epistemic justice. *IFIP Advances in Information and Communication Technology*, 645, 1–15. https://doi.org/10.1007/978-3-031-12825-7_1
- Mhlongo, S., Mbatha, K., Ramatsetse, B., & Dlamini, R. (2023). Challenges, opportunities, and prospects of adopting and using smart digital technologies in learning environments: An iterative review. *Heliyon*, *9*(6), e16348. https://doi.org/10.1016/J.HELIYON.2023 .E16348
- Ndibalema, P. (2022). Constraints of transition to online distance learning in higher education institutions during COVID-19 in developing countries: A systematic review. *E-Learning and Digital Media*, *19*(6), 595–618. https://doi.org/10.1177/20427530221107510
- Ndzinisa, N., & Dlamini, R. (2022). Responsiveness vs. accessibility: Pandemic-driven shift to remote teaching and online learning. *Higher Education Research & Development*, *41*(7), 2262–2277. https://doi.org/10.1080/07294360.2021.2019199
- Piper, B., Destefano, J., Kinyanjui, E. M., & Ong'ele, S. (2018). Scaling up successfully: Lessons from Kenya's Tusome national literacy program. *Journal of Educational Change*, 19(3), 293–321. https://doi.org/10.1007/S10833-018-9325-4
- Piper, B., & Zuilkowski, S. S. (2015). Teacher coaching in Kenya: Examining instructional support in public and nonformal schools. *Teaching and Teacher Education*, 47, 173–183. https://doi.org/10.1016/J.TATE.2015.01.001
- Piper, B., Zuilkowski, S. S., Kwayumba, D., & Strigel, C. (2016). Does technology improve reading outcomes? Comparing the effectiveness and cost-effectiveness of ICT interventions for early grade reading in Kenya. *International Journal of Educational Development*, 49, 204–214. https://doi.org/10.1016/J.IJEDUDEV.2016.03.006

- Popova, A., Evans, D. K., Breeding, M. E., & Arancibia, V. (2022). Teacher professional development around the world: The gap between evidence and practice. *The World Bank Research Observer*, *37*(1), 107–136. https://doi.org/10.1093/WBR0/LKAB006
- Prensky, M. (2001). Digital natives, digital immigrants Part 2: Do they really think differently? *On the Horizon*, *9*(6), 1–6. https://doi.org/10.1108/10748120110424843
- Rock, M. L., Schoenfeld, N., Zigmond, N., Gable, R. A., Gregg, M., Ploessl, D. M., & Salter, A. (2013). Can you Skype me now? Developing teachers' classroom management practices through virtual coaching. *Beyond Behavior*, 22(3), 15–23. https://doi.org/10.1177/1 07429561302200303
- Rock, M. L., Schumacker, R. E., Gregg, M., Howard, P. W., Gable, R. A., & Zigmond, N. (2014). How are they now? Longer term effects of e-coaching through online bug-in-ear technology. *Teacher Education and Special Education*, 37(2), 161–181. https://doi.org/10.1 177/0888406414525048
- Rodriguez-Segura, D. (2022). EdTech in developing countries: A review of the evidence. *The World Bank Research Observer*, *37*(2), 171–203. https://doi.org/10.1093/WBRO/LKA B011
- Scherer, R., Siddiq, F., & Tondeur, J. (2019). The technology acceptance model (TAM): A meta-analytic structural equation modeling approach to explaining teachers' adoption of digital technology in education. *Computers & Education*, 128, 13–35. https://doi.org /10.1016/J.COMPEDU.2018.09.009
- Shal, T., Ghamrawi, N., & Ghamrawi, N. A. (2025). Webinars for teacher professional development: Perceptions of members of a virtual professional community of practice. Open Learning: The Journal of Open, Distance and e-Learning. https://doi.org/10.1080/0268 0513.2023.2296645
- Smagorinsky, P., Lakly, A., & Johnson, T. S. (2002). Acquiescence, accommodation, and resistance in learning to teach within a prescribed curriculum. *English Education*, 34(3), 187–213. https://doi.org/10.58680/EE20021609
- Snilstveit, B., Stevenson, J., Menon, R., Phillips, D., Gallagher, E., Geleen, M., Jobse, H., Schmidt, T., & Jimenez, E. (2016, September). *The impact of education programmes on learning and school participation in low- and middle-income countries* (tech. rep.). International Initiative for Impact Evaluation (3ie). https://doi.org/10.23846/SRS007
- Soylu, F., Brady, C., Holbert, N., & Wilensky, U. (2014). The thinking hand: Embodiment of tool use, social cognition and metaphorical thinking and implications for learning design. *The Annual Meeting of the American Educational Research, Philadelphia, PA*. https://www.aera.net/Publications/Online-Paper-Repository/AERA-Online-Paper-Repository
- Stapleton, J., Tschida, C., & Cuthrell, K. (2017). Partnering principal and teacher candidates: Exploring a virtual coaching model in teacher education. *Journal of Technology and Teacher Education*, 25(4), 495–519. https://eric.ed.gov/?id=EJ1166362

- UNESCO. (2017). Accountability in education: Meeting our commitments; global education monitoring report, 2017/8 [Accessed: 3 April 2025]. https://www.unesco.org/gem-r eport/en/accountability
- Urbach, N., & Müller, B. (2012). The updated DeLone and McLean model of information systems success. *Integrated Series in Information Systems*, *1*, 1–18. https://doi.org/10.100 7/978-1-4419-6108-2_1
- Uribe-Banda, C., Wood, E., Gottardo, A., Biddle, J., Ghaa, C., Iminza, R., Wade, A., & Korir, E. (2023). Assessing blended and online-only delivery formats for teacher professional development in Kenya. *Cogent Education*, 10(1). https://doi.org/10.1080/2331186 X.2023.2191414
- Wingo, N. P., Ivankova, N. V., & Moss, J. A. (2017). Faculty perceptions about teaching online: Exploring the literature using the technology acceptance model as an organizing framework. Online Learning, 21(1), 15–35. https://doi.org/10.24059/olj.v21i1.761
- Zhang, T., Tao, D., Qu, X., Zhang, X., Lin, R., & Zhang, W. (2019). The roles of initial trust and perceived risk in public's acceptance of automated vehicles. *Transportation Research Part C: Emerging Technologies*, 98, 207–220. https://doi.org/10.1016/J.TRC.2018.11.018
- Zur, O., & Walker, A. (2011). On digital immigrants and digital natives: How the digital divide affects families, educational institutions, and the workplace [Accessed: 3 April 2025]. https://drzur.com/digital-divide/

Generative AI: Concerns, usage, challenges, opportunities and sentiments

Hossana Twinomurinzi[®] (b), Sibukele Gumbo^b (b)

^a Centre for Applied Data Science, University of Johannesburg, South Africa

^b Walter Sisulu University, East London, South Africa

ABSTRACT

Despite widespread adoption and enthusiasm, Generative AI faces persistent technical challenges, prompting the need to understand its broader impact. This study employed bibliometric analysis to examine scholarly discourse and key drivers of Generative AI adoption. The findings reveal the key factor driving adoption as transdisciplinary collaboration regardless of region; the concerns relate to inconsistent accuracy, demanding infrastructure, and ethical and privacy issues. Generative AI's rapid proliferation, spurred by ChatGPT, marked a post-digital era marked by transformative innovations across multiple fields. The scholarly discourse has grown by 3,164%, reflecting its significant influence on research and disciplinary progress. Education, medical, and biomedical sectors lead in adoption, to enhance teaching experiences, diagnostics, personalised medicine, and pandemic responses. However, the digital divide persists, leaving certain demographics and regions unable to access its benefits equally. The absence of robust ethical, legal, and regulatory frameworks further exacerbates these challenges. This study reveals Generative AI's transformative potential and its ability to bridge disciplines despite technical hurdles. The study makes recommendations for education reform to embrace new curricula, legal, and regulatory considerations; and to be intentional about bridging the widening digital divide.

Keywords artificial intelligence, generative AI, ChatGPT, bibliometric, ethics, scoping review, sentiment analysis, transdisciplinarity, post digital theory

 $\textbf{Categories} \bullet \textbf{Applied computing} \sim \textbf{Computers in other domains} \bullet \textit{Computing methodologies} \sim \textit{Artificial Intelligence}$

Email	Article hi	story
Hossana Twinomurinzi – hossanat@uj.ac.za (Corresponding)	Received:	27 June 2024
Sibukele Gumbo – sgumbo@wsu.ac.za	Accepted:	21 January 2025
	Online:	2 July 2025

1 INTRODUCTION

Machine Learning (ML), Artificial Intelligence (AI), and Natural Language Processing (NLP) have advanced rapidly in recent years, enabling the creation of powerful platforms like Chat-GPT and other Generative AI (GenAI) models, which are collectively referred to as Large Language Models (LLMs). While AI embraces a broad spectrum of technologies aimed at simulating human intelligence, GenAI specifically focuses on generating new content such as text,

Twinomurinzi, H., and Gumbo, S. (2025). Generative AI: Concerns, usage, challenges, opportunities and sentiments. *South African Computer Journal* 37(1), 120–145. https://doi.org/10.18489/sacj.v37i1.17890

Copyright © the author(s); published under a Creative Commons NonCommercial 4.0 License (SACJ is a publication of SAICSIT.Commons NonCommercial 4.0 License (SSN 1015-7999 (print) ISSN 2313-7835 (online))

images, or code by learning patterns from existing data (Hagos et al., 2024). GenAI technologies were initially discovered and developed in the mid-2010s but their widespread recognition surged from November 2022, following the launch of ChatGPT by OpenAI. ChatGPT reached over a million users within just five days of its launch, marking a pivotal moment that demonstrated the practical applications of GenAI and accelerated its widespread acceptance. This paper builds on preliminary findings presented at the South African Institute of Computer Scientists and Information Technologists (SAICSIT) 2023 Conference, extending the analysis to provide a more comprehensive exploration of GenAI's transformative potential, challenges, and societal implications (Twinomurinzi & Gumbo, 2023).

The core components of ChatGPT, and by extension other GenAI models, are constructed on three foundational building blocks; the transformer architecture, pre-training process, and fine-tuning process (Iskender, 2023). The transformer architecture utilises a neural network that is able to analyse large datasets at scale to allow for parallel processing of input data. Previously, the transformer model relied on self-attention mechanisms, which enhanced its ability to process sequential data efficiently. This foundational concept was later significantly advanced and expanded upon with the development of the Generative Pre-trained Transformer (GPT) series which used generative pre-training to learn language patterns from extensive text data. GPT-2 advanced the technology with multi-task training, improving the model's ability to generalise across various tasks. GPT-3 increased the model size, improving its fluency and comprehension capabilities (Iskender, 2023). With InstructGPT, the focus shifted towards integrating human feedback into the training process, resulting in more accurate and contextaware responses. The development of ChatGPT and GPT-4, meant GenAI was not only scaling up in size, but also integrating human feedback mechanisms (prompts) and the capacity to process image inputs (Iskender, 2023; OpenAI, 2025; Temsah et al., 2024; Zhong et al., 2024). Building upon these advancements, OpenAI also recently introduced the o1 model, which includes enhanced reasoning capabilities to reduce hallucinations and instances where AI generates incorrect or fabricated information (OpenAI, 2025). This improvement is particularly crucial in fields like healthcare, where accuracy is critical. For example, o1 has demonstrated superior performance in generating coherent and accurate radiology reports, outperforming other evaluated models (Temsah et al., 2024; Zhong et al., 2024).

GenAIs are versatile and, therefore, extend to a wide range of use cases, with about 75% of their value projected across the four key areas of customer operations, marketing and sales, software engineering, and R&D (McKinsey & Company, 2023). In customer operations, GenAI enhances customer experience and agent productivity. In marketing and sales, GenAI revolutionises content creation and personalisation. It facilitates the generation of personalised messages and marketing materials suited to individual customer preferences, behaviours, and interests. GenAI's capability extends to creating advertisements, social media posts, and product descriptions, substantially enhancing the productivity and creativity of marketing functions. In software engineering, GenAI is redefining the coding process by treating computer languages as natural languages. This transformation allows software developers to complete tasks 56% faster (McKinsey & Company, 2023). GenAI's influence in R&D is significant, particularly in

accelerating the development of new drugs and materials. For example, in the life sciences and chemical industries, GenAI foundation models are being used to generate candidate molecules, significantly speeding up the R&D process (McKinsey & Company, 2023). These advancements across different sectors illustrate the extensive impact of GenAI, not just in enhancing current practices but also in paving the way for novel approaches and efficiencies in various sectors. The technology's ability to understand and generate natural language plays a pivotal role in this transformation, marking a significant leap in automation and innovation potential. These examples highlight the versatility and potential of GenAIs to enhance productivity, streamline content creation, and improve the efficiency of text-related tasks in various domains. However, to ensure their responsible and ethical deployment in real-world applications, it is important to carefully consider the ethical implications and technological challenges associated with the use of GenAI. There are, therefore, divergent discourses surrounding the use of GenAI tools such as ChatGPT and similar GenAI. For example, Italy initially banned ChatGPT citing privacy and data infringement concern (McCallum, 2023).

The main objective of this study was to provide a holistic view of the usage, concerns, challenges, and optimisms of GenAI. A bibliometric analysis was done using the Scopus and Web of Science (WoS) databases, and the identified publications were analysed and visual-ised using Bibliometrix[®] and VOSViewer[®] tools (Aria & Cuccurullo, 2017; VOSviewer, 2025). Specifically, the study sought to answer the following research question:

How are prevalent patterns of GenAI usage represented in scholarly discourse, and what key factors are identified as driving the adoption of these technologies?

The study contributes to our understanding of GenAI integration and its societal impact. Firstly, it provides insights into the transdisciplinary dynamics shaping GenAI adoption across diverse sectors, highlighting the complex relationships between technological advancements, ethical considerations, regulatory frameworks, and societal values. Secondly, it offers an analysis of the ethical challenges inherent in GenAI development and deployment, shedding light on issues of bias, fairness, transparency, and accountability, and proposing strategies to navigate these ethical dilemmas responsibly. Thirdly, the study contributes to the discourse on regulatory governance and policy interventions necessary to ensure the responsible and equitable integration of GenAI, emphasising the importance of adaptive regulatory approaches that balance innovation incentives with societal values and individual rights. Lastly, the research underscores the need for inclusive and participatory approaches to GenAI development and deployment, emphasising the importance of engaging diverse stakeholders in transparent and informed discussions about the opportunities, risks, and ethical implications of GenAI.

This study distinguishes itself from other similar bibliometric reviews of GenAI by adopting a broad, cross-disciplinary lens to explore the impacts, going beyond domain-specific focuses like ChatGPT's role in healthcare (Lopes, 2024) or education (Liu et al., 2024). Unlike Khan et al. (2024), which emphasises general trends and geographic analyses, or R. Dwivedi and Elluri (2024), which limits its scope to technical fields and topic modelling, our study employs inductive thematic analysis to uncover richer insights across disciplines. Also, by separating pre- and post-ChatGPT trends and addressing GenAI's role in stimulating new fields and

educational reform, this study offers a unique and forward-looking perspective on GenAI's transformative potential.

The remainder of the study is structured as follows: The next section provides an overview of the methodology employed to conduct the study followed by the results and discussion. Finally, the contributions, limitations, conclusions of the study, and proposed research agenda are presented.

2 RESEARCH METHODOLOGY

Bibliometric analyses are increasingly popular as a means to identify trends in data, such as the influence of authors, the most cited articles, and top contributing countries (Aria & Cuccurullo, 2017). In this study, we employed the method to explore the scholarly discourse on GenAI, focusing on its usage, challenges, opportunities, concerns and sentiments. The process involved collecting, analysing and visualising data, and then reporting the findings and inferences.

For the data, we conducted a search of Scopus and WoS on 9 November 2023 using the search string: ('Generative AI' OR 'generative artificial intelligence' OR 'AI generation' OR 'GenAI' OR 'ChatGPT' OR 'large language models'). We used WoS and Scopus because they are compatible with bibliometric tools like Bibliometrix[®], ensure high-quality data, and currently represent the most practical options. Grey literature was not included to maintain a focus on peer-reviewed, high-quality, and standardised academic sources, which are critical for ensuring rigour and replicability.

We retrieved 3320 articles from Scopus and 1869 articles from WoS. Since we preferred peer-reviewed publications, we narrowed down the list to include only refereed journal articles and conference publications. We also included journal articles in press. Additionally, we limited the search to only those in the English language resulting in 1987 articles on Scopus and 1211 on WoS. After merging the two datasets, removing duplicate articles, and a peer review of each abstract and title, we found 1314 documents with missing abstracts or unrelated to GenAI leaving a total of 1884 documents. Biblioshiny, an online interface for Bibliometrix[®], was used to analyse and visually represent the bibliometric data. We also conducted the thematic analysis manually, and sentiment and usage analyses using OpenAI's interface guided by the results of the manual analysis. Figure 1 gives a visual representation of the method.

3 RESULTS AND DISCUSSION

3.1 Main Information

In the scoping review shown in Table 1, we manually delineated the document analysis into two distinct periods: 2012–2022, specifically 30 November 2022 when ChatGPT was released, and 01 December 2022–2024, to capture the evolving dynamics of GenAI, particularly highlighting the pivotal role of ChatGPT's emergence as a significant turning point (Twinomurinzi



Figure 1: Schema for identification of research articles and analysis of results

& Gumbo, 2023). The first period, 2012–2022, had 56 documents from 34 sources, with a steady growth rate of 37.97%, an average number of 4.39 co-authors per document, and 26.79% in international collaboration.

The second period, 2022–2024, revealed a sharp increase in research activity, with 1828 documents from 941 sources. The interest and research output grew exponentially by 3,164%, and with the number of authors surging to 6283. The average co-authors per document was 4.01, and a slightly lower rate of international collaboration at 22.98%. This period also saw

		Results	
Main information about data	Overall	Period 1	Period 2
Timespan	2012–2024	2012–2022	2022–2024
Sources (Journals, Books, etc.)	962	34	941
Documents	1884	56	1828
Annual Growth Rate %	25.99%	37.97%	3164.29%
Authors			
Authors	6423	184	6283
Authors collaboration			
Single-authored docs	336	5	320
Co-authors per document	4.02	4.39	4.01
International co-authorships %	23.09%	26.79%	<i>22.98%</i>
Document types			
article	1256	13	1243
article; early access	40	0	40
book	4	0	4
book chapter	27	2	25
conference paper	557	41	516

Table 1: Summary of the documents analysed

a notable increase in journal articles, including early access articles, and a substantial rise in conference papers to 516. ChatGPT was definitively an inflection or tipping point for GenAI and AI usage (Twinomurinzi & Gumbo, 2023).

3.2 Annual Scientific Production

The publication trends between 2012–2022, Figure 2(a), indicates a slow and steady increase in publications peaking at 25 publications until November 2022, suggesting a nominal interest in GenAI. In contrast, the period December 2022–2024, in Figure 2(b) reveals an explosive growth from 4 publications in December 2022 to 1808 in 2023. The sharp reduction in 2024 reflects the 16 early access publications in the forthcoming 2024.





Figure 2: Annual scientific production of GenAI

3.3 Top 10 Country Specific Production

Between 2012–2022, the scholarly discourse on GenAI was predominantly led by the United States of America (USA), which contributed 137 publications, nearly twice the total output of the nine other leading countries in GenAI research combined, 75 publications (Table 2(a)). China followed with 20, marking a substantial gap from the top article producer, USA. The presence of the Netherlands (9), Germany (8), Greece (7), Ireland (7), and Switzerland (5) reveals Europe's contribution in the first period. Japan (8), Canada (6), and Argentina (5) represented Asia, North America, and South America, respectively, depicting a nearly global diversity in scholarly production. Notably, African countries were absent. In the subsequent period (2022–2024), the USA maintained its lead with 1877 occurrences (Table 2(b)). China also saw a substantial increase to 639, reinforcing its position as a global research producer. In Europe, Germany (348), the United Kingdom (UK) (315), and Italy (299) were in the lead. Australia (286) emerged in the top 10, suggesting its growing role. Asia's expanded role had India (252), South Korea (143), and Singapore (126). Canada (138), showed a slower growth rate compared to other leading countries. Both periods had no representation from Africa in the top 10.

3.4 Most relevant sources

Between 2012–2022 (Table 3(a)), the sources were centered around Computer Science, particularly in specialist AI conference series. Between 2022–2024 (Table 3(b)), there was a significant shift in the scholarly discourse on GenAI, with the medical and biomedical fields emerging as the new leaders. This transition marks an increased integration of GenAI technologies in healthcare, particularly in diagnostics, personalised medicine, and pandemic response, reflecting the sector's growing reliance on GenAI.

#	Country	Continent	count	%
1	USA	North America	137	64.6%
2	China	Asia	20	9.4%
3	Netherlands	Europe	9	4.2%
4	Germany	Europe	8	3.8%
5	Japan	Asia	8	3.8%
6	Greece	Europe	7	3.3%
7	Ireland	Europe	7	3.3%
8	Canada	North America	6	2.8%
9	Argentina	South America	5	2.4%
10	Switzerland	Europe	5	2.4%
Tota	al		212	

Table 2: Country and continent specific production

(a) 2012–2022

(b) 2022	-2024
----------	-------

#	Country	Continent count		%
1	USA	North America	1877	42.4%
2	China	Asia	639	14.4%
3	Germany	Europe	348	7.9%
4	UK	Europe	315	7.1%
5	Italy	Europe	299	6.8%
6	Australia	Australia	286	6.5%
7	India	Asia	252	5.7%
8	South Korea	Asia	143	3.2%
9	Canada	North America	138	3.1%
10	Singapore	Asia	126	2.8%
Tota	al		4423	

3.5 Most relevant authors

From 2022–2024 (Table 4(b)), there was a notable shift in the leading scholars on GenAI. While all top 10 contributors in the period 2012–2022 (Table 4(b)) came from the USA, 2022–2024 saw a significant pivot towards Asia. China particularly emerged as the leading scholarly voice on GenAI. This shift indicates the rising influence of Asian researchers in GenAI.

Table 3: Most relevant sources

(a) 2012-2022

Source	count
CEUR Workshop Proceedings	8
Conference on Human Factors in Computing Systems – Proceedings	5
Lecture Notes in Computer Science ^a	5
ACM International Conference Proceeding Series	4
International Conference on Intelligent User Interfaces, Proceedings IUI	3
Proceedings of the 36th AAAI Conference on Artificial Intelligence, AAAI 2022	2
Proceedings of the National Academy of Sciences of the United States of America	2
2012 IEEE Congress on Evolutionary Computation, CEC 2012	1
2022 IEEE Congress on Evolutionary Computation, CEC 2022 – Conference Proceedings	1
35 th AAAI Conference on Artificial Intelligence, AAAI 2021	1

^{*a*} including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics

(b) 2022-2024

Source	count
Cureus Journal of Medical Science	56
CEUR Workshop Proceedings	55
Lecture Notes in Computer Science ^a	50
Proceedings of the Annual Meeting of the Association for Computational Linguistics	41
ACM International Conference Proceeding Series	32
Annals of Biomedical Engineering	19
Conference on Human Factors in Computing Systems – Proceedings	16
JMIR Medical Education	16
IEEE Access	13
Journal of Medical Internet Research	13

^a including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics

3.6 Top Affiliations

Between 2012–2024, the affiliations of leading contributors also underwent a notable shift. Initially (Table 5(a)), the USA dominated, reflecting its strong research infrastructure. However, from 2022–2024 (Table 5(b)), a significant diversification occurred, with a broader international representation emerging. This period saw the rise of institutions from the United Arab Emirates (UAE), Singapore, Australia, Poland, and the UK, among others.

3.7 Country collaboration

There was hardly any international collaboration on GenAI during the first 11-year period (Figure 3(a)), with only one discernible partnership between the USA and Argentina. This con-

(a) 2012–2022			(b) 2022–2024		
Author	Country	count	Author	Country	count
Houde S	USA	7	Chen J	China	13
Muller M	USA	7	Kim J	Korea	11
Agarwal M	USA	6	Li H	USA	11
Talamadupula K	USA	6	Li Y	China	11
Weisz Jd	USA	6	Liu H	China	11
Breazeal C	USA	5	Liu J	China	10
Martinez F	USA	5	Seth I	Australia	10
Richards J	USA	5	Wang X	China	9
Ali S	USA	4	Wang Y	China	8
Ross Si	USA	4	Zhang Y	USA	8
Total		55	Total		102

Table 4: Most relevant authors

trasts sharply with the surge in global collaboration from 2022 onward (Figure 3(b)), spanning nearly every continent.



(a) 2012-2022

(b) 2022–2024

129

Figure 3: Country collaboration

Within the past year, GenAI has markedly enhanced global collaboration, establishing itself as a pivotal platform and catalyst across various disciplines. Its broad appeal, as seen in its application across diverse research fields in Section 3.10, shows GenAI's role in fostering transdisciplinary integration. The integration has enabled scholars from diverse fields to pool their expertise and perspectives, thereby enriching the collaborative efforts.

GenAI platforms such as ChatGPT, Google's Bard, and Microsoft's Copilot have been designed with user-friendliness in mind, reducing the need for specialist technical expertise that was once necessary for engaging with advanced AI tools as seen in the period 2012–2022. This approachability has significantly contributed to their widespread adoption, allowing a broader range of users from various backgrounds to utilise these platforms effectively. By lowering the

Table 5	5: Top	o affilia	ations
---------	--------	-----------	--------

Affiliation	Country	count
Facebook AI Research	USA	6
Hefei University of Technology	China	6
Massachusetts Institute of Technology	USA	6
IBM Research AI	USA	5
Swiss Federal Institute of Technology (ETH Zurich)	Switzerland	5
University of Tsukuba	Japan	5
Universiti Malaysia Sabah	Malaysia	4
University of Illinois	USA	4
Kansas State University	USA	3
Shanghai Jiao Tong University	China	3
Total		47

(a) 2012–2022

(b) 2022–2024

Affiliation	Country	count
New York University, Abu Dhabi	UAE	41
University of Southern California	USA	36
National University of Singapore	Singapore	33
Icahn School of Medicine at Mount Sinai	USA	32
Monash University	Australia	30
University of California	USA	29
University of North Texas	USA	29
Wrocław University of Science and Technology	Poland	24
Mayo Clinic	USA	23
University of Cambridge	UK	23
Total		300

barrier to entry, these GenAI platforms have democratised access to advanced AI capabilities, facilitating a more inclusive environment for users worldwide including those from marginalised communities who often face barriers due to the digital divide. GenAI therefore offers opportunities for greater participation and representation in AI-driven advancements.

Their accessibility also ensures that individuals across different technical proficiencies can participate, further promoting an inclusive environment conducive to collaboration. This ease of access is a key factor driving GenAI's popularity as a collaborative tool.

Moreover, the rise of GenAI mirrors a broader shift in the research and academic culture towards openness and collective problem-solving. This paradigm shift acknowledges the complexity of contemporary challenges, which often demand efforts from more than one discipline, a process seamlessly facilitated by GenAI. The network effects triggered by GenAI's rapidly expanding user base further increase its value. As more scholars, individuals and institutions

engage with GenAI platforms, the resultant growth not only enhances GenAI's usefulness but also fosters a virtuous cycle of increased scholarly adoption and collaboration.

3.8 Top 10 cited documents

The top ten papers from both periods (Tables 6 and 7) illustrate a shift from the ethical concerns and technical proficiency in the use of GenAI to a large scale usage of GenAI particularly in the medical and education fields.

Source	Publication	тс	Summary
Rives et al. (2021)	Proceedings of the National Academy of Sciences of the United States of America	504	This paper investigated the use of unsupervised learning in AI to train a deep contextual LLM, leading to breakthroughs in biological data in- terpretation. The study demonstrated how the model's representations hold key biological insights from sequence data, offering a predictive edge in mutational effects and protein structure.
Gupta et al. (2017)	Molecular Informatics	285	This research presented a novel chemogenomic approach using gen- erative recurrent neural networks (RNNs) for drug design, bypassing the need for extensive compound libraries. It highlighted the method's precision and adaptability, particularly in low-data scenarios for drug discovery.
Mirsky and Lee (2021)	ACM Computing Surveys	138	This article explored the ethical concerns of deepfake technology. It detailed the creation and detection of deepfakes, the progression of the technology, and the need for further research to bolster defence mechanisms against its misuse.
Williams et al. (2019)	Conference on Human Factors in Computing Systems – Proceed- ings	79	This study developed Popbots, an AI educational platform for preschoolers. It found that early AI education could significantly shape children's understanding of AI, with varied perceptions based on their age and learning performance.
Weisz et al. (2021)	International Confer- ence on Intelligent User Interfaces, Pro- ceedings IUI	33	This paper examined the application of unsupervised neural machine translation for code generation. It assessed software engineers' toler- ance for imperfections in AI-generated code and discussed the potential for generative AI in software modernisation.
Sun et al. (2022)	International Confer- ence on Intelligent User Interfaces, Pro- ceedings IUI	24	This research interrogated explainable AI for generative models in soft- ware engineering. It proposed new explainable AI features and high- lighted the importance of human-centered design in the technical de- velopment of explainable AI.
Zhang et al. (2022)	Conference on Human Factors in Computing Systems – Proceed- ings	24	This study designed StoryBuddy, an AI system for interactive storytelling. The system aims to balance the need for parent involve- ment with the goal of enhancing parent-child bonding while also ac- commodating parents' busy schedules.
Suh et al. (2021)	Conference on Human Factors in Computing Systems – Proceed- ings	23	This paper observed the role of AI in human collaboration during creat- ive tasks. The key findings suggest AI can significantly influence social dynamics and creativity, offering insights for the future integration of AI in co-creative processes.
Wu et al. (2020)	Computers in Human Behaviour	20	This survey investigated perceptions of AI-generated content across different cultures. It found contrasting views between American and Chinese subjects regarding AI's role in artistic creation, enriching the dialogue on the impact of AI in the arts.

Table 6: Top 10 cited documents 2012–2022

[Continued ...]

https://doi.org/10.18489/sacj.v37i1.17890

Source	Publication	тс	Summary
Ali et al. (2021)	Conference on Human Factors in Computing Systems – Proceed- ings	17	This study proposed a learning trajectory for middle schoolers about GenAI. It emphasised the importance of educating the young on ethical aspects of AI, based on findings from interactive workshops.

Table 7: Top 10 cited documents 2022–2024

Source	Publication	тс	Summary
Y. K. Dwivedi et al. (2023)	International Journal of Information Man- agement	226	This study scrutinised the integration of GenAI in streamlining in- formation management processes, illustrating its pivotal role in enhancing data interpretation and decision-making within organ- isations.
Gilson et al. (2023)	JMIR Medical Educa- tion	205	The research investigated the deployment of GenAI in medical edu- cation, revealing innovative teaching tools that leverage AI to cre- ate dynamic learning materials and simulations, significantly im- pacting educational outcomes.
Rudolph et al. (2023)	Journal of Applied Learning & Teaching	123	The study highlighted the use of GenAI in crafting customised learning experiences, focusing on the practical application of AI- generated content to enhance teaching methodologies.
Tlili et al. (2023)	Smart Learning Envir- onments	112	This article presented the application of generative AI in smart learning environments, demonstrating how it can tailor educa- tional content to individual learner profiles, thereby revolution- ising traditional educational paradigms.
Ayers et al. (2023)	JAMA Internal Medi- cine	109	The study explored GenAI's role in internal medicine, particularly in developing predictive models for patient care, which has led to more precise and personalised treatment plans.
Pavlik (2023)	Journal of Mass Communication Education	108	This paper examined the impact of GenAI on mass communication education, emphasising how AI tools can generate diverse media content, thereby enriching the learning experience for students.
Cotton et al. (2023)	Innovations in Edu- cation and Teaching International	103	The research focused on the GenAI's contribution to educational innovation, particularly its capability to produce diverse and in- clusive educational materials that resonate with a global student body.
Cascella et al. (2023)	Journal of Medical Systems	101	This study investigated the incorporation of GenAI in medical sys- tems, noting its potential to revolutionise patient diagnostics and treatment pathways through advanced AI-generated models.
Salvagno et al. (2023)	Critical Care	98	The paper highlighted how GenAI is being utilised to create sim- ulation models for critical care training, offering realistic scen- arios that are instrumental in preparing healthcare professionals for emergency situations.
Alkaissi and McFarlane (2023)	Cureus Journal of Medical Science	87	This study outlined the use of GenAI in medical science education, underlining its role in generating complex biological models that aid in both teaching and research.

https://doi.org/10.18489/sacj.v37i1.17890

3.9 GenAI Usage and Opportunities

A thematic analysis using an open coding approach was conducted for the usage and opportunities, as well as the concerns, between the two periods (Braun & Clarke, 2006). Initially, all the abstracts in the period 2012–2024 were manually analysed to derive codes, and from the codes, overarching themes, and then finally constructing a narrative to describe the themes that emerged. Atlas.ti was used for this process. ChatGPT was then used to compare its thematic analysis with the manual analysis which proved sufficiently similar and revealed a few more codes. ChatGPT GPT-4 version was particularly useful for the second period 2022–2024 with 1828 abstracts.

Before the emergence of ChatGPT (Table 8), GenAI was prominently engaged in the arts and design sector due to its creative capabilities. In education, GenAI was leveraged to enrich learning experiences, frequently through robotic intermediaries. GenAI had already begun integrating into a few other disciplines, illustrating its potential even before ChatGPT.

Discipline	Code (number)	Brief integrative narrative
Computer Science (30)	Algorithm (2) Computing (5) Software (10) System (12) Simulation (1)	GenAI was used in system development, software engineering, and algorithm design. It also played a role in simulation and computing, reflecting its foundational impact on technological advancements. An example includes the novel use of a modified Pareto Differential Evolution (PDE) algorithm for bi-objective optimisation of the weights in an Artificial Neural Network (ANN) controller. This marked a significant development in the field, which previ- ously focused only on single-objective optimisation.
Art and Design (42)	Design (19) Art (9) Creative (10) Aesthetic (1) Visual (3)	GenAI played a pivotal role in the art and design sector, particularly in creative projects involving real-time strategy games. This indicated GenAI's emerging role in enhancing creativity and aesthetic judgment. Examples included creat- ing simulations and collaboration in the creation of classical Japanese poetry.
Environmental Science (2)	Environment (1) Sustainable (1)	While less prominent, GenAI used top-down strategy mechanisms like steering development processes and addressing misinformation. For instance, the use of interactive learning environments and digital literacy programs to combat deepfakes and social media misinformation demonstrated how to refine ideas and ensure alignment with desired outcomes.
Healthcare (4)	Drug (4)	GenAI models provided innovative approaches to chemogenomics and de novo drug design. They enabled researchers to focus on specific regions of the chem- ical space, as demonstrated by exploring the chemical space and generating new molecules. This assisted in the discovery of new drugs. GenAI was also used to maximise the use of limited data in making new drug discoveries.
Education (32)	Learning (25) Education (6) Teaching (1)	GenAI also played a role in education, especially in enhancing learning exper- iences. One notable example involved fine-tuning Recurrent Neural Network (RNN) predictions for specific molecular targets using transfer learning. Addi- tionally, the introduction of the NAO humanoid robot in elementary schools offered a unique perspective on learning. Here, students engaged in collaborat- ive discussions, questioning, and reflective writing activities centered around the robot.

Table 8: Usage of GenAI prior to ChatGPT (2012–2022)

[Continued ...]

Discipline	Code (number)	Brief integrative narrative
Robotics (2)	Robot (1) Humanoid (1)	In robotics, GenAI contributed to the development of sociable robots, which are increasingly being used as interfaces for various services. Notably, chil- dren born after 2010, the "AI Generation" became familiar with social robotic interfaces, influencing their mental development. Some children percieved the social robots as more intelligent and/or less intelligent than humans, and

Table 8: [...continued]

Post-ChatGPT (Table 9), there has been a substantial surge in the usage and integration of GenAI, marking a new era of technological adoption. The scholarly discourse has evolved past the previous ethical concerns surrounding GenAI, embracing its integration across various disciplines. Education has particularly been at the forefront of using GenAI, significantly enhancing the learning experiences. Similarly, healthcare has expanded its use of GenAI, enriching medical science curricula and improving patient outcomes. GenAI has also facilitated the development of autonomous robots equipped with communicative capabilities.

others simply viewed them as toys.

Discipline	Code (number)	Brief integrative narrative
Education (1208)	Learning (543) Education (495) Teaching (170)	GenAI has revolutionised education, significantly enhancing learning and teaching methodologies. Notably, the integration of LLMs into educational systems. These LLMs are being used to develop personalised study plans, innovative teaching methods, and more interactive learning materials. For instance, ensemble neural models have been proposed to generate probabilities from different pre-trained LLMs thereby offering more tailored and effective educational experiences.
Computer Science (83)	Algorithm (44) Computing (39)	LLMs have significantly advanced NLP, setting new performance standards, with ChatGPT having the most profound impact on GenAI. These LLMs are now fine-tuned using advanced ML techniques capable of generating fully autonomous NLP conversations, demonstrating the rapid progress and growing capabilities in AI and computing.
Psychology (108)	Behavioral (20) Cognitive (55) Mental (33)	GenAI is now being used in psychology to understand behavioral and cognit- ive dynamics, to analyse and predict human behavior and mental processes, thereby offering deeper insights into the human psyche. For example, studies on chatbot acceptance show how instrumental and non-instrumental gratific- ations, along with social norms, significantly influence user interactions with AI, revealing the intricate relationship between technology and human psy- chology.
Communication (685)	Information (460) Media (114) Communication (111)	While GenAI-generated content enhances information dissemination, it also poses risks of misinformation and factual inaccuracies. A critical examination of these issues, including algorithmic bias, privacy concerns, and the impact of GenAI on media and communication, reveals the need for ethical consider- ations and responsible use of technology in communication.

Table 9: Usage of GenAI post ChatGPT (2022–2024)

[Continued ...]

134

Discipline	Code (number)	Brief integrative narrative
Healthcare (694)	Medical (284) Clinical (238) Health (172)	GenAI has also led to transformative changes in healthcare. LLMs have re- volutionised medical curriculum development, teaching methodologies, and student assessments. These models hold promise for creating more effective and personalised medical training and patient care strategies. Their applica- tion in clinical settings is also expanding, with the potential to significantly improve patient outcomes and healthcare services.
Robotics (38)	Autonomous (25) Robotics (12) Humanoid (1)	In robotics, GenAI is enhancing the development of autonomous systems and humanoid robots. Technologies like ChatGPT are demonstrating the ability to autonomously generate natural language, which is vital for human-robot interaction. The ongoing research in this field is focused on optimising these interactions to be more ethical and to minimise harm, reflecting the growing sophistication and ethical considerations in the use of GenAI in robotics.

Table 9: [...continued]

3.10 GenAI Challenges and Concerns

Prior to ChatGPT (table 10), the concerns about GenAI included inaccuracies and ethical issues stemming from biased training data. These inaccuracies, ethical concerns and societal impacts remain the dominant challenge with GenAI post-ChatGPT (Table 11) despite the widespread usage. This shows that scholars in the GenAI community recognised these challenges early on and raised the alarm.

Discipline	Code (number)	Brief integrative narrative
Technical Challenges (10)	Performance (5) Errors (3) Accuracy (1) Scalability (1)	The most noted challenge was with the performance of GenAI models, particu- larly in ensuring their efficient and effective operation. There were other chal- lenges related to errors from inaccuracies and mistakes in GenAI-generated outputs. These concerns about accuracy further pointed to the challenges in maintaining precision in the outputs. Scalability problems reflected the dif- ficulties in expanding GenAI systems to handle larger or more complex tasks without a loss in performance or efficiency.
Ethical Concerns (3)	Bias (2) Privacy (1)	Bias was identified twice, highlighting the problem of GenAI systems poten- tially perpetuating or even amplifying existing societal biases. Privacy con- cerns were also noted, reflecting the complexities involved in responsibly handling personal data within GenAI systems.
Societal impact (2)	Society (2)	The two instances pointed to broader implications for society around the im- pact of GenAI on social dynamics and human interactions, raising questions about the long-term effects of increasingly relying on GenAI systems in vari- ous aspects of life. This is particularly relevant where decision-making, both at individual and organisational levels, becomes heavily dependent on GenAI recommendations. This dependency might reduce human engagement in crit- ical thinking and problem-solving, potentially leading to an erosion of these skills. Another concern relates to shifts in labour markets, and the moral im- plication in areas such as law enforcement or healthcare.

Table 10: Challenges with GenAI pre-ChatGPT (2012–2022)

Discipline	Code (number)	Brief integrative narrative
Technical challenges (791)	Performance (384) Accuracy (329) Complexity (39) Error (33) Scalability (6)	There is still a significant challenge with the performance of GenAI systems. For example, one abstract described the ongoing efforts to improve real-time processing capabilities in autonomous vehicles. The accuracy of predictions remains a crucial focus with research going into refining LLMs to better under- stand and generate human-like text. The issues of complexity and scalability further highlighted the ongoing work to simplify GenAI models for broader applications, particularly in fields such as healthcare, where complex data re- quires simplified yet effective analysis.
Ethical concerns (186)	Privacy (88) Bias (71) Consent (14) Moral (13)	There is a growing emphasis on privacy, especially in contexts where GenAI is used for personal data analysis, urging for enhanced data protection measures. Bias continues to be a critical ethical challenge, with some studies indicating that GenAI-driven hiring tools perpetuate gender bias. The need for informed consent and ethical considerations, such as the use of facial recognition tech- nology when used in public spaces, is important.
Societal impact (169)	Society (147) Employment (9) Human Interac- tion (8) Dependency (4) Social Impact (1)	The major concern is the way GenAI is transforming society particularly tradi- tional education systems, promoting personalised learning while raising ques- tions about reduced human interaction. There are envisaged and real employ- ment changes especially with job displacement, changed human interaction as a result of changes in social behaviour, and feared societal dependency on GenAI due to the increased AI integration.
Legal issues (163)	Legal (76) Law (37) Regulation (34) Compliance (16)	There is a need for new regulations in applications that use GenAI such as drone technology when it is used in surveillance. The implications have effects on compliance with the law and the associated regulations.
Data issues (36)	Data Privacy (20) Data Security (12) Data Quality (3) Data Availability (1)	Data privacy and security are highlighted for examples of research to secure GenAI systems against data breaches. Data quality and availability were illus- trated by studies on the impact of poor data quality on GenAI decision-making in critical sectors like finance.

Table 11: Challenges with GenAI post-ChatGPT (2022–2024)

Sentiments before and after ChatGPT 3.11

From 2012-2022 (Table 12), the sentiment towards GenAI was predominantly positive, reflecting optimism in sectors like healthcare, where GenAI's accuracy in disease identification promised to revolutionise diagnostics. However, concerns about privacy and ethical implications of GenAI in surveillance were noted, emphasising the need for thoughtful regulation. Years 2022–2024 witnessed a huge surge in positive sentiment towards GenAI, particularly for its applications in various fields, signaling its potential to address pressing global challenges. Yet, the rising optimism has not been without reservations, particularly around the impact of GenAI on job automation, highlighting the need for a balanced approach in embracing GenAI advancements.
Period	Sentiment	#	Brief narrative
2012 – 2022	Positive	46	The positive sentiment predominantly leans towards optimism and enthusiasm. An example is the breakthrough in AI-driven healthcare solutions where GenAI algorithms successfully identified diseases with greater accuracy than traditional methods. Such advancements demonstrate the potential of GenAI to revolutionise medical diagnostics and treatment, eliciting a sense of hope and excitement for the future of healthcare.
	Neutral	0	
	Negative	10	There were concerns and apprehensions about the impact of GenAI for example around the ethical implications of GenAI in surveillance, where fears about privacy invasion and misuse of GenAI for intrusive monitoring were highlighted. These concerns high- lighted the need for careful consideration and regulation in the deployment of AI tech- nologies to protect individual rights and freedoms.
2022 – 2024	Positive	1589	There is now an overwhelming positivity especially around the groundbreaking GenAI applications. For example, in environmental science, GenAI models actively predict climate change patterns, offering vital insights for mitigating environmental risks. These examples foster a sense of hope and anticipation for the potential of GenAI to address some of the world's most pressing challenges.
	Neutral	12	While limited the neutral sentiments indicate areas where the impact of GenAI is still uncertain or where its implications are yet to be fully understood. For example the use of GenAI in artistic endeavors, where there is a mix of curiosity and skepticism about the role of GenAI in creative processes.
	Negative	227	There are nonetheless still major concerns and reservations about GenAI and AI. A notable example was over GenAI in job automation, where there is apprehension about potential job losses and the displacement of human labour. These concerns highlight the complexities and ethical dilemmas that GenAI introduces, prompting a call for more stringent regulations and ethical guidelines.

Table 12: Sentiments before and after ChatGPT

4 DISCUSSION OF FINDINGS

In order to offer a balanced discussion of the key findings, we adopted a post-digital lens mainly because it allows for a critical examination of the seamless integration of digital technologies into daily life, highlighting the intertwined nature of digital and non-digital realms and the cultural, social, and economic implications of this ubiquity (Cramer, 2015).

4.1 More Rapid Technological Evolutions and their Impacts

The rapid integration of ChatGPT and similar GenAI technologies marks a significant evolution in AI capabilities. This evolution is characterised by the sophisticated development of transformer architectures, pre-training processes, and fine-tuning methodologies. These technical advancements have revolutionised sectors like healthcare, education, and software engineering, demonstrating the potential for significant innovations and efficiency enhancements (Y. K. Dwivedi et al., 2023). This technological leap signifies a transformative change in how problem-solving, knowledge creation, and service delivery are approached across different sectors. The leap introduces a new era of enhanced productivity and innovation, where GenAI's capabilities can be harnessed to tackle complex challenges more effectively. For example, a 2024 McKinsey survey shows that GenAI adoption has more than doubled in the previous five years, and there is an increase in GenAI investment (McKinsey & Company, 2024). For practitioners and policymakers, this new era points to the urgency of adapting to rapidly changing technologies. For educators and healthcare professionals, GenAI offers tools for unprecedented advancements in personalised education and patient care. Ultimately, GenAI impacts society at large, promising to elevate the quality of life through improved services and innovations.

4.2 Transdisciplinary Integration

The post-ChatGPT era has facilitated greater disciplinary collaboration regardless of region, enabling the integration of GenAI into diverse fields. This unprecedented collaboration is driven by the data-rich insights that GenAI platforms provide, which offer a deeper, evidence-based understanding of complex phenomena across domains. These data-driven capabilities empower researchers to not only test existing theories but also extend them into areas previously unexplored, thereby facilitating unique innovations and the creation of new theories and frameworks (Klein-Avraham et al., 2024). For example, researchers in Information Systems (IS) and Computer Science can use GenAI to provide unique opportunities to refine and adapt their core theories to address challenges in other domains, such as healthcare, education, and the arts (Twinomurinzi et al., 2022). This dynamic interaction allows for the tailoring of theoretical constructs to suit the specific needs and contexts of other disciplines (Y. K. Dwivedi et al., 2023).

Moreover, the insights generated through GenAI encourage the emergence of entirely new disciplines that reflect the convergence of traditional fields. For example, the integration of computational methodologies with social sciences or the arts leads to the development of hybrid fields such as computational sociology or AI-driven creative design (Davidson, 2024; Tan & Luhrs, 2024). This transdisciplinary potential not only bridges existing gaps between disciplines but also creates a fertile ground for the development of more holistic approaches to solving complex, global challenges.

4.3 Ethics Reflections

The rapid integration of GenAI has brought ethical considerations to the forefront, particularly concerning privacy, bias, and the equitable distribution of its benefits. These dilemmas challenge existing social and technical frameworks, necessitating careful navigation to ensure fairness and accountability (Xivuri & Twinomurinzi, 2022). For regulators and ethicists, this signals an urgent need to develop guidelines that promote the responsible and transparent use of GenAI (Cherner et al., 2024). For the general public, GenAI represents a dual-edged sword: it offers significant advancements while also posing risks to privacy, equity, and access to opportunity. Balancing these concerns is critical to ensuring that GenAI's transformative

potential benefits society as a whole and does not increase the digital divide.

4.4 Legal Implications and Societal Adjustments

The integration of GenAI into everyday life presents new and intricate legal, societal, and regulatory complexities. GenAI technologies challenge the boundaries of existing societal norms and legal frameworks, exposing significant gaps in their ability to address the rapidly evolving nature of emerging technologies. This means that legal experts and policymakers need to learn to innovate within legal structures to safeguard individual and collective rights without limiting technological progress.

The transformative potential of GenAI extends to the creation of entirely new disciplines and the jobs required within them. As these emerging fields take shape, educational institutions will need to rise to the challenge of designing entirely new curricula that equip learners with the skills necessary for the future world of work. This includes new programmes that merge technical expertise with ethical, social, and creative insights, ensuring future-fit skills.

For society, embracing a new world of work means rethinking traditional employment structures and adopting more flexible, adaptive approaches to workforce development. Policymakers should consider more experimental and dynamic methods in crafting policies that anticipate and respond to the fluid realities of a GenAI- and AI-driven economy. This could include policies that support continuous learning, universal access to reskilling opportunities, and equitable participation in the benefits of GenAI. The same applies to the legal domain which needs to become more reflexive and adaptive, developing mechanisms that not only regulate GenAI but also encourage its responsible and equitable deployment without stifling innovation.

The deeper societal issue is what it means to be human in an era shaped by these technologies that supercede human intelligence in many ways, and the role of human values such as creativity, spirituality, empathy, and connection.

4.5 Limitations

The study is limited in its usage of only two academic databases without taking into account the vast non-academic literature, nor other GenAI platforms that have emerged since ChatGPT. The limited scope is partially a result of the tools available to conduct bibliometric analyses.

5 CONCLUSIONS

The study investigated the scholarly discourse on GenAI since before ChatGPT and after Chat-GPT using a bibliometric review and thematic analysis. We sought to interrogate how prevalent patterns of GenAI usage have manifested in scholarly discourse and the primary drivers behind the adoption of these technologies. The reality of the transition from one million "early

adopters" in five days to an "early majority" of 100 million users within two months indicates that ChatGPT ushered an inflection point in society (Altman, 2022).

The key factor that drives the scholarly discourse of GenAI is transdisciplinary collaboration across regions, disciplines and sectors. Transdisciplinarity is about building bridges between disciplines that have different notions of reality, which bridges themselves become a reality of their own with a new set of language to make sense of the bridge (Klein, 2004). GenAI has managed to navigate the challenges of transdisciplinarity by addressing key limitations inherent in traditional, reductionist approaches. The traditional approaches often prioritise rational logic over relational or qualitative understanding, overlooking the relatedness of disciplines and the role of intuition in knowledge creation (Max-Neef, 2005). In contrast, GenAI fosters a more integrated perspective, bridging the gaps caused by the uni-disciplinary focus prevalent in most educational institutions, particularly at the undergraduate level, and the compartmentalisation of funding into narrowly defined tasks and sub-disciplines (Max-Neef, 2005). While current universities guard disciplinary silos and primarily engage in interdisciplinary collaboration, GenAI opens ways for truly transdisciplinary innovation.

While GenAI has democratised access to advanced AI capabilities, the digital divide continues to pose a significant challenge. Certain demographics and regions, especially in Africa, have unequal access to technology, infrastructure, and digital literacy. These disparities risk exacerbating existing inequalities, leaving marginalised communities further behind.

The rapid adoption of GenAI has also brought ethical dilemmas to the forefront, particularly around issues such as privacy, bias, and equitable access. Ethical lapses in GenAI development and deployment could deepen societal mistrust and perpetuate existing inequalities. A proactive approach is needed to embed ethical considerations into the design, training, and deployment of GenAI models. This includes promoting transparency, accountability, and inclusivity at every stage of development.

The absence of robust legal and regulatory frameworks for GenAI presents significant risks. Without clear guidelines, the deployment of GenAI could lead to misuse, data breaches, and violations of individual rights. Legal experts and policymakers must act swiftly to develop adaptive and reflexive regulations that safeguard privacy, ensure accountability, and promote equitable use. Such frameworks should encourage innovation while protecting societal interests, ensuring that GenAI evolves in a manner that benefits all.

This paper makes a contribution to knowledge and offers some implications for academia, researchers, practice and policy. In IS, GenAI provides a basis for extending existing IS theories using data-driven insights. GenAI allows for the creation of hybrid theoretical frameworks that incorporate both technical efficiency and social implications. In Computer Science, GenAI demands a rethinking of traditional frameworks to include human feedback, contextual accuracy, and ethical AI design. The advancements in transformer models and reasoning capabilities prompt theoretical explorations into designing adaptive systems that balance technical innovation with accountability.

Educators face the challenge of preparing learners for a world reshaped by GenAI. This requires the design of new curricula that integrate technical, ethical, and creative dimensions of

GenAI, equipping students with the skills to navigate hybrid roles across emerging fields and disciplines. Educational institutions must prioritise disciplinary collaboration, offering programmes that blend technical expertise with insights from the humanities and social sciences. Additionally, there is a pressing need to foster lifelong learning by embedding AI literacy and digital skills training across all levels of education. Such efforts will ensure that students are not only proficient in GenAI and AI technologies but also prepared to address their societal implications.

Practitioners will similarly need to adapt to the evolving demands of a GenAI-driven world by embracing continuous learning and innovation. Workforce development initiatives should focus on reskilling and upskilling to align with the hybrid roles created by GenAI.

Policymakers will be required to navigate the complexities of integrating GenAI into society by developing adaptive and reflexive regulatory frameworks. These regulations should address critical issues such as privacy, bias, accountability, and equitable use, while encouraging an environment that supports innovation and humanity. Closing the digital divide is another pressing priority, requiring investments in infrastructure, digital literacy, and development policies. Policymakers should also adopt experimental approaches, such as sandbox regulations, to allow for dynamic and iterative responses to the challenges posed by GenAI.

References

- Ali, S., DiPaola, D., Lee, I., Hong, J., & Breazeal, C. (2021). Exploring generative models with middle school students. *Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems*, 1–13. https://doi.org/10.1145/3411764.3445226
- Alkaissi, H., & McFarlane, S. I. (2023). Artificial hallucinations in ChatGPT: Implications in scientific writing. *Cureus*. https://doi.org/10.7759/cureus.35179
- Altman, S. (2022). *ChatGPT launched on Wednesday. Today it crossed 1 million users!* [Retrieved 15 June 2025]. https://x.com/sama/status/1
- Aria, M., & Cuccurullo, C. (2017). *bibliometrix*: An R-tool for comprehensive science mapping analysis. *Journal of Informetrics*, 11(4), 959–975. https://doi.org/10.1016/J.JOI.2017 .08.007
- Ayers, J. W., Poliak, A., Dredze, M., Leas, E. C., Zhu, Z., Kelley, J. B., Faix, D. J., Goodman, A. M., Longhurst, C. A., Hogarth, M., & Smith, D. M. (2023). Comparing physician and artificial intelligence chatbot responses to patient questions posted to a public social media forum. *JAMA Internal Medicine*, 183(6), 589. https://doi.org/10.1001/jamaint ernmed.2023.1838
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, *3*(2), 77–101. https://doi.org/10.1191/1478088706QP0630A
- Cascella, M., Montomoli, J., Bellini, V., & Bignami, E. (2023). Evaluating the feasibility of ChatGPT in healthcare: An analysis of multiple clinical and research scenarios. *Journal of Medical Systems*, 47(1). https://doi.org/10.1007/s10916-023-01925-4

- Cherner, T., Foulger, T. S., & Donnelly, M. (2024). Introducing a generative AI decision tree for higher education: A synthesis of ethical considerations from published frameworks & guidelines. *TechTrends*, 69(1), 84–99. https://doi.org/10.1007/S11528-024-0102 3-3
- Cotton, D. R. E., Cotton, P. A., & Shipway, J. R. (2023). Chatting and cheating: Ensuring academic integrity in the era of ChatGPT. *Innovations in Education and Teaching International*, *61*(2), 228–239. https://doi.org/10.1080/14703297.2023.2190148
- Cramer, F. (2015). What is 'Post-digital'? In D. M. Berry & M. Dieter (Eds.), *Postdigital aesthetics: Art, computation and design* (pp. 12–26). London: Palgrave Macmillan UK. https://doi .org/10.1057/9781137437204_2
- Davidson, T. (2024). Start generating: Harnessing generative artificial intelligence for sociological research. *Socius*, *10*. https://doi.org/https://doi.org/10.1177/2378023124125 9651
- Dwivedi, R., & Elluri, L. (2024). Exploring generative artificial intelligence research: A bibliometric analysis approach. *IEEE Access*, *12*, 119884–119902. https://doi.org/10.1109 /ACCESS.2024.3450629
- Dwivedi, Y. K., Kshetri, N., Hughes, L., Slade, E. L., Jeyaraj, A., Kar, A. K., Baabdullah, A. M., Koohang, A., Raghavan, V., Ahuja, M., Albanna, H., Albashrawi, M. A., Al-Busaidi, A. S., Balakrishnan, J., Barlette, Y., Basu, S., Bose, I., Brooks, L., Buhalis, D., ... Wright, R. (2023). Opinion paper: "So what if ChatGPT wrote it?" Multidisciplinary perspectives on opportunities, challenges and implications of generative conversational AI for research, practice and policy. *International Journal of Information Management*, *71*, 102642. https://doi.org/10.1016/J.IJINFOMGT.2023.102642
- Gilson, A., Safranek, C. W., Huang, T., Socrates, V., Chi, L., Taylor, R. A., & Chartash, D. (2023). How does ChatGPT perform on the United States Medical Licensing Examination (USMLE)? The implications of large language models for medical education and knowledge assessment. *JMIR Medical Education*, *9*, e45312. https://doi.org/10.2196 /45312
- Gupta, A., Müller, A. T., Huisman, B. J., Fuchs, J. A., Schneider, P., & Schneider, G. (2017). Generative recurrent networks for de novo drug design. *Molecular Informatics*, *37*(1–2). https://doi.org/10.1002/minf.201700111
- Hagos, D. H., Battle, R., & Rawat, D. B. (2024). Recent advances in generative AI and large language models: Current status, challenges, and perspectives. *IEEE Transactions on Artificial Intelligence*, 5, 5873–5893. https://doi.org/10.1109/TAI.2024.3444742
- Iskender, A. (2023). Holy or unholy? Interview with Open AI's ChatGPT. *European Journal of Tourism Research*, 34, 3414. https://doi.org/10.54055/EJTR.V34I.3169
- Khan, N., Khan, Z., Koubaa, A., Khan, M. K., & bin Salleh, R. (2024). Global insights and the impact of generative AI-ChatGPT on multidisciplinary: A systematic review and bibliometric analysis. *Connection Science*, *36*(1), 2353630. https://doi.org/10.1080/0 9540091.2024.2353630

- Klein, J. T. (2004). Prospects for transdisciplinarity. *Futures*, *36*(4), 515–526. https://doi.org /10.1016/J.FUTURES.2003.10.007
- Klein-Avraham, I., Greussing, E., Taddicken, M., Dabran-Zivan, S., Jonas, E., & Baram-Tsabari, A. (2024). How to make sense of generative AI as a science communication researcher? A conceptual framework in the context of critical engagement with scientific information. *Journal of Science Communication*, 23(6), A05. https://doi.org/10.22323/2.2306 0205
- Liu, J., Wang, C., Liu, Z., Gao, M., Xu, Y., Chen, J., & and, Y. C. (2024). A bibliometric analysis of generative AI in education: Current status and development. *Asia Pacific Journal of Education*, 44(1), 156–175. https://doi.org/10.1080/02188791.2024.2305170
- Lopes, A. A. (2024). Generative artificial intelligence in health professions: A bibliometric descriptive analysis. 18th International Technology, Education and Development Conference, INTED2024 Proceedings. https://doi.org/10.21125/INTED.2024.1936
- Max-Neef, M. A. (2005). Foundations of transdisciplinarity. *Ecological Economics*, *53*(1), 5–16. https://doi.org/10.1016/J.ECOLECON.2005.01.014
- McCallum, S. (2023). ChatGPT banned in Italy over privacy concerns [Retrieved 12 May 2025]. https://www.bbc.com/news/technology-65139406
- McKinsey & Company. (2023). The economic potential of generative AI: The next productivity frontier [Retrieved 12 May 2025]. https://www.mckinsey.com/capabilities/mckinsey -digital/our-insights/the-economic-potential-of-generative-ai-the-next-productivity -frontier#/
- McKinsey & Company. (2024). What is generative AI? [Retrieved 12 May 2025]. https://ww w.mckinsey.com/featured-insights/mckinsey-explainers/what-is-generative-ai
- Mirsky, Y., & Lee, W. (2021). The creation and detection of deepfakes: A survey. ACM Computing Surveys, 54(1), 1–41. https://doi.org/10.1145/3425780
- OpenAI. (2025). OpenAI [Retrieved 12 May 2025]. openai.com/about/
- Pavlik, J. V. (2023). Collaborating with chatgpt: Considering the implications of generative artificial intelligence for journalism and media education. *Journalism & Mass Communication Educator*, 78(1), 84–93. https://doi.org/10.1177/10776958221149577
- Rives, A., Meier, J., Sercu, T., Goyal, S., Lin, Z., Liu, J., Guo, D., Ott, M., Zitnick, C. L., Ma, J., & Fergus, R. (2021). Biological structure and function emerge from scaling unsupervised learning to 250 million protein sequences. *Proceedings of the National Academy of Sciences*, *118*(15), e2016239118. https://doi.org/10.1073/pnas.2016239118
- Rudolph, J., Tan, S., & Tan, S. (2023). ChatGPT: Bullshit spewer or the end of traditional assessments in higher education? *Journal of Applied Learning and Teaching*, 6(1), 342–363. https://doi.org/10.37074/JALT.2023.6.1.9
- Salvagno, M., Taccone, F. S., & Gerli, A. G. (2023). Can artificial intelligence help for scientific writing? *Critical Care*, 27(1). https://doi.org/10.1186/s13054-023-04380-2
- Suh, M. (, Youngblom, E., Terry, M., & Cai, C. J. (2021). AI as social glue: Uncovering the roles of deep generative AI during social music composition. *Proceedings of the 2021*

CHI Conference on Human Factors in Computing Systems. https://doi.org/10.1145/341 1764.3445219

- Sun, J., Liao, Q. V., Muller, M., Agarwal, M., Houde, S., Talamadupula, K., & Weisz, J. D. (2022). Investigating explainability of generative AI for code through scenario-based design. 27th International Conference on Intelligent User Interfaces, 212–228. https://doi .org/10.1145/3490099.3511119
- Tan, L., & Luhrs, M. (2024). Using generative AI Midjourney to enhance divergent and convergent thinking in an architect's creative design process. *The Design Journal*, 27(4), 677– 699. https://doi.org/10.1080/14606925.2024.2353479
- Temsah, M., Jamal, A., Alhasan, K., Temsah, A. A., & Malki, K. H. (2024). OpenAI o1-Preview vs. ChatGPT in healthcare: A new frontier in medical AI reasoning. *Cureus*, *16*(10), e70640. https://doi.org/10.7759/CUREUS.70640
- Tlili, A., Shehata, B., Adarkwah, M. A., Bozkurt, A., Hickey, D. T., Huang, R., & Agyemang, B. (2023). What if the devil is my guardian angel: ChatGPT as a case study of using chatbots in education. *Smart Learning Environments*, *10*(1). https://doi.org/10.1186/s 40561-023-00237-x
- Twinomurinzi, H., & Gumbo, S. (2023). ChatGPT in scholarly discourse: Sentiments and an inflection point. In A. Gerber & M. Coetzee (Eds.), *SAICSIT 2023* (pp. 258–272, Vol. 1878). Springer. https://doi.org/10.1007/978-3-031-39652-6_17
- Twinomurinzi, H., Mhlongo, S., Bwalya, K., Bokaba, T., & Mbeya, S. (2022). Multidisciplinarity in data science curricula. *African Conference on Information Systems and Technology*, 21. https://digitalcommons.kennesaw.edu/acist/2022/presentations/21
- VOSviewer. (2025). VOSviewer, visualizing sientific landscapes [Retrieved 12 May 2025]. ht tps://www.vosviewer.com/
- Weisz, J. D., Muller, M., Houde, S., Richards, J., Ross, S. I., Martinez, F., Agarwal, M., & Talamadupula, K. (2021). Perfection not required? Human-AI partnerships in code translation. 26th International Conference on Intelligent User Interfaces, 402–412. https://doi.or g/10.1145/3397481.3450656
- Williams, R., Park, H. W., & Breazeal, C. (2019). A is for artificial intelligence: The impact of artificial intelligence activities on young children's perceptions of robots. Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems, 1–11. https://doi.org /10.1145/3290605.3300677
- Wu, Y., Mou, Y., Li, Z., & Xu, K. (2020). Investigating American and Chinese subjects' explicit and implicit perceptions of AI-generated artistic work. *Computers in Human Behavior*, 104, 106186. https://doi.org/10.1016/j.chb.2019.106186
- Xivuri, K., & Twinomurinzi, H. (2022). A Habermasian approach to fair processes in AI algorithms. *Communications in Computer and Information Science*, 1551 CCIS, 335–343. https://doi.org/10.1007/978-3-030-95070-5_22
- Zhang, Z., Xu, Y., Wang, Y., Yao, B., Ritchie, D., Wu, T., Yu, M., Wang, D., & Li, T. J. (2022). StoryBuddy: A human-AI collaborative chatbot for parent-child interactive storytelling

with flexible parental involvement. *CHI Conference on Human Factors in Computing Systems*, 1–21. https://doi.org/10.1145/3491102.3517479

Zhong, T., Liu, Z., Pan, Y., Zhang, Y., Zhou, Y., Liang, S., Wu, Z., Lyu, Y., Shu, P., Yu, X., Cao, C., Jiang, H., Chen, H., Li, Y., Chen, J., Hu, H., Liu, Y., Zhao, H., Xu, S., ... Liu, T. (2024, September). Evaluation of OpenAI o1: Opportunities and challenges of AGI [Retrieved 12 May 2025]. https://arxiv.org/abs/2409.18486v1